

## **REQUEST FOR PROPOSAL**

for

## **AUTOMOBILE PARTS**

RFP#: DGS-57224

**September 16, 2009** 

State of California
Department of General Services
707 3<sup>rd</sup> Street
West Sacramento, CA 95605

## **Table of Contents**

Section	Section Title		Page	
1	Introduction and Overview of Requirements		1-1	
	1.1	Purpose of this Request for Proposal (RFP)	1-2	
	1.2	Overview of the RFP	1-3	
	1.3	Scope of the RFP and Bidder Admonishment	1-3	
	1.4	Availability	1-4	
	1.5	Procurement Official and Department Contact	1-4	
	1.6	Key Action Dates	1-5	
	1.7	Americans with Disabilities Act (ADA)	1-6	
2	Rules Gov	verning Competition	2-1	
	2	Rules Governing Competition	2-2	
	2.1	Identification and Classification of RFP Requirements	2-2	
	2.1.1	Mandatory Requirements	2-2	
	2.1.2	Desirable Items	2-2	
	2.2	Proposal Requirements and Conditions	2-2	
	2.2.1	General	2-2	
	2.2.2	RFP Documents	2-3	
	2.2.3	Examination of the Work	2-3	
	2.2.4	Questions Regarding the RFP	2-3	
	2.2.5	Request to Change the Requirements of the RFP	2-4	
	2.2.6	Bidders' Conference	2-4	
	2.2.7	Addenda	2-4	
	2.2.8	State of California Plastic Trash Bag Certification Violations	2-5	
	2.2.9	Bonds	2-5	
	2.2.10	Follow-on Contracts (PCC 10365.5, PCC 10410, and PCC 10411)	2-5	
	2.2.11	Prompt Payment Discounts	2-5	
	2.2.12	Joint Proposals	2-6	
	2.2.13	State of California Air or Water Pollution Violations	2-6	
	2.2.14	State of California Fair Employment and Housing Commission	2-6	
		Regulations		
	2.2.15	Exclusion for Conflict of Interest	2-6	
	2.2.16	State of California Seller's Permit	2-6	
	2.2.17	Disclosure of Financial Interests	2-7	
	2.2.18	State of California Unfair Practices Act and Other Laws	2-7	
	2.3	Bidding Steps	2-7	
	2.3.1	Preparation of Proposals	2-7	
	2.3.2	Bidder's Intention to Submit a Proposal	2-7	
	2.3.3	Draft Proposals	2-8	
	2.3.4	Confidential Discussions	2-9	
	2.3.5	Final Proposal	2-9	
	2.3.5.1	Submission of Final Proposal	2-9	
	2.3.5.2	Bidder's Costs	2-9	
	2.3.5.3	Proposal Responsiveness	2-9	
	2.3.5.4	False or Misleading Statements	2-9	
	2.3.5.5	Proposal Signature	2-10	
	2.3.5.6	Delivery of Proposals	2-10	
	2.3.5.7	Withdrawal and Resubmission/Modification of Proposals	2-10	
	2.3.5.8	Clarification of Final Proposals	2-11	

Section	Section T	itle	Page
2	Rules Gov	verning Competition (continued)	
	2.3.5.9	Flawed Final Proposals	2-11
	2.3.5.10	Confidentiality	2-11
	2.3.6	Demonstrations (if required)	2-12
	2.3.7	Sealed Cost Openings	2-12
	2.3.8	Rejection of Proposals	2-12
	2.3.8.1	General	2-12
	2.3.8.2	Evaluation Questions	2-12
	2.3.8.3	Errors in the Final Proposal	2-12
	2.3.9	Contract Award	2-13
	2.3.10	Debriefing	2-13
	2.4	Other Information	2-14
	2.4.1	Protests	2-14
	2.4.1.1	Requirements Protest	2-14
	2.4.1.2	Award Protest	2-14
	2.4.2	News Releases	2-15
	2.4.3	Disposition of Proposals	2-15
3	Current S		3-1
	3	Current System	3-2
4	Proposed	System	4-1
	4.1	Overall Goal	4-2
	4.2	State of California Environmentally Preferable Purchasing	4-2
5	Administr	ative Requirements	5-1
	5.1	Introduction	5-2
	5.2	Responses to Administrative Requirements	5-2
	5.3	Designation of Requirements	5-2
	5.4	Administrative Requirements	5-3
	5.4.1	Bidder Responsibility (M)	5-3
	5.4.2	Confidentiality (M)	5-3
	5.4.3	Contractor Responsibility (M)	5-3
	5.4.4	Contractor Name Change (M)	5-4
	5.4.5	Bidder Declaration (M)	5-4
	5.4.5.1	Subcontractors (M)	5-4
	5.4.5.2	Notice of Subcontractors (M)	5-5
	5.4.6	Commercially Useful Function (M)	5-5
	5.4.7	Non-collusion Affidavit (M)	5-5
	5.4.8	Payee Data Record (M)	5-5
	5.4.9	State of California Seller Permit (M)	5-5
	5.4.10	State of California Darfur Contracting Act of 2008 (M)	5-6
	5.4.11	Contract Terms (M)	5-6
	5.4.12	Non-Discrimination Toward WTO GPA Signatories (M)	5-6
	5.4.13	California Disabled Veteran Business Enterprise (DVBE) Program	5-6
	F 4 4 4	Requirement (Waived)	   F 7
	5.4.14	DVBE Declaration	5-7
	5.5	Preferences and Incentives	5-7
	5.5.1	Small Business Preference (NM)	5-7
	5.5.2	DVBE Incentive (NM) (Waived)	5-8
	5.5.3	Target Area Contract Preference Act (TACPA) Preference,	5-9
		Enterprise Zone Act (EZA) & Local Agency Military Base Recovery	
		Area (LAMBRA) Act Preference Preference (NM)	

Section	Section Title		Page
6	Technical	Requirements	6-1
	6.1	Introduction	6-1
	6.1.1	Statement of Work	6-1
	6.2	Responses to Technical and Business Requirements	6-1
	6.3	Designation of Requirements	6-2
		Technical Requirements	6-6
		Business Requirements	6-18
		EPP Requirements	6-24
7	Cost		7-1
	7.1	Introduction	7-2
	7.2	Pricing Structure	7-2
	7.3	Quantities	7-2
	7.4	Worksheet Instructions	7-2
	7.4.1	Download and Save	7-3
	7.4.2	Cost Worksheet	7-3
	7.4.3	Discount Worksheet	7-4
	7.4.4	Important worksheet details	7-4
	7.5	Cost Submittal	7-4
8	Proposal Format		8-1
	8.1	Introduction	8-2
	8.2	General Format Requirements	8-2
	8.3	Final Proposal Format and Content	8-3
	8.3.1	Instructions	8-3
	8.3.2	Proposal Copies	8-3
	8.3.3	Proposal Content	8-3
	8.4	Draft Proposal Format and Content	8-5
9	Evaluation		9-1
	9.1	Introduction	9-2
	9.2	State Evaluation Team	9-2
	9.3	Receipt	9-2
	9.4	Negotiations	9-2

Section	ion Section Title		
10	Exhibits		
	10.1	Letter of Intent to Bid	
	10.2	Statement of Work	
	10.3	Contract Development/Standard Agreement (STD 213)	
	10.4	Competitive Bidding and Proposal Responsiveness	
	10.5	Administrative Requirements Response	
	10.6	Retailer's Seller Permit Information	
	10.7	Contractor Contract Manager Information	
	10.8	Contractor Ordering Information	
	10.9	Cost Worksheet, Light Duty	
	10.10	Post-consumer Recycle Content (PCRC) Certification Workbook	
	10.11	California Department of Corrections and Rehabilitation (CDCR)	
		Special Provisions	
	10.12	Darfur Contracting Act Form	
	10.13	Contract Usage Report	
	10.14	Customer Reference Form	
	10.15	WSCA Terms and Conditions	
	10.16	State of California Small Business Preference Information	
	10.17	Authorized Locations	
	10.18	Wholesale Distribution Center Locations	
	10.19	Market Basket Evaluation Form (Sample)	

\*\*\*\*

## **SECTION 1**

## INTRODUCTION AND OVERVIEW OF REQUIREMENTS

LIGHT DUTY AUTOMOTIVE PARTS

### 1. Introduction and Overview of Requirements

### 1.1 Purpose of this Request for Proposal (RFP)

The purpose of this Request for Proposal (hereafter called the RFP) is to elicit proposals from firms qualified to provide after market Light Duty Automotive Parts to the State of California (State), participating local agencies, and participating Western States Contracting Alliance (WSCA) States, at contracted pricing in accordance with the requirements of this RFP.

Bidders submitting a proposal to become the State's contractor for Auto Parts must bid all categories listed (Exhibit 10.9). The State intends that the Supplier(s) will work in cooperation with the State to furnish quality products at the lowest price available in a timely and efficient manner. The selected Suppliers will be the primary point of providing warehousing, data collection, reporting, and distribution of Auto Parts to the State.

The scope of this RFP includes automotive products as identified in Exhibit 10.9, Cost Worksheet. The State reserves the right to eliminate any specific category from the final award, if indicated by an analysis of RFP proposals.

The term of the contract resulting from this RFP will be for three (3) years. The contract will also contain options to extend the contract for two (2) additional one (1) year periods upon agreement between the supplier and the State. All discounts, terms, and conditions will remain the same for the entire contract period including any extension periods.

The use of this contract will be non-mandatory for all State departments.

The awarded supplier(s) shall permit local governmental agencies and participating Western States Contracting Alliance (WSCA) States to participate under this contract (See Exhibit 10.15 WSCA Standard Contract Terms and Conditions). For the purpose of this proposal, local government agencies are defined as "any city, county, city and county, district or other governmental body or corporation empowered to expend public funds for the acquisition of products", (Reference Public Contract Code Chapter 2, Section 10298 (a) (b)). Local governmental agencies shall have the same rights and privileges as the State under the terms of this contract. Any local governmental agency desiring to participate shall be required to submit to the same responsibilities as do State agencies and has no authority to amend, modify or change any condition of this contract.

The Western States Contracting Alliance (WSCA) is a cooperative group-contracting consortium for state government departments, institutions, agencies and political subdivisions (i.e., colleges and universities, school districts, counties, cities, etc.,) for the States of Alaska, Arizona, California, Colorado, Hawaii, Idaho, Minnesota, Montana, Nevada, New Mexico, Oregon, South Dakota, Utah, Washington, and Wyoming. WSCA is a subset of the National Association of State Procurement Officials (NASPO). NASPO is a non-profit association dedicated to strengthening the procurement community through education, research, and communication. It is made up of the directors of the central purchasing offices in each of the 50 states, the District of Columbia and the territories of the United States.

Exhibit 10.15 contains WSCA Terms and Conditions applicable to a WSCA procurement process and subsequent contract. When there is duplication between the California Terms and Conditions and the WSCA Terms and Conditions, CA DGS-PD and all bidders shall comply with the California Terms and Conditions.

The Price Agreement(s) resulting from this procurement (Master Price Agreements) may be used by WSCA member states and their political subdivisions (such as city and county governments, public education and institutions of higher education). WSCA price agreements, including the Master Price Agreement(s) resulting from this RFP, may be used by other NASPO states with authorization from the WSCA directors and subject to approval of the individual state procurement director and local statutory provisions.

All WSCA Participating States and Purchasing Entities reserve the right to competitively solicit additional sources for commodities within the scope of this RFP during the Price Agreement term, where deemed to be in the best interest of the state(s) or entities involved. Further, Participating States may have, or develop other contracts for commodities within the scope of this RFP. Any Participating State reserves the right to contract for partial commodity categories or not participate in the award if deemed to not be in the best interests of that Participating State.

Local Entity Participation: Use of specific WSCA cooperative contracts by state agencies, political subdivisions and other entities (including cooperatives) authorized by individual state's statutes to use state contracts are subject to the approval of the respective State Chief Procurement Official. Issues of interpretation and eligibility for participation are solely within the authority of the respective State Chief Procurement Official.

#### 1.2 Overview of the RFP

The State of California, Department of General Services, Procurement Division (DGS-PD) is issuing a two phase RFP. Phase 1 will be used to determine and score administrative and technical compliance for responsive and responsible bidders. Phase 2, will be cost. After the RFP responses have been received and evaluated by the State's WSCA Evaluation Team, the State reserves the right to award the contract to the lowest cost supplier meeting the technical and administrative requirements and any supplier within 20% of the low cost analysis meeting the technical and administrative requirements.

The awarded bidder(s) will be responsible for supplying all goods and services described in this RFP to meet the requirements of the State.

### 1.3 Scope of the RFP and Bidder Admonishment

This RFP is being conducted under the policies developed by the Department of General Services as provided under Public Contract Code Section 10298 et seq. This RFP contains the instructions governing the requirements for a proposal to be submitted by interested Bidders. The format used to submit information and the material to be included therein follows. This RFP also addresses the requirements that Bidders must meet to be eligible for consideration, as well as addressing Bidders' responsibilities.

This procurement will follow a phased approach designed to increase the likelihood that Final Proposals will be received without disqualifying defects. The additional step(s) will:

- 1) Ensure that the Bidders clearly understand the State's requirements before attempting to develop their final solutions;
- 2) Ensure that the State clearly understands what each Bidder intends to propose before those proposals are finalized; and
- 3) Give the State and each Bidder the opportunity to discuss weak or potentially unacceptable elements of a Bidder's proposal and give the Bidder the opportunity to modify its proposal to correct such problems.

Specific information regarding such steps is found in Section 2, Rules Governing Competition, Section 8, Proposal Format and Section 9, Evaluation.

If a Bidder expects to be afforded the benefits of the steps included in this RFP, the Bidder must take the responsibility to:

- 1) Carefully read the entire RFP;
- 2) Seek clarification by asking questions in a timely manner;
- Submit all required responses, complete to the best of Bidders ability, by the required dates and times;
- 4) Make sure that all procedures and requirements of the RFP are accurately followed and appropriately addressed; and
- 5) Carefully re-read the entire RFP before submitting a bid.

### 1.4 Availability

The products proposed to meet the requirements of this RFP must be available throughout the duration of the contract term.

#### 1.5 Procurement Official

The Procurement Official contact information for this RFP is listed below:

Eileen Tardiff
Procurement Division
Department of General Services
707 Third Street, 2nd Floor
West Sacramento, CA 95605
Telephone: (916) 375-4390
Facsimile: (916) 375-4663

E-mail: eileen.tardiff@dgs.ca.gov

All inquiries, questions, and proposal submittals must be directed to this person only, unless otherwise instructed in this RFP or if this person is changed by addendum to this RFP. The Procurement Official will also act as the Department Contact for visits or other information.

Oral communications directly with procurement officers and employees concerning this RFP are discouraged and shall not be binding to the State. Bidders should only rely on written statements issued from the Procurement Official.

## 1.6 Key Action Dates

Listed below are the important dates and times related to this RFP. If the State finds it necessary to change any of these dates before the Final Proposal submission, it will be accomplished by an addendum. All dates after the Final Proposal submission deadlines are approximate and may be adjusted as conditions indicate without addendum to the RFP.

Unless otherwise indicated below, Bidder submissions shall be due by the close of business on the date indicated. Close of business day shall be 5:00 p.m. All times are Pacific Time.

No.	Event/Action	Action Date	Time
1.	RFP Release	9/16/09	
2.	Last day to submit questions for clarification of RFP for Bidders' Conference	9/23/09	5:00 pm (PDT)
3.	Last day to submit Letter of Intent to Bid	9/28/09	5:00 pm (PDT)
4.	Bidders' Conference  Location: Ziggurat Building Department of General Services 707 Third Street, 2nd Floor West Sacramento, California 95605  See Section 2.2.6, Bidders' Conference for more	9/30/09	1:00 pm (PDT)
	information.		
5.	Last day to submit final questions for clarification of RFP prior to submittal of Draft Proposals	10/5/09	5:00 pm (PDT)
6.	Last day to request a change in the requirements of the RFP	10/5/09	5:00 pm (PDT)
7.	Response to Bidders' questions released	10/8/09	5:00 pm (PDT)
8.	Last day to protest the RFP requirements	10/13/09	5:00 pm (PDT)
9.	Submission of <u>Draft</u> Proposals*	10/20/09	5:00 pm (PDT)
10.	Confidential Discussions with Individual Bidders	10/26/09	9:00 am to 5:00 pm (PDT)
11.	Submission of Final Proposals	10/30/09	5:00 pm (PDT)
12.	Cost Opening	11/4/09	10:00 am (PDT)
13.	Notice of Intent to Award	11/18/09	
14.	Last day to submit Notice of Intent to Protest Award	11/25/09	5:00 pm (PDT)

No.	Event/Action	Action Date	Time
15.	Contract Award	11/30/09	
16.	Start Date of Agreement	1/1/10	

Additional action dates may be inserted as needed.

#### 1.7 Americans with Disabilities Act

Following is the Policy of the Procurement Division of the State Department of General Services to Comply with Title II of the Americans with Disabilities Act (ADA) Compliance Policy of Nondiscrimination on the Basis of Disability.

To meet and carry out compliance with the nondiscrimination requirements of Title II, Americans with Disabilities Act (ADA), it is the policy of the Procurement Division (within the State Department of General Services) to make every effort to ensure that its programs, activities, employment opportunities, and services are available to all persons, including persons with disabilities.

For persons with a disability needing reasonable accommodation to participate in the Procurement process, or for persons having questions regarding reasonable accommodation for the Procurement process, please contact the Procurement Division at 916-375-4400 (main office). Procurement Division's TTY/TDD (telephone device for the deaf) and California Relay Service numbers are listed below. The bidder may also directly contact the Procurement Official identified in 1.5 above.

IMPORTANT: To ensure that the bidder's accommodation can be met, it is best that requests be submitted at least ten (10) working days before the scheduled event (i.e., bidder conference) or deadline due-date for procurement documents.

The Procurement Division TTY telephone number is:

Sacramento Office: (916) 376-1891

The California Relay Service Telephone Numbers are:

Voice: 1-800-735-2922 or 1-888-877-5379

TTY: 1-800-735-2929 or 1-888-877-5378

Speech to Speech: 1-800-854-7784

<sup>\*</sup> Although recommended, this key action date is an optional step in the RFP process.

## **SECTION 2**

## **RULES GOVERNING COMPETITION**

### 2. Rules Governing Competition

The purpose of competitive bidding is to secure public objectives in the most value-effective manner and avoid the possibilities of graft, fraud, collusion, etc. Competitive bidding is designed to benefit the State and is not for the benefit of the Bidders. It is administered to accomplish its purposes with sole reference to the public interest. It is based upon full and free bidding to satisfy State specifications, and acceptance by the State of the most value-effective solution to the State's requirements, as determined by the evaluation criteria contained in the Request for Proposal (RFP).

### 2.1 Identification and Classification of RFP Requirements

Section 2 of this RFP describes the entire procurement process. Specific guidelines for the submission of this RFP response are found in Section 8, Proposal Format.

### 2.1.1 Mandatory Requirements

The State<sup>1</sup> has established certain requirements with respect to proposals to be submitted by prospective Bidders<sup>2</sup>. The use of "shall," "must," or "will" in this RFP indicates a requirement or condition which is mandatory. A deviation, if not material, may be waived by the State. A deviation from a requirement is material if the response is not in substantial accord with the RFP requirements, provides an advantage to one Bidder over other Bidders, or has a potentially significant affect on the delivery, quantity or quality of items bid<sup>3</sup>, amount paid to the Bidder, or on the cost to the State. Material deviations cannot be waived.

#### 2.1.2 Desirable Items

The words "should" or "may" in this RFP indicate desirable attributes or conditions, but are non-mandatory in nature. Deviation from, or omission of, such a desirable feature, even if material will not in itself cause rejection of the bid.

### 2.2 Proposal Requirements and Conditions

#### 2.2.1 General

This RFP, the evaluation of responses, and the award of any resulting contract shall be made in conformance with current competitive bidding procedures as they relate to the procurement of goods and services by the State of California. A Bidder's Final Proposal is an irrevocable offer for 120 calendar days following the scheduled date for Final Proposal specified in Section 1.6, Key Action Dates. A Bidder may extend the offer in the event of a delay of contract award.

<sup>&</sup>lt;sup>1</sup> For the purposes of this RFP, "the State" references all participating WSCA states, unless otherwise indicated.

<sup>&</sup>lt;sup>2</sup> For the purposes of the instructions of this RFP, all entities that have identified their intent to be a Bidder to the Procurement Official are called Bidders until such time that the Bidder withdraws or other facts indicate that the Bidder has become nonparticipating.

<sup>&</sup>lt;sup>3</sup> The word "bid" as used throughout this document is intended to mean "proposed," "propose" or "proposal" as appropriate.

#### 2.2.2 RFP Documents

This RFP includes, in addition to an explanation of the State's needs which must be met, instructions which describe the format and content of bids to be submitted and the model(s) of the contract(s) to be executed between the State and the successful bidder(s). The State's contract development instructions have been identified in Exhibit 10.3, Contract Development/Standard Agreement (STD 213).

If a Bidder discovers any ambiguity, conflict, discrepancy, omission, or other error in this RFP, the Bidder shall immediately notify the Procurement Official identified in Section 1.5, Procurement Official, of such error in writing and request clarification or modification of the document.

Modifications will be made by addenda issued pursuant to Section 2.2.7, Addenda. Such clarifications shall be provided to all parties that have identified themselves as bidders to the Procurement Official for this RFP, without divulging the source of the request. Insofar as practicable, the State will give such notices to other interested parties, but the State shall not be responsible therefore.

If this RFP contains an error known to the Bidder, or an error that reasonably should have been known, the Bidder shall propose at its own risk. If the Bidder fails to notify the State of the error prior to the date fixed for submission of proposals, and is awarded the contract, the Bidder shall not be entitled to additional compensation or time by reason of the error or its later correction.

#### 2.2.3 Examination of the Work

The Bidder should carefully examine the entire RFP and any addenda thereto, and all related materials and data referenced in this RFP or otherwise available to the Bidder, and should become fully aware of the nature and location of the work, the quantities of the work, and the conditions to be encountered in performing the work. Specific conditions to be examined are listed in Section 5, Administrative Requirements and Section 6, Technical Requirements.

#### 2.2.4 Questions Regarding the RFP

Bidders requiring clarification of the intent or content of this RFP or on procedural matters regarding the competitive proposal process may request clarification by submitting questions, in an email or envelope clearly marked "Questions Relating to RFP 57224" to the Procurement Official listed in Section 1.5, Procurement Official. To ensure a response prior to submission of the proposals, questions must be received by the Procurement Official, in writing, by the scheduled date(s) as specified in Section 1.6, Key Action Dates. If a Bidder submits a question after the scheduled dates(s) the State will attempt to answer the question but does not guarantee that the answer will be prior to the proposal due date. Question and answer sets will be provided to all Bidders. The State will publish the questions as they are submitted including any background information provided with the question; however, the State at its sole discretion may paraphrase the question and background content for clarity.

A bidder who desires clarification or further information on the content of the RFP, but whose questions relate to the proprietary aspect of that bidder's proposal and which, if disclosed to other bidders, would expose that bidder's proposal, may submit such questions in the same manner as above, but also marked "CONFIDENTIAL," and not

later than the scheduled date specified in Section 1.6, Key Action Dates, to ensure a response. The bidder must explain why any questions are sensitive in nature. If the State concurs that the disclosure of the question or answer would expose the proprietary nature of the proposal, the question will be answered and both the questions and answer will be kept in confidence. If the State does not concur with the proprietary aspect of the question, the question will not be answered in this manner and the bidder will be so notified.

### 2.2.5 Request to Change the Requirements of the RFP

If the Bidder believes that one or more of the RFP requirements is onerous, unfair, or imposes unnecessary constraints on the Bidder in proposing less costly or alternate solutions, the Bidder may request a change to this RFP by submitting, in writing, the recommended change(s) and the facts substantiating this belief and reasons for making the recommended change. Such requests must be submitted to the Procurement Official, identified in Section 1.5, Procurement Official, by the date specified in Section 1.6, Key Action Dates, for requesting a change in the requirements.

#### 2.2.6 Bidders' Conference

A Bidders' Conference will be held, during which suppliers will be afforded the opportunity to meet with State personnel and discuss the content of the RFP and the procurement process. Suppliers attending the conference will benefit from the information provided, networking opportunity, and to be introduced to the Procurement Official identified in Section 1.5, Procurement Official. The time, date and location of the Bidders' Conference are included in Section 1.6, Key Action Dates.

Written questions received prior to the cutoff date for submission of such questions, as noted in Section 1.6, Key Action Dates, will be answered at the conference without divulging the source of the query. The State may also accept written or oral questions during the conference and will make a reasonable attempt to provide answers prior to the conclusion of the conference. A transcript of the discussion, or those portions which contain the questions and appropriate answers, will normally be transmitted within approximately ten (10) working days to all suppliers who have submitted an intention to bid and those who have submitted a written request to receive the information. Written requests are to be submitted to the Procurement Official identified in Section 1.5, Procurement Official. If questions asked at the conference cannot be adequately answered during the discussion, answers will be provided with the transcribed data. Oral answers shall not be binding on the State.

<u>Please Note</u>: Although attendance is not mandatory, it is strongly suggested all potential bidders plan on attending.

### 2.2.7 Addenda

The State may modify the RFP prior to the date fixed for Submission of Final Proposal by issuance of an addendum to all Bidders that are participating in the procurement process at the time the addendum is issued. If the amendments are such as to offer the opportunity for non-participating Bidders to become participating, the addendum will be sent to all Bidders that have identified their intent to be a Bidder to the Procurement Official identified in Section 1.5, Procurement Official.

Addenda will be numbered consecutively. Addenda will be posted on the eProcurement website. It is the responsibility of the bidder to view the website for any updates to the RFP. If any Bidder determines that an addendum unnecessarily restricts its ability to propose, the Bidder is allowed five (5) working days to submit a protest to the addendum according to the instructions contained in Section 2.4.1, Protests.

#### 2.2.8 State of California Plastic Trash Bag Certification Violations

Public Resources Code Section 42290 et seq. prohibits the State from contracting with any supplier, manufacturer, or wholesaler, and any of its divisions, subsidiaries, or successors that have been determined to be noncompliant to the recycled content plastic trash bag certification requirements. This includes award of a State contract or subcontract or renewal, extension, or modification of an existing contract or subcontract. Prior to award, the State shall ascertain if the intended awardee or proposed subcontractor is a business identified on the current California Integrated Waste Management Board noncompliant list(s). In the event of any doubt of the status or identity of the business in violation, the State will notify the Board of the proposed award and afford the Board the opportunity to advise the State. No award will be made when either the Bidder or a subcontractor has been identified either by published list or by advice from the Board, to be in violation of certification requirements.

#### 2.2.9 **Bonds**

If a performance bond is required for this procurement, it will be specified in Section 5, Administrative Requirements.

#### 2.2.10 Follow-on Contracts (PCC 10365.5, PCC 10410, and PCC 10411)

Not Applicable

### 2.2.11 Prompt Payment Discounts

In connection with any discount offered, except when provision is made for a testing period preceding acceptance by the State, time will be computed from date of delivery of the supplies or equipment as specified, or from date correct invoices are received in the office specified by the State if the latter date is later than the date of delivery. When provision is made for a testing period preceding acceptance by the State, date of delivery shall mean the date the supplies or equipment are accepted by the State during the specified testing period. Payment is deemed to be made, for the purpose of earning the discount, on the date of mailing the State warrant or check.

Cash discounts offered by Bidders for the prompt payment of invoices will not be considered in evaluating offers for award purposes; however, all offered discounts will be taken if the payment is made within the discount period, even though not considered in the evaluation of offers.

### 2.2.12 Joint Proposals

A joint proposal (two or more Bidders quoting jointly on one proposal) may be submitted and each participating Bidder must sign the joint proposal. If the contract is awarded to joint Bidders, it shall be one indivisible contract.

Each joint bidder will be jointly and severally responsible for the performance of the entire contract, and the joint Bidders must designate, in writing, one individual having authority to represent them in all matters relating to the contract. The State assumes no responsibility or obligation for the division of orders or purchases among joint contractors.

#### 2.2.13 State of California Air or Water Pollution Violations

For the State of California, unless the contract is less than \$25,000 or with a non-competitively bid contractor, Government Code Section 4477 prohibits the State from contracting with a person, including a corporation or other business association, who has been determined to be in violation of any state or federal air or water pollution control law.

Prior to an award, the State shall ascertain if the intended awardee is a person included in notices from the Boards. In the event of any doubt of the intended awardees identity or status as a person who is in violation of any state or federal air or water pollution law, the State will notify the appropriate Board of the proposed award and afford the Board the opportunity to advise the Department that the intended awardees is such a person.

No award will be made to a person who is identified either by the published notices or by advice, as a person in violation of state or federal air or water pollution control laws.

#### 2.2.14 State of California Fair Employment and Housing Commission Regulations

Not applicable to this RFP

#### 2.2.15 Exclusion for Conflict of Interest

Not Applicable to this RFP

## 2.2.16 State of California Seller's Permit

For the State of California, this RFP is subject to all requirements set forth in Sections 6452, 6487, 7101 and 18510 of the Revenue and Taxation Code, and Section 10295.1 of the Public Contract Code. To sell product to the State of California, the supplier is required to provide a copy of their retailer's seller's permit or certification of registration, and, if applicable, the permit or certification of all participating affiliates issued by the State of California's Board of Equalization. See Section 5.4.9, Seller Permit, for information on the retailer's seller permit requirement. All other participating States will have their own requirements for its independent State.

#### 2.2.17 Disclosure of Financial Interests

Not Applicable

#### 2.2.18 State of California Unfair Practices Act and Other Laws

For the State of California, the bidder warrants that its proposal complies with the Unfair Practices Act (Business and Professions Code Section 17000 et seq.) and all applicable State and Federal laws and regulations.

### 2.3 Bidding Steps

### 2.3.1 Preparation of Proposals

Competitive Bidding and Proposal Responsiveness (Exhibit 10.4), emphasizes the requirements of competitive bidding and contains examples of common causes for rejection of proposals. Bidders are encouraged to review this exhibit.

Proposals are to be prepared in such a way as to provide a straightforward, concise delineation of the Bidder's compliance with the requirements of this RFP. Expensive bindings, colored displays, promotional materials, etc., are not necessary or desired. Emphasis should be concentrated on conformance to the RFP instructions, responsiveness to the RFP requirements, and on completeness and clarity of content.

Before submitting each document, the Bidder should carefully read their Proposal for errors and adherence to the RFP requirements.

### 2.3.2 Bidder's Intention to Submit a Proposal

Suppliers that want to receive the latest updates are encouraged to submit the attached letter for Intention To Bid (ITB) (see Exhibit 10.1) by the date specified in Section 1.6, Key Action Dates. The letter should identify the contact person, their email address, phone number and fax number for the RFP process. The State is also interested as to a supplier's reasons for not submitting a proposal; for example, requirements that cannot be met or unusual terms and conditions which arbitrarily raise costs.

The State is responsible for notifying one (1) contact person per Bidder. Those suppliers that submit their ITB as specified will continue to receive additional correspondence throughout this procurement. It shall be the Bidder's responsibility to immediately notify the Procurement Official identified in Section 1.5, Procurement Official, in writing, regarding any revision to the information. The State shall not be responsible for proposal correspondence not received by the Bidder if the Bidder fails to notify the State, in writing, about any change pertaining to the designated contact person.

If suppliers have indicated significant problems with the RFP requirements, the State will examine the stated reasons for the problems and will attempt to resolve any issues in contention, if not contrary to the State's interest, and may amend the RFP if appropriate. All suppliers who have submitted an ITB will be advised by the State of any actions taken as a result of suppliers' responses. If after such actions, a supplier determines that the requirements of the RFP unnecessarily restrict its ability to bid, the supplier is allowed five (5) working days to submit a protest to those RFP requirements or the State's action, according to the instructions contained in Section 2.4.1, Protests.

Should a Bidder not participate in a RFP step, the State reserves the right to drop them from the participating Bidder list and they may not receive any further correspondence until they contact the Procurement Official identified in Section 1.5, Procurement Official, to indicate that they would like further correspondence.

## 2.3.3 Draft Proposals

Submission of a Draft Proposal is strongly recommended so that the Bidder may obtain the benefit of the two-step procurement process. However, Bidders are not required to submit a Draft Proposal. If Draft Proposals are submitted, Bidders must submit them by the date and time specified in Section 1.6, Key Action Dates. The Draft Proposals must be complete in all respects except that dollar cost information **must be replaced by XXXs**. At the sole discretion of the State, the State Evaluation Team shall evaluate each Draft Proposal received by the Draft Proposal submission date specified in Section 1.6, Key Action Dates, unless the Draft Proposal is so defective that the State does not believe that its time to evaluate the Draft Proposal would result in the Bidder being able to correct the deficiencies prior to the Final Proposal due date, or the Bidder has placed conditions on the Draft Proposal that are unacceptable to the State.

For Draft Proposals that are evaluated the Bidder will be notified of any defects the State has noted. Notifying the Bidder of defects is intended to minimize the risk that the Final Proposal will be deemed non-compliant; however, the State will not provide any warranty that the Draft Proposal will be evaluated even if accepted for review, or that any or all defects in the Draft Proposal have been detected. Notification of defects in the Draft Proposal will not preclude rejection of the Final Proposal, if undiscovered defects contained in the Draft Proposal are later found in the Final Proposal. Draft proposals received late may be reviewed if the Procurement Official believes there is enough time and resources to do so.

#### 2.3.4 Confidential Discussions

The State may conduct confidential discussions with Bidders submitting Draft Proposals that have been reviewed by the State. At the confidential discussion, the State will identify areas of the Bidder's Draft Proposal that may not be fully compliant with the requirements of the RFP and areas that are confusing to the State Evaluation Team. Oral statements made by either party during confidential discussions shall not be binding.

### 2.3.5 Final Proposal

### 2.3.5.1 Submission of Final Proposal

Proposals must be complete in all respects as required by Section 8, Proposal Format. A Final Proposal may be rejected if it is conditional or incomplete, or if it contains any alterations of form or other irregularities of any kind. A Final Proposal shall be rejected if any such defect or irregularity constitutes a material deviation from the RFP requirements. The Final Proposal must contain all costs required by Section 7, Cost, and Section 8, Proposal Format. Cost data, including any electronic copies (as identified in Section 8, Proposal Format) must be submitted under separate, sealed cover. If the cost data are not submitted under separate sealed cover, the proposal may be rejected. This section describes specific guidelines applicable to the submission of the Final Proposal to the RFP and is applicable to the new Final Proposal if all original Final Proposals are declared flawed by the Evaluation Team.

#### 2.3.5.2 Bidder's Costs

Costs for developing Proposals are entirely the responsibility of the Bidder and shall not be chargeable to the State.

### 2.3.5.3 Proposal Responsiveness

Competitive Bidding and Proposal Responsiveness (Exhibit 10.4), emphasizes the requirements of competitive bidding and contains examples of common causes for rejection of proposals. Bidders are encouraged to review this exhibit.

#### 2.3.5.4 False or Misleading Statements

Proposals which contain false or misleading statements or which provide references which do not support an attribute or condition claimed by the Bidder may be rejected. If, in the sole opinion of the State, such information was intended to mislead the State in its evaluation of the Proposal and the attribute, condition, or capability is a requirement of this RFP, it will be the basis for rejection of the Proposal.

### 2.3.5.5 Proposal Signature

A cover letter (which shall be considered an integral part of the submission) shall be signed by an individual who is authorized to bind the bidding firm contractually.

The signature block must indicate the title or position that the individual holds in the firm. An unsigned Final Proposal shall be rejected.

If a Draft Proposal is submitted it must also contain the cover letter, similarly prepared, including the title of the person who will sign, but need not contain the signature.

### 2.3.5.6 Delivery of Proposals

The Draft Proposal and Final Proposal must be submitted no later than the date and time specified in Section 1.6, Key Action Dates. If mailed, proposals must be received by the Department of General Services on or before the specified date and time. Mail or deliver proposals to the Department of General Services Procurement Official listed in Section 1.5, Procurement Official. If mailed, it is suggested that the bidder use certified or registered mail with return receipt requested as delivery of proposals is done at the Bidder's own risk of untimely delivery, lost mail, etc.

Proposals must be received in the number of copies stated in Section 8, Proposal Format. Proposals must be received no later than the dates and times in Section 1.6, Key Action Dates. One (1) copy must be clearly marked "Master Copy". All copies of proposals must be under sealed cover, which is to be plainly marked "FINAL PROPOSAL for RFP DGS-57224". Final Proposals not received by the date and time in Section 1.6, Key Action Dates, shall be rejected.

As required in Section 8, Proposal Format, all cost data must be submitted under separate, sealed cover and clearly marked "COST DATA for RFP DGS-57224". If cost data is not submitted separately sealed, the proposal may be rejected. Proposals submitted under improperly marked covers may be rejected. If discrepancies are found between two or more copies of the proposal, the proposal may be rejected. However, if not rejected, the Master Hard Copy is the basis for resolving discrepancies.

### 2.3.5.7 Withdrawal and Re-submission/Modification of Proposals

A Bidder may withdraw its Final Proposal at any time prior to the final proposal submission date and time in Section 1.6, Key Action Dates, by submitting a written notification of withdrawal signed by the Bidder authorized in accordance with Section 2.3.5.5, Proposal Signature. The Bidder may thereafter submit a new or modified proposal prior to such final proposal submission date and time.

Modification offered in any other manner, oral or written, will not be considered. Clarifications of Final Proposals shall not be considered to be modifications. Final Proposals cannot be changed or withdrawn after the date and time designated for receipt, except as provided in Section 2.3.8.3, Errors in the Final Proposal.

### 2.3.5.8 Clarification of Final Proposals

During the evaluation of the Final Proposal, the State Evaluation Team may ask the Bidder to clarify their submitted information but will not allow the Bidder to change their proposal.

### 2.3.5.9 Flawed Final Proposals

At the State's sole discretion it may declare all Final Proposals to be Draft Proposals in the event that the State Evaluation Team determines that Final Proposals from all Bidders contain material deviations. A bidder may not protest the evaluation team's determination that all proposals have material deviations. If all proposals are declared noncompliant, the State will issue an addendum to the RFP and confidential discussions will be held with Bidders who are interested in submitting a Final Proposal. Each Bidder will be notified of the due date for the submission of a new Final Proposal to the State. This submission must conform to the requirements of the original RFP, and as amended by any subsequent addenda. The new Final Proposals will be evaluated as required by Section 9, Evaluation.

## 2.3.5.10 Confidentiality

Bidders should be aware that marking any portion of a Final or Draft Proposal as "confidential", "proprietary" or "trade secret" shall exclude it from evaluation or consideration for award, unless specifically approved by the State in writing prior to submission of the Final or Draft Proposal. Even providing such Proposal is accepted for evaluation by the State; such markings in a Proposal will not keep that document, after notice of intent to award, from being released as part of the public record, unless a court of competent jurisdiction has ordered the State not to release the document.

The content of all working papers and discussions relating to the Bidder's Draft Proposal, if retained, shall be held in confidence indefinitely unless the public interest is best served by an item's disclosure because of its direct pertinence to a decision or the evaluation of the proposal.

Any disclosure of State confidential information by the Bidder is a basis for rejecting the Bidder's proposal and ruling the Bidder ineligible to further participate.

Maintaining the confidentiality of information that is designated as confidential by the State is paramount; this cannot be over emphasized.

### 2.3.6 Demonstrations (If required)

Not applicable

### 2.3.7 Sealed Cost Openings

Final Proposals will not have their sealed cost envelopes opened until the State has evaluated the technical and administrative submission. Cost Proposals will only be opened for responsive proposals from responsible Bidders. On the date of the cost opening, the State will post a summary of the evaluation of each Bidder. This summary will be provided to all the Bidders and the public in attendance at the cost opening as well as to persons who request the summary.

### 2.3.8 Rejection of Proposals

The State may reject any or all proposals and may waive any immaterial deviation or defect in a proposal. The State's waiver of any immaterial deviation or defect shall in no way modify the RFP documents or excuse the Bidder from full compliance with the RFP specifications if awarded the contract.

#### 2.3.8.1 General

Final Proposals will be evaluated according to the procedures contained in Section 9, Evaluation.

#### 2.3.8.2 Evaluation Questions

During the evaluation and selection process, the State may require a Bidder's representative to answer specific questions and provide clarification, as long as the answers and clarification are in writing.

### 2.3.8.3 Errors in the Final Proposal

Errors in the Final Proposal may cause the rejection of that proposal. However, the State, in its sole discretion, may retain the proposal and make certain corrections. In determining if a correction will be made, the State will consider the conformance of the proposal to the format and content required by the RFP and any unusual complexity of the format and content required by the RFP.

The State, at its sole discretion, may correct obvious clerical errors.

The State, in its sole discretion, may correct discrepancies and arithmetic errors. If intent is not clearly established by the complete proposal submittal, the Master Hard Copy shall have priority over additional copies. The Proposal Master Hard Copy narrative shall have priority over the cost sheets. If necessary, the extensions and summary will be re-computed from the lowest level of detail, even if the lowest level of detail is obviously misstated. If a Bidder does not follow the instructions for computing costs which are not related to the contract (e.g., State personnel costs), the State may reject the proposal, or at its sole discretion, re-compute such costs based on instructions contained in the RFP.

The State may at its sole discretion correct errors of omission. In the following three situations, the State may take the actions identified below if the Bidder's intent (as determined by the State) is not clearly established by the complete proposal submittal.

- If an item is described in the narrative and omitted from the cost data provided in the proposal, for evaluation purposes it will be interpreted to mean that the item will be provided by the Bidder at no cost.
- 2) If an item is not mentioned at all in the proposal, the proposal will be interpreted to mean that the Bidder does not intend to supply that item.
- If an item is omitted, and the omission is not discovered until after contract award, the Bidder shall be required to supply that item at no cost.

The determination of whether an item is minor or major will be the responsibility of the State.

## It is absolutely essential that Bidders carefully review the cost elements in their Final Proposal.

In the event an ambiguity or discrepancy is detected between the general requirements described in Section 3, Current System, and the specific functional and technical requirements set forth in Section 6, Technical Requirements after the opening of proposals, then Section 6, Technical Requirements, and the Bidder's response thereto, shall have priority over Section 3, Current System and the Bidder's response thereto.

#### 2.3.9 Contract Award

If a contract is awarded, the award will be in accordance with Section 9, Evaluation, to a responsible Bidder whose Final Proposal complies with the requirements of the RFP and any addenda thereto, except for such immaterial defects as may be waived by the State. Award, if made, will be made within 120 calendar days after the scheduled date for the Contract Award as specified in Section 1.6, Key Action Dates, unless a protest is received. If a protest is received, the Award, if made, will be made within 120 calendar days after the protest is resolved.

The State reserves the right to modify or cancel the RFP in whole or in part prior to contract award.

#### 2.3.10 Debriefing

At the request of any bidder after contract award, a debriefing will be held for the purpose of receiving specific information concerning the evaluation. The discussion will be based primarily on the technical and cost evaluations of the Bidder's Final Proposal. A debriefing is not the forum to challenge the RFP specifications or requirements.

#### 2.4 Other Information

#### 2.4.1 Protests

Before a protest is submitted regarding any issue other than selection of the "successful Bidder," the Bidder must make full and timely use of the procedures described in Section 2.2.4, Questions Regarding the RFP, and Section 2.2.5, Request to Change the Requirements of the RFP, to resolve any outstanding issue(s) between the Bidder and the State. This procurement procedure is designed to give the Bidder and the State adequate opportunity to submit questions and discuss the requirements before the Final Proposal is due. The protest procedure is made available in the event that a Bidder cannot reach a fair agreement with the State after exhausting these procedures. There are two types of protests:

- 1) Requirements protests
- 2) Award protests

A protest shall be submitted according to the procedure below.

### 2.4.1.1 Requirements Protest

Protests regarding any issue other than selection of the "successful bidder" are requirements protests and will be heard and resolved by the Deputy Director of the Department of General Services, Procurement Division, whose decision will be final.

All protests must be made in writing, signed by an individual authorized under Section 2.3.5.5, Proposal Signature, and contain a statement of the reason(s) for protest; citing the law, rule, regulation or procedures on which the protest is based. The protester must provide facts and evidence to support the claim.

Requirements Protests must be mailed or delivered to:

Street Address:
Deputy Director
Procurement Division
707 Third Street, 2nd Floor
West Sacramento, CA 95605

Mailing Address:
Deputy Director
Procurement Division
P.O. Box 989052
Sacramento, CA 95798-9052

All protests to the RFP requirements or procedures must be received by the Deputy Director of the Procurement Division as promptly as possible, but not later than the respective time and date specified in Section 1.6, Key Action Dates.

### 2.4.1.2 Award Protest

Protests concerning the evaluation, recommendation, or other aspects of the selection process must be received by the Deputy Director of the Procurement Division as promptly as possible, but not later than the respective time and date specified in Section I for such protests or the respective date of the Notification of Intent to Award, whichever is later.

The bidder will be allowed five (5) working days to submit a protest to the Intent to Award, according to the instructions contained in this section. Certified or registered mail should be used unless delivered in person, in which case the protester should obtain a receipt of delivery.

All protests must be made in writing, signed by an individual authorized under **Section 2.3.5.5**, **Proposal Signature**, and contain a statement of the reason(s) for protest; citing the law, rule, regulation or procedures on which the protest is based. The protester must provide facts and evidence to support the claim. Protests must be mailed or delivered to:

#### Street Address:

Deputy Director Procurement Division 707 Third Street, 2nd Floor West Sacramento, CA 95605

#### **Mailing Address:**

Deputy Director Procurement Division P.O. Box 989052 Sacramento, CA 95798-9052

#### 2.4.2 News Releases

News releases or any publications relating to a contract resulting from this RFP shall not be made without prior written approval of the Procurement Official identified in Section 1.5, Procurement Official.

### 2.4.3 Disposition of Proposals

All materials submitted in response to this RFP are the property of the State of California and will be returned only at the State's option and at the Bidder's expense. At a minimum, the Master Copy of the Final Proposal shall be retained for official files and will become a public record after the Notification of Intent to Award is issued, as specified in Section 1.6, Key Action Dates. However, materials the State considers in its sole opinion to be confidential information (such as confidential financial information submitted to show bidder responsibility) will be returned upon the request of the Bidder.

## SECTION 3 Current System

## **SECTION 3**

## **CURRENT SYSTEM**

9-16-09 3 - 1

## SECTION 3 Current System

## 3. Current System

The DGS-PD is the central purchasing and materials management agency for the State of California. Currently there is no master agreement in place for the purchase of Automotive Parts. California state agencies are now purchasing Auto Parts from various suppliers throughout the State. The State of California has no historical spend data for Automotive Parts. Other participating States may or may not have a contract in place. Below are a list of participating States and their estimated annual spend of the contract:

### **Estimated Annual Spend Data Per Participating State:**

Alaska \$1,600,000.00

Arizona Unknown

California Unknown

Colorado Unknown

Hawaii \$ 40,000.00

Idaho Unknown

Louisiana Unknown

Maine \$ 1,500,000.00

Minnesota \$ 2,950,000.00

Nebraska Unknown

Nevada \$ 150,000.00

New Hampshire \$300,000.00

North Dakota \$8,000.00

Oregon \$4,000,000.00

South Dakota Unknown

Tennessee Unknown

Utah \$ 200,000.00

Vermont \$50,000.00

9-16-09 3 - 2

# SECTION 4 Proposed System

## **SECTION 4**

## **PROPOSED SYSTEM**

9-16-09 4 - 1

## SECTION 4 Proposed System

## 4. Proposed System

#### 4.1 Overall Goal

The overall goal when purchasing Auto Parts is to take advantage of volume purchasing to leverage the State's buying power while reducing environmental impacts that may result from products or packaging through their life cycles.

The State of California is proposing to establish a master agreement with multiple suppliers for the purchase of aftermarket light duty automotive parts. In addition, this agreement will be made available to any participating Western States Contracting Alliance (WSCA) state.

### 4.2 State of California Environmentally Preferable Purchasing

The State of California, Department of General Services, Procurement Division is responsible for the implementation of Environmentally Preferable Purchasing (EPP) as mandated by AB 498 (2001, Chan), enacted in the California Public Contract Code, Chapter 6, Sections 12400-12404, and supported by the Governor's Executive Order S-20-04.

The procurement and use of environmentally preferable products and services can result in a profound improvement in human health and a positive impact on the environment. The EPP program objective is "the procurement or acquisition of goods and services that have a lesser or reduced effect on human health and the environment when compared with competing goods and services that serve the same purpose". It is the intent of the State of California to integrate human health and environmental considerations into all procurement processes by considering the full life cycle impact of our procurement of those goods and services in order to achieve maximum overall health and environmental benefits.

Through the EPP program, the State of California promotes the procurement of products that demonstrate Extended Producer Responsibility (EPR); incorporate post-consumer recycled content; are durable and reusable and can be remanufactured; conserve energy and water, use resources from sustainably harvested forests; use low impact, renewable bio-based materials; reduce volatile organic compound (VOC) emissions; reduce greenhouse gas and ozone-depleting emissions; use totally or process chlorine-free manufacturing processes; are lead-free and mercury-free; use minimal, reusable, biodegradable packaging material; employ clean fuel distribution systems; and otherwise protect our health and our environment.

We consider all health and environmental factors as early as possible in our acquisition planning and decision-making processes, and ensure that the responsibility for EPP is shared among program, acquisition, and procurement personnel. It is the EPP program personnel's responsibility to ensure that individual product standards are clear and reasonable in order to secure a market-best value to the State through a competitive process.

9-16-09 4 - 2

## SECTION 4 Proposed System

In making specific procurement decisions, we give careful consideration to the following during evaluation

- Raw Materials: Are the materials used sustainable and renewable; do they
  incorporate maximum post consumer recycled content; do they incorporate
  resources from sustainably harvested forests and/or renewable bio-based
  materials that efficiently and minimally impact the environment including land,
  water, air and worker and community health; and are they regionally or
  domestically sourced?
- The Manufacturing Processes: Are the manufacturing processes resource efficient including energy, materials and water; do they use renewable or green power energy sources; are they toxin and pollutant free; how do they process their waste stream; to what extent do they incorporate recycled content; how do they package their products for shipment; do they have an ISO 14001 or equivalent environmental management plan in place and do they have a tracking and reporting system for that plan?
- The Distribution Process: Are products regionally sourced and manufactured and are they shipped and delivered by efficient, clean-fuel carriers using scheduled distribution patterns?
- Product Operation and Use: Are purchased goods Life Cycle Cost (LCC)
  effective compared to alternatives, do they have Life Cycle Impact Assessment
  (LCIA) declarations, are they warranted or guaranteed, are they supported and
  serviced over their useful life, does their operation and use require the use of or
  result in the production of any toxic or polluting substances, do they meet all
  required performance characteristics, and are they durable, safe, and energy
  efficient?
- The Product Disposal Process: Does the supplier have an EPR program? Can
  the product be reused or re-manufactured; is there a product buy-back or recycle
  program at the end of the useful product life; and to what extent are products
  recyclable and/or biodegradable?

9-16-09 4 - 3

## **SECTION 5**

## **ADMINISTRATIVE REQUIREMENTS**

### 5. Administrative Requirements

#### 5.1 Introduction

In addition to meeting the technical requirements of this RFP, Bidders must adhere to all the administrative requirements of this RFP to be responsive. These include:

- The schedule and other requirements as specified in Section 1, Introduction and Overview of Requirements;
- The rules as specified in Section 2, Rules Governing Competition;
- The administrative requirements of Section 5, Administrative Requirements;
- The completion of cost sheets as specified in Section 7, Cost;
- The format as specified in Section 8, Proposal Format; and

### 5.2 Responses to Administrative Requirements

All of the requirements described in this section are service levels and/or terms and conditions that must be satisfied by the Bidders. Bidders must indicate their willingness and ability to satisfy these requirements.

Responses to the requirements in this section must be included in Bidder's Proposal, Volume 1 (Refer to Section 8, Proposal Format). Exhibit 10.5, Administrative Requirements Response, must be completed and submitted with the Bidder's proposal as specified in Section 8, Proposal Format. The Bidder's proposal must include the required proposal items cross-referenced by the Administrative Requirement number. It may also include any appropriate exhibits to illustrate responsiveness. These exhibits must be clearly labeled and cross-referenced to the proposal narrative by Administrative Requirement number.

#### 5.3 Designation of Requirements

The requirements specified in this RFP Section are classified as either "Mandatory" or "Non-Mandatory".

#### **Mandatory Requirements (M):**

All items labeled "Mandatory" or "M" are not negotiable. A Pass/Fail evaluation will be utilized for all mandatory requirements in this RFP. To be considered responsive, responsible and eligible to participate in the RFP, all questions identified as mandatory must be marked "Yes" to pass.

As specified with each mandatory requirement listed below, Bidders must:

- Indicate whether their proposal meets the individual requirement by marking either a "Yes" or "No" on Exhibit 10.5, Administrative Requirements Response; and
- Provide a narrative of the Bidder's compliance and/or response for each item.

### **Non-Mandatory Requirements (NM):**

Some listed standards are identified as "Non-Mandatory" or "NM". Bidders are not required to comply with these requirements in order to be compliant with the RFP requirements. However, if a Bidder offers any of these non-mandatory requirements, the Bidder must meet the minimum requirements at stated in this section. Non-Mandatory items will not be considered when identifying Bidders who are responsive and responsible.

As specified with each non-mandatory requirement listed below, Bidders must:

- Indicate whether their proposal meets the individual requirement by marking either a "Yes" or "No" on Exhibit 10.5, Administrative Requirements Response; and
- Provide a narrative of the Bidder's compliance and/or response for each item.

### 5.4 Administrative Requirements

### 5.4.1 Bidder Responsibility (M)

Prior to award of the contract, the State must be assured that the Bidder(s) selected has all of the resources to successfully perform under the contract. This includes, but is not limited to, personnel in the numbers and with the skills required, equipment of appropriate type and in sufficient quantity, financial resources sufficient to complete performance under the contract, and experience in similar endeavors. If, during the evaluation process, the State is unable to assure itself of the Bidder's ability to perform under the contract, if awarded, the State has the option of requesting from the Bidder any information that the State deems necessary to determine the Bidder's responsibility. If such information is required, the Bidder will be so notified and will be permitted approximately five (5) working days to submit the information requested.

The Bidder must provide the State with sufficient information to allow the State to confirm the Bidder's ability to perform successfully under the contract.

#### 5.4.2 Confidentiality (M)

Not Applicable

### 5.4.3 Contractor Responsibility (M)

The State requires that the contract resulting from RFP DGS-57224 be issued to multiple contractors who shall be responsible for successful performance of this RFP. The prime contractor shall be responsible for successful performance of any subcontractors and will be required to assume responsibility for design, development, testing, and implementation of all systems in their proposal.

Furthermore, the State will consider the prime contractors to be the sole point of contact with regard to contractual matters, payment of any and all charges resulting from the outsource or purchase of the equipment and maintenance of the equipment for the term of the contract.

The contractors will be responsible for compliance with requirements under the contract, even if requirements are delegated to subcontractors. All State policies, guidelines, and requirements apply to subcontractors. The contractors and subcontractors shall not in any way represent themselves in the name of the State of California without prior written approval.

### 5.4.4 Contractor Name Change (M)

An amendment is required to change the contractor's name as listed on this agreement. Upon receipt of legal documentation of the name change the State will process the amendment. Payment of invoices presented with a new name cannot be paid prior to approval of said amendment.

### 5.4.5 Bidder Declaration (M)

All bidders must complete the Bidder's Declaration form. When completing the declaration, bidders must identify all sub-contractors proposed for participation in the contract. Bidders awarded a contract are contractually obligated to use the sub-contractors for the corresponding work identified, unless the State agrees to a substitution.

At the State's option prior to award, bidders may be required to submit additional written clarifying information. Failure to submit the requested written information may be grounds for bid rejection.

Refer to <a href="http://www.documents.dgs.ca.gov/pd/delegations/GSPD105.pdf">http://www.documents.dgs.ca.gov/pd/delegations/GSPD105.pdf</a> for a downloadable version of this document.

### 5.4.5.1 Subcontractors (M)

If the Bidder's proposal involves the use of subcontractors for a portion of the contract pertaining to a California Certified Small Business or a California Certified Disabled Veteran, the Bidder must explain how they will manage and control the work of those subcontractors.

The Bidder must agree that all requirements of those subcontractors will be adhered to and that requirements will apply to subcontractors even if subcontractor concurrence is not specifically defined in the Administrative Requirements. All subcontractors must comply fully with the administrative and technical requirements that are applicable with the portion of the work being delegated to the subcontractor.

Bidders awarded a contract are contractually obligated to use those subcontractors for the corresponding work identified unless the State agrees to a substitution and it is incorporated by amendment to the contract. The contractor must have written agreement from the State prior to replacement or substitution of any subcontractor.

### 5.4.5.2 Notice of Subcontractors (M)

Upon award to a contractor requesting Small Business or DVBE participation, notice shall be given by the State to the California Certified Small Business or a California Certified Disabled Veteran listed in Section 5.4.5, Bidder Declaration Form, of their participation in the contract. Notification to the subcontractor by the prime contractor is encouraged immediately after award of an RFP.

There will be no assignment of responsibility to these subcontractors without prior written approval from the State of the Small Business or DVBE.

### 5.4.6 Commercially Useful Function (M)

Suppliers, whether the Bidder or a subcontractor, who have a California certification for one or more of the socio-economic programs (e.g., Small Business or Disabled Veteran Business Enterprise), must perform a commercially useful function (CUF) in the resulting contract. CUF is defined in the Military and Veterans Code Section 999(b)(5)(B) for Disabled Veteran Business Enterprises and in the Government Code Section 14837(d)(4)(A) for Small Business. Bidders must describe compliance with CUF requirement.

At the State's option, Bidders may be required to submit additional written clarification regarding CUF. Failure to submit the requested written information as specified may be grounds for bid rejection.

#### 5.4.7 Non-Collusion Affidavit (M)

Not applicable to this RFP

#### 5.4.8 Payee Data Record (M)

All Bidders must complete a Payee Data Record, and include it with the proposal. A copy of this form is available for download at <a href="http://www.documents.dgs.ca.gov/osp/pdf/std204.pdf">http://www.documents.dgs.ca.gov/osp/pdf/std204.pdf</a>.

#### 5.4.9 State of California Seller's Permit (M)

This RFP is subject to all requirements set forth in Sections 6452, 6487, 7101 and 18510 of the Revenue and Taxation Code, and Section 10295 of the Public Contract Code, requiring suppliers to provide a copy of their retailer's seller's permit or certification of registration, and, if applicable, the permit or certification of all participating affiliates issued by the State of California's Board of Equalization.

Bidders are required to submit a copy of their company's Retailer's Seller's Permit, and include it with the proposal.

### 5.4.10 State of California Darfur Contracting Act of 2008 (M)

Effective January 1, 2009, all State of California Invitations for Bids (IFB) or Requests for Proposals (RFP) for goods or services must address the requirements of the Darfur Contracting Act of 2008 (Act). (Public Contract Code sections 10475-10481). See <a href="http://www/leginfo.ca.gov/cgi-bin/displaycode?section=pcc&group=10001-11000&file=10475-10481">http://www/leginfo.ca.gov/cgi-bin/displaycode?section=pcc&group=10001-11000&file=10475-10481</a> for further clarification.

The Public Contract Code section 10478 (a) required a company that currently has (or within the previous three years has had) business activities or other operations outside of the United States to certify that it is not a "scrutinized" company when it submits a bid or proposal to a State agency. (Bidder must complete and return Exhibit 10.12 (Darfur Contracting Act Form) with the bidder's submittal.)

#### 5.4.11 Contract Terms (M)

Bidder(s) shall agree to accept all of the terms and conditions identified in Exhibit 10.3, Contract Development/Standard Agreement (STD 213), Exhibit 10.15, WSCA Terms and Conditions, including:

- General Provisions Non-IT Commodities (GSPD-401) http://www.documents.dgs.ca.gov/pd/modellang/GPnonIT0407.pdf
- Bidders Instructions, Non-IT Commodities (GSPD 451)
   <a href="http://www.documents.dgs.ca.gov/pd/modellang/BidderInstructions080307.p">http://www.documents.dgs.ca.gov/pd/modellang/BidderInstructions080307.p</a>
   <a href="http://www.documents.dgs.ca.gov/pd/modellang/BidderInstructions080307.p">http://www.documents.dgs.ca.gov/pd/modellang/BidderInstructions080307.p</a>
   <a href="http://www.documents.dgs.ca.gov/pd/modellang/BidderInstructions080307.p">http://www.documents.dgs.ca.gov/pd/modellang/BidderInstructions080307.p</a>
   <a href="http://www.documents.dgs.ca.gov/pd/modellang/BidderInstructions080307.p">http://www.documents.dgs.ca.gov/pd/modellang/BidderInstructions080307.p</a>
  </a>
- Supplemental Terms and Conditions using ARRA Funds dated 8/10/09 <a href="http://www.documents.dgs.ca.gov/pd/poliproc/ARRATand%20C081009final.p">http://www.documents.dgs.ca.gov/pd/poliproc/ARRATand%20C081009final.p</a> df

#### 5.4.12 Non-Discrimination Toward WTO GPA Signatories (M)

This RFP is subject to the requirements of the World Trade Organization (WTO) Government Procurement Agreement (GPA). All bidders offering products or services of countries that are signatories to the WTO GPA and that have agreed to cover reciprocal sub-central coverage under the WTO GPA will be accorded non-discriminatory treatment in the award of contracts under this solicitation. A list of countries is available at <a href="http://www.wto.org/english/tratop\_E/gproc\_e/memobs\_e.htm">http://www.wto.org/english/tratop\_E/gproc\_e/memobs\_e.htm</a>

### 5.4.13 California Disabled Veteran Business Enterprise (DVBE) Program Requirement

For this RFP, DVBE has been waived.

### 5.4.14 State of California DVBE Declaration (M)

Bidders who have been certified by California as a DVBE (or bidders who have obtained the participation of subcontractors certified by California as a DVBE) must complete the DVBE Declaration Form (STD 843) and include it with the proposal. A copy of this form can be downloaded at

http://www.documents.dgs.ca.gov/pd/poliproc/STD-

<u>843FillPrintFields.pdf#search=STD%20843&view=FitH&pagemode=none</u>. All disabled veteran owners and disabled veteran managers of the DVBE(s) must sign this form.

#### 5.5 State of California Preferences and Incentives

Bidders are not required to request the following preferences and incentives and the State's denial(s) of these preferences and incentives is not a basis for rejection of a proposal.

Bidders who are requesting a preference or incentive must certify they will comply with the requirements of the applicable Acts as appropriate. The State will monitor the life of a contract awarded with any applied preference or incentive for compliance to statutory, regulatory and contractual requirements. After award, the contractor is required to respond to the State's request for evidence of performance. The State will take appropriate corrective actions and may apply sanctions as necessary to enforce preference and incentive programs.

### 5.5.1 State of California Small Business Preference (NM)

- The regulations concerning the Small Business application and calculation of the small business preference, small business certification, responsibilities of small business, department certification, and appeals are revised. The new regulations can be viewed at www.pd.dgs.ca.gov/smbus. Access the regulations by clicking on "Small Business Regulations" in the right sidebar.
- 2. Non-Small Business (Non-SB) Sub-contractor Preference

A 5% bid preference is available to a Non-SB claiming 25% California certified small business sub-contractor participation. If applicable, submit a corresponding letter claiming the preference.

- 3. Non-SB if claiming the 25% preference are required to adhere to and provide with their bid the following:
  - a. If claiming the Non-SB sub-contractor preference, the bid response must include a list of the small business(es) with which the bidder commits to subcontract in an amount of at least twenty-five percent (25%) of the net bid price with one or more California certified small businesses in accordance with Government Code Section 14838(b) (1)(2).

- b. If Applicable, non-SB should provide a list of California Certified Small Business subcontractor's information with its bid response that contains the following:
  - Sub-contractor(s) name;
  - Sub-contractor(s) address;
  - Sub-contractor(s) phone number;
  - Sub-contractor(s) description of the work to be performed and/or products supplied; and
  - the sub-contractor(s) percentage amount of the net bid price (as specified in the solicitation) per sub-contractor.
- Each listed certified small business must perform a "Commercially Useful Function" (CUF) in the performance of the contract as defined in Government Code Section 14837(d)(4).

### 4. 4. Small Business Preferences

Bidders claiming the 5% preference must be certified by California as a small business or must commit to subcontract at least 25% of the net bid price with one or more California certified small businesses. Certification must be obtained no later than 5:00 p.m. on the bid due date.

Small Businesses are desired and encouraged to participate in this bid. Section 14835, et seq. of the California Government Code requires a 5% preference be given to bidders who qualify as a small business. The rules and regulations of this law, including the definition of a small business for the delivery of goods and services, are contained in California Code of Regulations, Title 2, Section 1896, et seq. The small business preference is for California-based small businesses.

a. To claim the small business preference, which may not exceed \$50,000 for any bid, the firm must have it's principal place of business located in California, have a complete application (including proof of annual receipts) on file with the State Office of Small Business and Disabled Veteran Business Certification (OSDC) by 5:00 p.m. on the bid due date (see front page) and be verified by such office.

Bidders desiring to claim the preference must submit completed Exhibit 10.16, Small Business Preference Information and the Bidder's Declaration Form, with the proposal.

Questions regarding the preference approval process should be directed to the OSDS at (916) 375-4940. A copy of the regulations, instructions and format for claiming the small business preference is available at <a href="http://www.pd.dgs.ca.gov/smbus/default.htm">http://www.pd.dgs.ca.gov/smbus/default.htm</a>.

#### 5.5.2 DVBE Incentive (NM)

For this RFP, DVBE has been waived.

### 5.5.3 Target Area Contract Preference Act (TACPA)/Enterprise Zone Act (EZA) & Local Agency Military Base Recovery Preference Request (LAMBRA) (NM)

This bid contains new and revised procedures for Bidders who wish to apply for TACPA, EZA, or LAMBRA preferences. Bidders are encouraged to review the package carefully to ensure that their submittals conform to the programs requirements.

- Target Area Contract Preference Act (TACPA), Government Code Section 4530 et seq.; Enterprise Zone Act (EZA), Government Code Section 7070, et seq.; and Local Agency Military Base Recovery Act (LAMBRA), Government Code Section 7118, et seq. information Package. Contract preference will be granted to California-based Bidders whenever contracts for goods or services are in excess of \$100,000 and the Bidders meet certain requirements as defined in the California Administrative Code (Title 2, Section 1806.30 et seq.) regarding labor needed to provide the goods being procured.
- 2. Bidders desiring to claim TACPA, EZA, LAMBRA preferences must submit a fully executed copy of the Standard Form 830, 831, 832, respectively, with their bid. These forms can be found at: http://www.pd.dgs.ca.gov/disputes/default.htm

NOTE: Bidders are not required to apply for TACPA, EZA, or LAMBRA preferences. Denial of TACPA, EZA, or LAMBRA preference requests is not a basis for rejection of the bid. Contracts awarded with applied preferences will be monitored throughout the life of the Contract(s) for compliance to statutory, regulatory and contractual requirements. The State will take appropriate corrective action to apply sanctions as necessary to enforce performance programs.

### **SECTION 6**

### **TECHNICAL REQUIREMENTS**

#### **Technical Requirements**

#### 6.1 Introduction

This Section contains the detailed technical and business requirements pertaining to this RFP. See Section 2, Rules Governing Competition, Section 5, Administrative Requirements, and Section 8, Proposal Format, for other requirements that must be met in order to be considered responsive to this RFP.

The State has determined that it is best to define its requirements, desired operating objectives, and desired operating environment. The State will not tailor these needs to fit some solution a Bidder may have available; rather, the Bidder shall propose to meet the State's needs as defined in this RFP.

#### 6.1.1 Statement of Work

Section 6, Technical Requirements, in its entirety is incorporated into the Statement of Work for RFP DGS 57224. (See Exhibit 10.2, Statement of Work)

### 6.2 Responses to Technical and Business Requirements

All of the requirements described in this Section are service levels and/or terms and conditions that the State expects to be satisfied by the Bidders. Bidders must indicate their willingness and ability to satisfy these requirements.

Responses to the requirements in this Section must be included in Bidder's Proposal, Volume 1 (Refer to Section 8, Proposal Format). The Bidder's proposal must include the required proposal items cross-referenced by the requirement number. It may also include any appropriate exhibits to illustrate responsiveness. These exhibits must be clearly labeled and cross-referenced to the proposal narrative by requirement number.

As applicable, responses to the requirements will be evaluated Pass/Fail in accordance with Section 9, Evaluation.

### 6.3 Designation of Requirements

The requirements specified in this RFP Section are classified as either "Mandatory" or "Non-Mandatory".

#### **Mandatory Requirements (M)**

All items labeled "Mandatory" or "M" are not negotiable. A Pass/Fail evaluation will be utilized for all mandatory requirements in this RFP. To be considered responsive, responsible and eligible to participate in the RFP, all questions identified as mandatory must be marked "Yes" to pass. Failure to respond to any mandatory requirements shall result in disqualification of the proposal.

As specified with each "Mandatory" or "M" requirement listed below, Bidders must indicate whether their proposal meets the individual requirement by marking either a "Yes" or "No" in the column next to the requirement, provide the location of a thorough narrative response as specified in Section 8, Proposal Format.

### Non-Mandatory Requirements (NM):

Some listed standards are identified as "Non-Mandatory" or "NM". Bidders are not required to comply with these requirements in order to be compliant with the RFP requirements. However, if a Bidder offers any of these non-mandatory requirements, the Bidder must meet the minimum requirements as stated in this section. Non-Mandatory items will not be considered when identifying Bidders who are responsive and responsible.

As specified with each "Non-Mandatory" or "NM" requirement listed below, Bidders must indicate whether their proposal meets the individual requirement by marking either a "Yes" or "No" in the column next to the requirement, provide the location of a thorough narrative response as specified in Section 8, Proposal Format.

### **Technical Requirements**

Section	Requirement Description	Туре	<b>Me</b> Yes	 Proposal Page Number
Product	s			
	Product Listing (Cost Worksheet)	М		
	The products included as part of this RFP are listed on Exhibit 10.9, Cost Worksheet. Cost Worksheet Light Duty Exhibit 10.9 must be completed as specified in Section 7, Cost.  Bidders shall furnish, for all Auto Parts bid, published literature, such as catalogs, product specifications, sales brochures, etc. in support of the proposed Auto Parts contract. Statements made by the bidder that are not supported by published literature or test results as applicable will not constitute satisfactory proof and will not be accepted as such.			
	The bidder shall provide proposed items for all categories identified on Exhibit 10.9 Cost Worksheet Light Duty. The State reserves the right to eliminate any specific categories from the final award, if indicated by an analysis of RFP Proposals.			

Catalogs							
	Catalogs	M					
	The bidder will be required to submit Auto Parts catalogs for this contract with its proposal.						
	The Auto Parts Catalog must:						
	<ul> <li>Contain all Categories bid from Exhibit 10.9 Worksheet.</li> </ul>						
	<ul> <li>Contain the base price so the State can verify a net price.</li> </ul>						
	<ul> <li>Be available at no charge to State and local governmental agencies for ordering under this contract.</li> </ul>						
	Catalog Format Requirements	M					
	The bidder will be required to provide the Auto Parts Catalogs in three formats as follows:						
	Hard Copy version						
	Electronic (PDF) version						
	Electronic (Excel) version						
	Hard Copy Version	M					
	The bidder will be required to provide one (1) hard copy of the Auto Parts Catalogs as described in the Cost Worksheet categories.						
	Electronic (PDF) Version	М					
	The bidder will be required to provide an electronic (PDF) version of the Auto Parts Catalogs. The PDF version will be provided in Adobe 7.0.						
	The Auto Parts Catalogs electronic PDF version shall display the following elements for each item:						
	Description of product						
	List price of the item						
	Discount to the State						
	Net cost to the State						
	Part number/Sku number						

Electronic (Excel) Version	М		
The bidder will be required to provide an electronic (Excel) version of the Auto Parts Catalogs. The Auto Parts Catalogs excel version will be provided in Microsoft Excel 2003 on a CD-ROM.			
The Auto Parts Catalogs electronic PDF version shall display the following elements for each item:			
Description of product			
List price of the item			
Discount to the State			
Net cost to the State			
Part number/Sku number			
Availability	М		
The contractor shall make the electronic (PDF) version of the Auto Parts Catalogs readily available from the contractor's home page.			
The contractor shall not have any catalog viewing restrictions for State departments or local governmental agencies trying to access the electronic catalogs.			
Submission and Approval	М		
Within thirty (30) calendar days of proposed award, the contractor must submit a preliminary electronic (PDF) version of the web catalogs and suggested website location to the State Contract Administrator for review.			
The State will review and determine acceptability of the catalogs and website location. Once the electronic (PDF) version has been approved by the State, the State shall provide written authorization to publish versions on contractor's website.			
EXTRA INFORMATION	M		
Provide details regarding the contractor's approach to maintaining the up-to-date information in both the contractor's on-line and hard copy catalogs. Describe the frequency of updates and how these updates are communicated to eligible users.			

Ordering						
	Ordering  The contractor(s) shall accept orders from any State Department or local governmental agency. Contractor(s) must not refuse to accept orders from any State Department or local governmental agency without written authorization from the State Contract Administrator.	М				
	Purchase Execution  California State Departments will submit orders on a Purchasing Authority Purchase Order (Std. 65) via one of the ordering methods as specified in Section 6, Ordering Methods.  Local governmental agencies may submit orders on their own purchase document directly to the contractor via one of the ordering methods as specified in Section 6, Ordering Methods. However, local California governmental agencies must obtain a State issued billing code prior to submitting the order.  Contractor(s) shall not accept purchase documents from State departments or local governmental agencies that are incomplete or contain non-contract items.	M				
	Minimum Order  There is no minimum order on this contract.	M				
	Ordering Methods  Contractor(s) shall accept orders through the following methods:  • Electronic (email) • Phone • Facsimile • Mail • Internet/Web	M				

Electronic (email) Orders	М		
Contractor(s) will accept orders via email as part of this contract. An email address to be used by ordering agencies for placing orders is to be in place before the commencement of this contract.			
Bidder shall provide email address for orders on Exhibit 10.8, Contractor Ordering Information.			
Phone Orders	М		
Contractor(s) will accept orders via telephone as part of this contract. A toll-free telephone number to be used by ordering agencies for placing orders is to be in place before the commencement of this contract. Telephone orders must be accepted between the hours of 7:00 AM and 7:00 PM, Monday through Saturday, in the time zone the order is placed.			
Bidder shall provide a toll-free telephone number for orders on Exhibit 10.8, Contractor Ordering Information.			
Facsimile Orders	М		
Contractor(s) will accept orders via facsimile as part of this contract. A toll-free facsimile number to be used by ordering agencies for placing orders is to be in place before the commencement of this contract. Facsimile orders must be accepted between the hours of 7:00 AM and 7:00 PM, Monday through Saturday, in the time zone the order is placed.			
Bidder shall provide toll-free facsimile number for orders on Exhibit 10.8, Contractor Ordering Information.			
Internet/Web Orders	М		
Contractor(s) will accept orders via internet/web as part of this contract. Web address to be used by ordering agencies for placing orders is to be in place before the commencement of this contract. Internet/web access must be 24 hours but orders must be accepted by the contractor between the hours of 7:00 AM and 7:00 PM, Monday through Saturday, in the time zone the order is placed.			
Bidder shall provide internet/web address for orders on Exhibit 10.8, Contractor Ordering Information.			

U.S. Mail Orders	M		
Contractor(s) will accept orders via U.S. Mail as part of this contract. Contractor must have the capability to receive orders by mail in place before the commencement of this contract.			
Bidder shall provide mailing address for orders on Exhibit 10.8, Contractor Ordering Information.			
Order Acknowledgement	M		
Contractor(s) must provide the ordering agencies with an order receipt acknowledgment via email or facsimile within twenty-four (24) hours after receipt of an order. The acknowledgement will include:			
Ordering Agency Name			
Agency Order Number (Purchase Order Number)      October Order Number			
<ul><li>Contractor Order Number</li><li>Description of goods and/or related products</li></ul>			
purchased			
<ul><li>Total Cost</li><li>Delivery Date</li></ul>			
Identification of any Out of Stock Items			
During the life of the contract, the contactor shall work with the State to provide any other future requirements the State may deem necessary.			
Out-of-Stock Remedy	M		
Upon receipt of order acknowledgment identifying out of stock items, the ordering agencies shall have the following options:			
Request back order			
<ul> <li>Request a substitute contract item</li> <li>Cancel the item from the order.</li> </ul>			
Under no circumstance is the contractor(s) permitted to make unauthorized substitutions or cancellations, without the authorization of the ordering agency.			

	1		
Warranty (ies)	М		
All products provided under this contract being identified as new and unused, must be of the latest model or design and of recent manufacturer. Statement of warranty is to be provided by supplier with items delivered under this contract.			
All remanufactured or rebuilt products must be explicitly identified as such in all RFP submittals and product literature or catalogs provided to customers in addition to all customer communication. These parts must be warranted to be free of defective parts and workmanship, provided they are properly installed on the vehicle for which they were intended, for the time and mileage shown in the product literature or catalog.  For items ordered into stock, warranty begins on the date			
of installation.			
Dealer Authorization	М		
The Bidder, if other than the parts manufacturer or major distributor, shall provide a current, dated, and signed authorization from the manufacturer that the Bidder is an authorized distributor, dealer, or service representative and is authorized to sell the manufacturer's parts. Failure to comply with this requirement may result in bid rejection.			
Franchise and Dealer Participation	М		
The Bidder, if representing multiple franchise and dealer locations, shall positively ensure that these locations have agreed to participate. Following award, refusal by Contractor's location(s) to participate may be considered cause for contract termination.  Complete and submit Exhibit 10.17, Authorized Locations and Exhibit 10.18, Wholesale Distribution Center			
Locations and submit with the proposal.			
Remanufactured Items	М		
The term "remanufactured" as used in this contract shall mean the part in question will be remanufactured so that all parts shall meet the Original Equipment Manufacturer's (OEM) specifications. Unless specifically requested by Purchaser, no product substitutions will be allowed.			

Core Exchange/Charges	М		
All parts supplied which require core exchange shall be monitored by the successful Bidder(s) to ensure proper cores are returned to the Bidder. Core credit is to be issued the same day as received. Any core that is not usable or not returned shall be billed at the core price listed in the manufacturer's price sheet for that particular part.			
Satisfaction Guarantee	М		
Bidder agrees that all parts purchased on this contract are supplied with a 100% satisfaction guarantee. All parts not meeting the form, fit, and/or function requirements for the legal purposes of the Purchaser will be provided a full refund including return shipping charges.			
OSHA Requirements	М		
Contractor agrees to comply with conditions of the Federal Occupational Safety and Health Acts of 1970 (OSHA), as may be amended, and the standards and regulations issued and certifies that all items furnished and purchased under this order will conform to and comply with said standards and regulations. Contractor further agrees to indemnify and hold harmless purchaser from all damages assessed against purchaser as a result of Contractor's failure to comply with the acts and standards and for the failure of the items furnished under this order to so comply.			

Delivery						
	Delivery Locations  Deliveries are to be made to the location specified on the individual order, which may include, but not be limited to inside buildings, high-rise office buildings, and receiving docks.	M				
	Delivery Schedules  Delivery of ordered product shall be completed within two (2) working days after receipt of an order, unless otherwise agreed to by the ordering agency. Since receiving hours for each ordering agency will vary by facility, it will be the Contractor's responsibility to check with each facility for their specific receiving hours before delivery occurs.  Contractor(s) must notify the ordering agency within 12 hours of scheduled delivery time, if delivery cannot be made within the time frame specified in the contract.	M				
	Delivery from Commercial Outlets  Delivery of standard in-stock items to locations within five (5) miles from company's commercial store shall be accomplished within 45 minutes. Delivery outside the five (5) mile range will be accomplished by a scheduled delivery.	M				
	Security Requirements for Institutions  Deliveries may be made to locations inside secure institutional grounds (such as the California State Prisons) that require prior clearances to be made for delivery drivers.  The Contractor(s) will be responsible for contacting the secure location for security clearance procedures, hours of operation for deliveries and service, dress code, and other rules of delivery. These Security clearance procedures may vary from facility to facility.  Deliveries that are delayed due to drivers not being cleared to enter institutional grounds may cause the contractor to be declared in default of contract requirements.	M				

	1		
Free on Board (F.O.B.) Destination	M		
All prices offered shall be F.O.B. destination; freight prepaid by the Contractor, to the ordering organization's receiving point. Responsibility and liability for loss or damage for all orders shall remain with the contractor until final inspection and acceptance, when all responsibility shall pass to the ordering organization, except the responsibility for latent defects, fraud, and the warranty obligations.			
Special order automotive items will be F.O.B. Pre-Pay and Add. A special order item will be the decision of the contractor (i.e. engines, transmissions, common carrier deliveries). Delivery charges may apply.			
Packing Slip	М		
A packing slip will be included with each shipment, which will include at least the following information in no particular order:			
Agency Order Number (Purchase Order Number)			
Ordering Agency Name			
Line item description			
Quantity ordered			
Quantity included in shipment			
<ul> <li>Any back order items, out of stock, and availability date of unfilled and partial shipment</li> </ul>			
Number of parcels			
Destination			
<ul> <li>All information contained on the packing label (see below)</li> </ul>			
Packaging Label	М		
A packing label will also be included with each order shipped and include the following items, visible on the outside of the box:			
Authorized Purchaser			
Address			
Department and floor			
Authorized Purchaser Contact Name			
Authorized Purchaser Telephone number.			

	Carton Weight	M		
	When packing multiple items for shipment, maximum weight of the individual cartons shall not exceed 50 pounds.			
	State of California Hazardous Materials Documentation	M		
	For the State of California, products that contain hazardous chemicals, as defined by California Code of Regulations, Title 8, §339, shall comply with the requirements of Title 8 of the California Code of Regulations including §340, Material Safety Data Sheets (MSDS).			
	Contractor(s) must provide a Material Safety Data Sheet (MSDS), with all legally required special handling/packaging documentation, to the ordering agency upon delivery of products containing hazardous materials. This information shall also be provided by facsimile or e-mail to any ordering agency upon written request.			
	All hazardous chemical products shall be labeled, tagged or marked with the following information:			
	<ul> <li>Identity of the hazardous chemical(s).</li> </ul>			
	Appropriate hazard warning.			
	<ul> <li>Name and address of the chemical manufacturer, importer or other responsible party.</li> </ul>			
Quality /	Assurance			
	Warranty	M		
	Contractor(s) must honor all manufacturers' warranties and guarantees on all products offered as part of this contract. If the manufacturer offers an on-site warranty, those warranty services must be passed on to the users of this contract. If a product warranty extends beyond the term of the contract, the Contractor(s) must agree to provide warranty services throughout the life of the warranty.			

Return Policy	M		
Bidder must agree to accept all products purchased against the contract for return at no cost to the State within 30 calendar days of delivery, and credit the customer in full.			
Note: Products returned for credit should be in the packaging as delivered and include all documentation.			
All returns shall be picked up within seven (7) working days of notification. Notification is defined as notice in writing, by facsimile or e-mail.			
Credit Policy	M		
Contractor(s) will be responsible for the credit or replacement of all products, including those covered by manufacturer warranties.			
Contractor(s) shall provide credit for the following items:			
<ul> <li>Items ordered or shipped in error</li> </ul>			
<ul> <li>Items that are returned within 30 calendar days of delivery</li> </ul>			
Defective or freight-damaged items			
The contractor(s) cannot require the ordering agency to deal directly with the manufacturer. In all cases, the ordering agency shall have the option of taking an exchange or receiving a credit.			
Restocking Fees	М		
Contractor(s) will not impose a restocking fee on the ordering agency for the following situations:			
Item(s) returned that were damaged upon receipt			
<ul><li>Incorrect items shipped</li><li>Items that are returned within 30 calendar days of</li></ul>			
delivery			
<ul> <li>Items that are returned, but exchanged for other items</li> </ul>			
Re-stocking fees for all other reasons can be no greater than ten percent (10%) of the value of the items needing re-stocking.			

	Product Recall Procedures	M		
	The Contractor(s) shall provide recall notification, regardless of level, in writing to the State Contract Administrator and each ordering agency through the most expedient method possible. The notices, at a minimum, shall include a complete product description and/or identification, contract number, delivery order number and disposition instructions.			
	Contractor(s) shall pick up, test, destroy or return recalled products to the manufacturer at no expense to the State. The Contractor shall issue replacement of product or credit for any product removed or recalled. Each ordering agency shall have the option of accepting either replacement product or credit in exchange for recalled/removed products.			
	Recall Notification	M		
	In the event of a product recall (or product service bulletin, etc.), the Contractor will notify the State Contract Administrator of the recall in addition to affected Purchasers.			
Invoicing	g/Payment			
	Invoicing	M		
	Ordering agencies may require separate invoicing, as specified by each ordering agency. The invoices must use the Contractor's invoice that includes at a minimum its name, address and telephone number. The Contractor will also issue invoices identifying, at a minimum, the			
	components listed below:			
	<ul> <li>Leveraged Procurement Number (State's Contract Number)</li> </ul>			
	Leveraged Procurement Number (State's Contract)			
	Leveraged Procurement Number (State's Contract Number)			
	<ul> <li>Leveraged Procurement Number (State's Contract Number)</li> <li>Agency Order Number (Purchase Order Number)</li> </ul>			
	<ul> <li>Leveraged Procurement Number (State's Contract Number)</li> <li>Agency Order Number (Purchase Order Number)</li> <li>Agency Billing Code</li> <li>Ordering Agency Name</li> <li>Delivery address of the order</li> </ul>			
	<ul> <li>Leveraged Procurement Number (State's Contract Number)</li> <li>Agency Order Number (Purchase Order Number)</li> <li>Agency Billing Code</li> <li>Ordering Agency Name</li> <li>Delivery address of the order</li> <li>Product(s) description</li> </ul>			
	<ul> <li>Leveraged Procurement Number (State's Contract Number)</li> <li>Agency Order Number (Purchase Order Number)</li> <li>Agency Billing Code</li> <li>Ordering Agency Name</li> <li>Delivery address of the order</li> <li>Product(s) description</li> <li>Manufacturer's product(s) number(s)</li> </ul>			
	<ul> <li>Leveraged Procurement Number (State's Contract Number)</li> <li>Agency Order Number (Purchase Order Number)</li> <li>Agency Billing Code</li> <li>Ordering Agency Name</li> <li>Delivery address of the order</li> <li>Product(s) description</li> <li>Manufacturer's product(s) number(s)</li> <li>Price(s) per unit(s) and extended price</li> </ul>			
	<ul> <li>Leveraged Procurement Number (State's Contract Number)</li> <li>Agency Order Number (Purchase Order Number)</li> <li>Agency Billing Code</li> <li>Ordering Agency Name</li> <li>Delivery address of the order</li> <li>Product(s) description</li> <li>Manufacturer's product(s) number(s)</li> <li>Price(s) per unit(s) and extended price</li> <li>Quantities of merchandise issued</li> </ul>			
	<ul> <li>Leveraged Procurement Number (State's Contract Number)</li> <li>Agency Order Number (Purchase Order Number)</li> <li>Agency Billing Code</li> <li>Ordering Agency Name</li> <li>Delivery address of the order</li> <li>Product(s) description</li> <li>Manufacturer's product(s) number(s)</li> <li>Price(s) per unit(s) and extended price</li> </ul>			

Listing of returns or back ordered items			
<ul> <li>Discounts (i.e. Cash Discounts, Prompt Payment Discounts, Volume Discounts, etc.). If applicable, discounts shall be clearly indicated on each invoice as a written statement. (Sample: "Prompt Payment Discount: 1% 20 Days")</li> </ul>			
<ul> <li>Totals for each order. Each invoice shall have a total for all orders, a total for all credits, and amount due from each agency</li> </ul>			
<ul> <li>Any other mutually agreed upon requirements the State may later deem necessary.</li> </ul>			
Within 30 days of cost opening, the Contractor(s) must submit a preliminary copy of their invoice which shall contain all the elements that have been specified above in the invoice requirements. The authorized copy shall become the standard for this contract review and determine acceptability of the invoice			
State of California CAL-Card Invoicing/P Card	М		
For the State of California, all CAL-Card invoices are to be processed separately from other payment methods and include the elements identified in Invoicing (see above). For additional information regarding the State of California, Procurement Division's CAL-Card program, see the following website: <a href="https://www.pd.dgs.ca.gov/calcard">www.pd.dgs.ca.gov/calcard</a> .			
Note: The total invoice amount for each CAL-Card order must reflect a zero balance due or credit, if applicable, and state "paid by CAL-Card".			
Other participating states may have their own P Card/Cal Card Type programs. For those participating states, the individual State's rules apply.			
Payment	М		
State departments and local government agencies may pay by check, electronic funds transfer, or with the State's CAL-Card credit card (currently VISA). Payments are to be made in accordance with paragraph 30 of the State's General Provisions.			

Implementation						
	Proposed Implementation Plan	M				
	Bidders are required to provide a proposed implementation plan approach for this project using the timeline above as part of the proposal. Relevant details pertaining to training, communication, contract adoption issues, contract requirements and timelines must be included.					
	The implementation plan shall address and provide timelines for all contract and RFP requirements. It shall provide a full description of:					
	<ul> <li>Contractor's quality assurance program which will affect service and delivery.</li> </ul>					
	<ul> <li>A tentative schedule for modifying the Contractor's system to add each agency to the contract.</li> </ul>					
	The method proposed and escalation process to ensure Contractor's ability to effectively handle implementation problems (e.g., adding extra staff during the implementation period, Contractor's staff training, State staff training).					

### **Business Requirements**

Section	Requirement Description	Туре	Meets		Proposal Page
Occion	requirement Description	Турс	Yes	No	Number
Reportin	g				
	Contract Usage Report	M			
	The State will provide all Contractor(s) with a template of the electronic Contract Usage Report. This report will be furnished to the State Contract Administrator on a quarterly basis. The report shall detail all invoiced purchases against the contract during the specified reporting period.				
	The Contract Usage Report shall be provided in the format specified in Exhibit 10.13, Contract Usage Report Template. The Contract Usage Report must contain at a minimum, but not BE limited to, the data elements identified in this exhibit.				
	The Contract Usage Report shall be provided to the State Contract Administrator in Excel 2003 format via email or via U.S. Mail on a CD-ROM.				
	The Contract Usage Report shall be submitted on a calendar quarter basis. All Contract Usage Reports shall be due on the 15th day following the ending of the specified reporting period.				

Section	Requirement Description	Туре	Meets Yes No	Proposal Page Number
	Post Consumer Recycled Content (PCRC) Certification	M		
	State agencies are required to report purchases made within 11 product categories in the California Integrated Waste Management Board's (CIWMB) State Agency Buy Recycled Campaign (SABRC). In order to comply with those requirements, bidders are required to certify, in writing, the minimum percentage, if not the exact percentage, of post consumer recycled content material (PC RC) in each of the products offered as part of this RFP.			
	Exhibit 10.10, Post Consumer Recycled Content (PCRC) Certification Workbook contains the following documents:			
	PCRC Percentages Worksheet – Bidders shall complete the PCRC Percentages Worksheet listing the percentage of post consumer recycled content material for each product offered.			
	2) Letter of Certification – Bidders shall print and sign the Letter of Certification certifying that the minimum percentage, if not exact percentage, listed in the PC RC Percentages Worksheet is true and accurate. The Letter of Certification shall be furnished under penalty of perjury. The Letter of Certification shall be provided regardless of content, even if the product(s) contain no post consumer recycled material.			
	<ol> <li>Reportable Product Categories Table – This table is provided for informational purposes only and identifies the 11 reportable SABRC product categories.</li> </ol>			
	Within 30 calendar days from contract award, the Contractor will be required to provide signed Letter of Certification and the completed PCRC Percentages Worksheet to the State Contract Administrator. The PCRC Percentages Worksheet shall be provided in hard copy and electronic copy (on a CD-ROM in Microsoft Excel format).			
	During the life of the contract, the Contractor will be required to submit revised PCRC Certification Workbook information if percentages are adjusted or if substitute line items are approved by the State Contract Administrator.			

	SB Sales Report	М		
	For the State of California, if using a Small Business, contractor(s) shall provide detailed reports of the California-certified Small Business Enterprises usage. Reports shall be supplied to the ordering agencies upon their request. Reports shall include the following elements.			
	Manufacturer Part Number			
	Item Description			
	Manufacturer			
	Quantity Shipped			
	Contract Price per Unit			
	<ul> <li>Extended contract price including all discount(s) applied</li> </ul>			
	<ul> <li>Year-to-Date Quantity Shipped</li> </ul>			
	<ul> <li>Year-to-Date Product Value</li> </ul>			
	Year-to-Date Number of Orders Placed			
	Ad Hoc Reporting	М		
	The Contractor(s) shall have the ability to provide ad hoc reports at no cost to the State. The Contractor shall permit and provide access to all data that pertains to any procurement action taken by an ordering agency or the State as a whole. The State or ordering agency may make copies of procurement data in any form and the use of such data shall not be restricted.  Dependent on future reporting requirements, the State may ask that certain reports become standard and delivered to the State on a monthly or quarterly basis.			
Contract	Management	I		
	State Contract Administrator	М		
	The State Contract Administrator is specified in Section 1.5 and is the contact person throughout the life of the contract.			
	Any exception to the requirements contained in the contract may only be authorized by the State Contract Administrator or his/her designee through an amendment.			

Contractor Contract Manager	М		
The Contractor(s) will assign a National Account Manager contact person for contract management purposes. The contract manager is to be identified in Exhibit 10.7, Contractor Contract Manager Information. The Contractor Contract Manager must be authorized to make decisions on behalf of the Contractor.			
Customer Service	М		
The Contractor(s) will have a customer service unit that is dedicated to this contract. The customer service unit shall be staffed with individuals that are trained in the requirements of this contract and have the authority to take administrative action to correct problems that may occur.			
The Contractor(s) shall respond to customer inquiries within one (1) business day of initial contact.			
The Contractor(s) shall provide customer service unit personnel information (phone number(s), names, etc.) within 15 days of contract award.			
Problem Resolution	М		
The Contractor(s) shall promptly notify the State Contract Administrator in writing of any unresolved issues or problems that have been outstanding for more than five (5) working days. The State Contract Administrator shall notify the Contractor of the same.			
WSCA Management Fee	М		
The contractor is required to pay a quarterly WSCA administrative fee of 0.5% (0.005) of ALL sales under the Price agreement. The WSCA administrative fee is to be included in the discount offered in the price proposal.			
State of California Management Fee	М		
A 1% management fee will be included in the cost of all goods purchased against the State of California Auto Parts contract. This 1% (0.01) fee will then be rebated back to the State of California based on a quarterly Contract Usage Reports. This fee will be due to the State on the 15 <sup>th</sup> day following the ending of the specified reporting period.			

Other State Specific Administrative Fee:  If a particular State requires a state administrative fee, or the administrative fee is different than 1%, it will be negotiated and articulated in that State's participating addendum.	M		
Joint Outreach  The Contractor(s), in coordination with the State Contract Administrator, may be required to perform, at the State's request, broad outreach marketing the contract terms and conditions to ordering agencies.	M		
Performance-Service Level  The State, as part of an on-going quality management effort, will measure the user agencies' satisfaction during the term of the contract. Measurement will identify areas where service levels can be improved. Bidders must agree to work with the State in resolving any issues resulting from the measurement of customer satisfaction. The successful bidder(s) and the State Contract Administrator will work together to continually monitor the success levels of the contract.	М		
Transition Plan at End of Contract  The Contractor(s) must agree that at the end of this contract, should the State conduct another procurement and award a new contract, the Contractor(s) will work with the State Contract Administrator to ensure that an efficient and effective transition takes place.	M		

Pricing				
	Price Adjustments	М		
	This contract may be amended by mutual consent.			
	During the contract term, if the State becomes aware of better percentage off of market offered by the Contractor(s) for substantially the same or a smaller quantity of a product outside the contract, but upon the same or similar terms of the contract, then upon mutual agreement between the State and the Contractor(s), the percentage over market under the contract shall be immediately reduced to the lower percentage. If no agreement can be reached, the State reserves the right to not purchase the product under this contract.			
Custome	er References			
	Customer References	M		
	Customer references are intended to gather information related to the Bidder's experience successfully servicing accounts with requirements similar to those of this solicitation. Bidders shall submit three (3) customer references from different contracts.  Customer references shall be provided on Exhibit 10.14, Customer Reference Form. Customer Reference Forms from any other procurement will not be accepted for this RFP.  Customers must fully complete the Exhibit 10.14, Customer Reference Form, and return directly to the Bidder. The Bidder must then submit the completed exhibit with the proposal as specified in Section 8, Proposal Format.  Each contract reference shall:  Be valued at \$1,000,000.00, at a minimum;  Be for work performed within the last five (5) years;  Have similar scope of work to that requested in this solicitation  Failure to submit three (3) completed customer references, meeting the requirements above, will result in the Bidder's proposal being considered non-responsive.			

EPP Requirements						
Contrac Supplies	tor's Packaging and Shipping Containers and	NM				
paper pa orders (i recyclab	irst year of the contract, all contractor-provided ackaging for carry out, delivery and shipping e.e. bags, boxes, cartons, containers, etc.) shall be le and contain 20% (at a minimum) post er recycled content (PCRC).					
shipping contractor the State minimum containe content a and not or inks. Bid plan and PCRC for signed be	first year of the contract, the packaging and containers and supplies employed by the or for carry out, delivery and shipping shall meet a Agency Buy Recycled Campaign (SABRC) in PCRC content requirements (30%). The rs shall be legibly marked with the percent PC RC and shall be easily deconstructed for recycling contaminated with any non-recyclable stickers or dders shall provide, with their proposal, a written commitment to meet the 30% SABRC minimum or all shipping boxes, by end of the first year, by the company's Chief Executive Officer.					
employe addition consume post con	ng and shipping containers and supplies d by the contractor for shipping may contain, in to the recyclable minimum SABRC post er recycled content (PC RC) material, virgin nonsumer recycled content fiber that is certified from hable forest or mill certification (alphabetical order)					
	<ol> <li>American Forest &amp; Paper Association's Sustainable Forestry Initiative® (SFI) certification http://www.afandpa.org/Content/NavigationMe nu/Environment_and_Recycling/SFI/SFI.htm</li> <li>American Tree Farm System (ATFS) certification http://www.treefarmsystem.org/cms/pages/26 19.html</li> </ol>					
	3. Canadian Standards Association's (CSA) certification to the Canada's National Standard for Sustainable Forest Management (SFM) CAN/CSA-Z809 <a href="http://www.csa-international.org/product_areas/forest_product_s_marking/Default.asp?language=english">http://www.csa-international.org/product_areas/forest_product_s_marking/Default.asp?language=english</a>					
	4. Forest Stewardship Council (FSC) certification <a href="http://www.fscus.org/paper/">http://www.fscus.org/paper/</a>					

Toxics in Packaging Purchasing Specification All packages, packaging components, and packaged retail-ready products offered during the life of the contract must be in full compliance with all requirements of the Toxics in Packaging Prevention Act (California Health and Safety Code (HSC) sections 25214.11-25214.26.) "Package" and "packaging components" are defined in HSC section 25214.12.  Upon request by the State, the awarded supplier shall provide a Certificate of Compliance which covers each type of package, packaging component, or packaged retail-ready product supplied, distributed, or sold under this specification. The certification shall be based either on direct written correspondence from the packaging product manufacturer or from any written advertisement issued by the product manufacturer including, but not limited to, the product's label, the manufacturer's catalog, or the manufacturer's or vendor's Internet Web site.	NM		
Contractors Catalogs  After the first year of the contract any printed market or ordering documents or catalogs for the State of California shall be printed on Process Chlorine Free (PCF), minimum 30 % post-consumer recycled content paper.  Any printed market or ordering documents or catalogs for the State of California may contain, in addition to the recyclable minimum SABRC post consumer recycled content (PC RC) material, virgin non-post consumer recycled content fiber that is certified from a sustainable forest or mill certification (alphabetical order):  5. American Forest & Paper Association's Sustainable Forestry Initiative® (SFI) certification http://www.afandpa.org/Content/NavigationMenu/Environment_and_Recycling/SFI/SFI.htm  6. American Tree Farm System (ATFS) certification http://www.treefarmsystem.org/cms/pages/26_19.html  7. Canadian Standards Association's (CSA) certification to the Canada's National Standard for Sustainable Forest Management (SFM) CAN/CSA-Z809 http://www.csa-international.org/product_areas/forest_product_s_marking/Default.asp?language=english  Forest Stewardship Council (FSC) certification http://www.fscus.org/paper/  Hard copies of the catalog shall be distributed to State employees and local governmental agencies for ordering under this contract only upon written request. These hard	NM		

	<ul> <li>Green Product marking and labeling</li> <li>After the first year of the contract, contractor shall mark and identify all of its green products on its website, in the catalog, and in its store floors including but not limited to:         <ul> <li>Products that possess third party certified ecologos certifications.</li> <li>Products that meet or exceed the California SABRC minimum post consumer recycled content requirements.</li> </ul> </li> <li>After the first year of the contract, contractor shall compile and distribute upon request to any Agency/entity a list of green product offerings updated at least annually. The list shall be in electronic and hardcopy form.</li> </ul>	NM		
EPP Cor	porate Performance Requirements			
	CORPORATE ENVIRONMENTAL MANAGEMENT SYSTEM (EMS) Bidders are requested to disclose their company Environmental Management System (EMS) information on Attachment X – EMS Questionnaire including:  • Corporate Environmental Policy Statement. • EMS Plan and Goals • EMS Tracking and Annual Management Reporting  Bidders are required to return completed Attachment X – EMS Questionnaire with the bid response.	NM		
	Environmental Management System (EMS) Bidders may provide documentation of third party or self certification to ISO 14001 environmental management system (EMS) standard as follows:  1. Option 1 - Third Party Certification Bidders must provide a copy of their current third party certification to ISO 14001.  2. Option 2 - Self Certification Bidders must provide a copy of the self certification checklist for compliance to ISO 14001.  Bidders that are certified must provide the following with their proposal:  • A link to the companies Corporate Environmental Policy Statement  • A link to the company's EMS Plan on the companies website  • A link to the companies declared environmental goals addressing energy, water, resources, toxic materials	NM		

<ul> <li>A link to the companies EMS time table for impact reductions</li> <li>A link to the companies EMS Tracking and Annual</li> </ul>			
Registered for green house gas emissions footprint disclosure and any verified emission reductions achieved  Bidders may provide documentation that their operations have been registered in a Climate Registry that uses the ISO 14064 metrics and accounting methodology for their operation's green house gas emissions for their facilities and their transportation activities.	NM		
Toxic use reduction specifications  The sale of wheel weights containing lead, brake pads containing asbestos, and products containing mercury such as thermostats shall be prohibited from sale on this contract.	NM		
Toxic use reduction specifications – MSDS  Bidders may provide upon request an MSDS for any product containing a hazardous material. After the first year of the contract the contractors may make the MSDS available on their website for any product the State of California has purchased in the first year containing a hazardous material.	NM		
Bidders may not sell, supply, offer for sale, or manufacture for sale in California consumer products which do not comply with California Code of Regulations, Title 17, Division 3, Chapter 1 – Air Resources Board, Subchapter 8.5 – Consumer Products, Article 2 – Consumer Products, § 94507 to , § 94517 (available at: <a href="http://www.arb.ca.gov/consprod/regs/cp.pdf">http://www.arb.ca.gov/consprod/regs/cp.pdf</a> and <a href="http://www.arb.ca.gov/enf/title17_94509.pdf">http://www.arb.ca.gov/enf/title17_94509.pdf</a> .)  Bidders shall not sell, supply, offer for sale, or manufacture for sale in California consumer products which do not comply with California Code of Regulations, Title 17 AIRBORNE TOXIC CONTROL MEASURE FOR EMISSIONS OF CHLORINATED TOXIC AIR CONTAMINANTS title 17, California Code of Regulations, 93111	NM		

### **Recycled Content Certifications**

Public Contract Code (PCC) section 12201(c) requires State Departments to report recycled content of products purchased through statewide contracts to the legislature each year. State Agency Buy Recycled Campaign (SABRC) is a state mandated program that requires the reporting of all purchases made within the 11 specified product categories identified on Exhibit 10.10, Reportable Product Categories Table.

All bidders shall certify in writing to the contracting officer the minimum percentage, if not exact percentage, of post consumer recycled-content (PC) material in the products, materials, goods, or supplies being offered or sold to the State regardless of whether the product meets the minimum content requirements specified in law. The certification shall be furnished under penalty of perjury. The certification shall be provided regardless of content, even if the product contains no recycled material.

The following are the definitions that shall be applied for certifying the post consumer recycled content:

- Post Consumer Material A finished material that would have been disposed of as a solid waste, having completed its life cycle as a consumer item, and does not include manufacturing wastes.
- Recycled Products Goods or materials that meet the requirements identified in Public Contract Code Section 12209, including any good or material that has been reused or refurbished without substantial alteration of its original form.

In order to meet this requirement, bidders are required to:

 Complete and sign Exhibit 10.10, Post Consumer Recycle Content (PCRC) Certification Workbook.

### SECTION 7 COST

### **SECTION 7**

**COST** 

9-16-09 7 - 1

#### 7. Cost

#### 7.1 Introduction

Cost is the primary evaluation criterion for the award for this RFP. Cost evaluation will be based on the highest discount (%) to the State, as calculated according to the methodology outlined in Section 9, Evaluation. All proposed discounts (%) for all Categories must include the cost of any services required in this RFP.

The State's intent is to structure the pricing format in order to facilitate a straightforward comparison among all bidders and foster competition to obtain the best market pricing. Consequently, the State requires that each bidder's quote be in the format outlined in this section. Therefore, Bidders are advised that failure to comply with the instructions listed in this section, such as submission of incomplete offers or use of alternative pricing structures or different formats than the one requested may result in the rejection of their proposals for non-responsiveness.

Note: It is imperative that cost information not be included in the body of the proposal. Cost information is to be submitted in accordance with Section 8, Proposal Format.

### 7.2 Pricing Structure

The pricing information is categorized under eighteen (18) main Categories.

The Category list provides detail on those products commonly purchased by the State. The pricing proposal for this Category list will only be considered compliant if the bidder provides pricing for all Categories listed. A blank Category will be interpreted as a noncompliant pricing proposal and the bidder will be deemed non responsive.

The contract will be awarded on a multiple award basis. All mandatory Categories must be bid or the proposal will be considered non-responsive and will be rejected.

#### 7.3 Quantities

No quantities will be shown in Exhibit 10.9 Cost Worksheet, as the State has no estimate of the anticipated contract purchase pattern. Actual purchases may vary and the State will not guarantee any quantities will be purchased. Consequently, there is no quaranteed dollar value for this contract.

#### 7.4 Worksheet Instructions

Exhibit 10.9 Cost Worksheet Light Duty, contains the worksheet that the Bidders shall use to enter cost information for Light Duty Cars and Trucks. The Light Duty Cost Worksheet **must** be filled out completely or the proposal will be rejected.

Bidders must follow the instructions below to complete the Cost Worksheet:

9-16-09 7 - 2

### 7.4.1 Completion of Worksheet

For Categories 1-18 contained in this worksheet, please input the following for each Category (unless otherwise instructed):

The Discount (%) being offered to the State for that Category.

The minimum discount offered in the proposal is fixed for the life of the contract. However, the contractor can offer greater discounts at any time during the course of the contract.

### 7.4.2 Important details

- Catalog List Prices are to be those listed in the current version of the bidder's commercially available catalogs and will be the basis for which all discounts are applied to arrive at the State's net cost.
- 2) The catalogs provided shall consist of both a hard copy commercially available catalog and a publicly accessible on-line catalog and must contain a full range of auto parts in all categories. All catalogs must be named numbered and dated. Bidder may offer more than one catalog per category but only one discount per category.

#### 7.5 Cost Submittal

Final pricing will be submitted in a separate sealed envelope as specified in Section 8, Proposal Format.

Final pricing will be submitted in a hard copy version and CD-ROM format in a sealed envelope as part of the final proposal, refer to Section 3.2.3 and 3.2.5, Format and Contact. Please ensure that a whole number percentage discount (%) with no decimal places is entered as indicated in the instructions and examples for the pricing worksheets. The number that is submitted on the pricing CD as part of the final proposal, will be used for evaluation purposes.

### 7.6 Additional Savings

For the purpose of cost, if at any time a commercial sale price anywhere in the nation is offered to the public and is cheaper than a contract price, the sale price must be offered to the State in lieu of the contracted discount price.

9-16-09 7 - 3

### SECTION 8 Proposal Format

### **SECTION 8**

### **PROPOSAL FORMAT**

9-16-09 8 - 1

### 8. Proposal Format

#### 8.1 Introduction

These instructions prescribe the mandatory proposal format and the approach for the development and presentation of proposal data. All proposals must be submitted in the format described in this section.

The Bidder's RFP response must contain all the material requested and address all requirements identified in the RFP. Format instructions must be adhered to for the proposal presentation. All requirements and questions in the RFP must be responded to and all requested data must be supplied.

Each Bidder is responsible for providing sufficient information and documentation for their RFP response to be thoroughly evaluated. Failure to do so may result in rejection of the proposal. There is no intent to limit the content of the proposal. Additional information deemed appropriate by the Bidder should be included. However, non-relevant material in the proposal which conflicts with the State's best interest may be cause for rejection of the proposal.

It is the Bidder's responsibility to ensure their RFP response is submitted in a manner that enables the State Evaluation Team to easily locate response descriptions and exhibits for each requirement of this RFP.

### 8.2 General Format Requirements

Draft and Final Proposals shall be completely sealed, marked and delivered as indicated in Section 2.3.5.6, Delivery of Proposals. The Draft and Final Proposals shall be submitted to the Procurement Official identified in Section 1.5 by the time specified in Section 1.6, Key Action Dates.

The pages in the Bidder's RFP response must be standard 8.5" x 11" paper. Larger paper is permissible for charts, diagrams, spreadsheets, etc. Font size should be no smaller that 10 font. Pages must be placed within a binder with tabs delineating each section.

The following must be shown on each page of the RFP response:

- Name of Bidder
- RFP 57224
- Volume Number
- Page Number

Figures, tables, charts, etc., must be assigned index numbers and must be referenced by these numbers in the RFP text and in the RFP response Table of Contents. Figures, tables, charts, etc. should be placed as close to text references as possible.

### 8.3 Final Proposal Format and Content

#### 8.3.1 Instructions

The Final Proposal shall be completely sealed, marked and delivered as indicated in Section 2.3.5.6, Delivery of Proposals.

The Final Proposal shall be submitted to the Procurement Official identified in Section 1.5 by the time specified in Section 1.6, Key Action Dates.

### 8.3.2 Proposal Copies

The Final Proposal must be submitted in the number of copies indicated below. Note: One complete set of all required volumes shall be clearly marked "Master Copy".

Volume	Number of copies
	1 "Master Copy"
Volume I – Response to Requirements	7 Copies
	2 CD-ROMs
	1 "Master Copy"
Volume II – Cost Data	4 Copies 2 CD-ROMs
	2 CD-ROMs

### 8.3.3 Proposal Content

The content of each required volume is described below.

### **VOLUME I – Response to Administrative Requirements**

This volume must contain all responses to the non-technical and technical requirements of the RFP. All applicable forms, except cost data, must be completed and included in this volume. This volume must contain a Table of Contents which identifies the major parts of the RFP response, including forms and exhibits.

<u>Note:</u> Cost information shall not be included in this volume. Bidder must put "XXX" in place of all cost figures and percentages related to costs.

The organization of this volume is to be as follows:

### 1) Section 1 – Cover Letter

This section must contain a signed Cover Letter on the Bidder's official business Letterhead stationery in accordance with Section 2.3.5.5, Proposal Signature. Price information must not be included in the Cover Letter.

The Cover Letter must contain a statement that substantiates that the person who signs the letter is authorized to bind the Bidder's firm contractually. The Cover Letter must also contain statements that substantiate that the Bidder will comply with all RFP requirements and that the Bidder agrees with all terms and conditions.

# 2) Section 2 – Response to Administrative Requirements (RFP Section 5)

This section must contain a narrative of the Bidder's compliance and/or response for each item (all paragraphs and subparagraphs) identified in RFP Section 5, Administrative Requirements. Each response must reference the RFP Section number to which it is responding. Responses shall be provided in the same paragraph sequence as they are provided in the RFP.

This section must also contain all required and/or applicable forms included in this RFP, including but not limited to, the following:

- Exhibit 10.3 Contract Development/Standard Agreement (STD 213)
- Exhibit 10.5 Administrative Requirements Response
- Exhibit 10.6 Retailer's Seller's Permit Information
- Exhibit 10.12 State of California Darfur Contracting Act Form
- Small Business Preference Information (Note: Bidder only needs to submit completed exhibit if requesting this preference)
- Bidder's Declaration Form (GSPD-05-405)

### 3) Section 3 – Response to Technical Requirements (RFP Section 6)

This section must contain the Bidder's completed Section 6, Technical Requirements, in its entirety.

This section must also contain all narrative responses to each requirement identified in RFP Section 6, Technical Requirements and must also include applicable literature necessary to support the response. References may include brochures, specification information, etc. Substantive technical information should be highlighted on the literature page. Mark the applicable pages (i.e. circling, underlining, bracketing, or using a highlight marker that will still show when reproduced, etc.) and list the corresponding Technical Requirement number in the margin.

Bidders must provide a response to every "Mandatory" requirement. Each response must identify the specific requirement being addressed by the requirement section number.

Failure to respond to a requirement may be cause for rejection of the Bidder's proposal. The State reserves the right to determine if the Bidder's response to a requirement, as detailed in their description and/or supporting documentation, supports or contradicts the Bidder's claim of intended compliance.

This section must also contain all required and/or applicable forms included in this RFP, including but not limited to, the following:

- Exhibit 10.9 Cost Worksheet(s) (without cost figures)
- Exhibit 10.7 Contractor Contract Manager Information
- Exhibit 10.8 Contractor Ordering Information

Reminder: No cost information should be contained in this volume. Bidder must put "XXX" in place of all cost figures and percentages related to costs. Any product supporting literature containing prices or rates (such as catalogs, maintenance service rates, etc.) submitted as part of the RFP must have price figures replaced with "XXX."

### **VOLUME II – Cost Data**

This volume must be in a separately sealed, clearly identified marked envelope or container containing all completed cost sheets, exhibits identifying cost, and any other document(s) with cost data. This volume must contain all cost sheets required by Section 7, Cost. The MS Excel workbook must be used for this Volume and the cost sheet must contain the required items.

Note: Pricing figures shall not appear in any proposal submissions except the Final Proposal.

The organization of this volume is to be as follows:

### 1) Section 1 - Cost Worksheet

This section must contain the completed Exhibit 10.9, Cost Worksheet(s), as specified in Section 7, Cost.

### 2) Section 2 – Additional Cost Documents

This section should contain all other exhibits and documents with cost data.

### 8.4 Draft Proposal Format and Content

The Draft Proposal shall be completely sealed, marked and delivered as indicated in Section 2.3.5.6, Delivery of Proposals. The Draft Proposal shall be submitted to the Procurement Official identified in Section 1.5 by the time specified in Section 1.6, Key Action Dates.

The Draft Proposal must be submitted in the number of copies indicated below. Note: One complete set of all required volumes shall be clearly marked "Master Copy".

Volume	Number of copies
Volume I – Response to Requirements	1 "Master Copy"
	7 Copies
	2 CD-ROMs
Volume II – Cost Data	1 "Master Copy"
	4 Copies
	2 CD-ROMs

The Draft Proposal must contain all the material specified in Section 8.3.3, Proposal Content, for the Final Proposal except cost figures. Bidder must put "XXX" in place of all cost figures and percentages related to costs. The completed Standard Agreement (STD 213) must also be included but need not be signed. However, the Standard Agreement (STD 213) must contain the name and title of the person who will be signing for the Bidder at the submission of the Final Proposal.

Note: Pricing figures must not appear in any proposal submissions except the Final Proposal, Volume II. The inclusion of pricing in the Draft Proposal may be a basis for rejecting the proposal and notifying the Bidder that further participation in the procurement is prohibited.

## SECTION 9 EVALUATION

\_\_\_\_\_\_

# **SECTION 9**

**EVALUATION** 

9-16-09 9 - 1

#### 9. Evaluation

#### 9.1 Introduction

This Section describes how the State will evaluate the proposals and identify those proposals that meet the RFP's objectives and provide the best value for the State. It is the State's intent to conduct a comprehensive, impartial evaluation of all proposals received. The State will use a pass/fail and final cost, two-envelope method of selection.

#### 9.2 State Evaluation Team

The State has established a multi-state sourcing team that will be responsible for the review and evaluation of Bidder proposals in accordance with the process described in the RFP. The State may engage additional qualified individuals during the process to assist the State Evaluation Team in understanding technical, financial, legal, contractual, or program matters.

Evaluation Criteria: All Mandatory Administrative and Technical Requirements listed in Section 5 and Section 6 of RFP DGS-57224 will be scored on a pass/fail basis. All costs submitted per Section 7 Cost for RFP DGS- 57224 will be evaluated by the following criteria:

The State Evaluation Team will select a representative sample of like or comparable items from each category for each bidder. These samples will be referred to as the Market Basket for each bidder (see Exhibit 10.19). Each State which has a member on the evaluation team, has submitted multiple vehicles from their individual State fleets to evaluate. A part from each category will be selected for evaluation for each vehicle. The Evaluation Team's Subject Matter Experts will ensure that the parts evaluated from each bidder are comparable (to ensure fairness to each bidder). The State Evaluation Team will arrive at a final net cost to the State for each item listed in each Bidder's Market Basket, by applying the appropriate discount listed for each category. All items listed in each Bidder's Market Basket will then be totaled to arrive at a Grand Total. All bidder's Grand Totals that fall within the established 20% competitive range and are in full compliance with all the requirements of RFP DGS- 57224 will then be considered for award.

### 9.3 Receipt

Proposals must be delivered by the date and time stated in RFP Section 1.6, Key Action Dates. Each Proposal will be dated and time marked as it is received, and verified that it is properly sealed. Proposals will remain sealed until the designated time for opening. Final Proposals received after the date and time specified in RFP Section 1.6, Key Action Dates, for receipt of Final Proposals will be deemed non-responsive and will be rejected.

### 9.4 Negotiations

Not Applicable to this RFP

9-14-09 9 - 2

# **SECTION 10**

# **EXHIBITS**

9-16-09 10 - 1

## SECTION 10 Exhibits

## 11. Exhibits

This section contains the following exhibits included as part of this RFP:

Exhibit No.	Description	
10.1	Letter of Intent to Bid	
10.2	Statement of Work	
10.3	Contract Development/Standard Agreement (STD 213)	
10.4	Competitive Bidding and Proposal Responsiveness	
10.5	Administrative Requirements Response	
10.6	Retailer's Seller's Permit Information	
10.7	Contractor Contract Manager Information	
10.8	Contractor Ordering Information	
10.9	Cost Worksheet, Light Duty	
10.10	Post-Consumer Recycled Content (PCRC) Certification Workbook	
10.11	California Department of Corrections and Rehabilitation (CDCR) Special Provisions	
10.12	Darfur Contracting Act Form	
10.13	Contract Usage Report	
10.14	Customer Reference Form	
10.15	WSCA Terms and Conditions	
10.16	State of California Small Business Preference Information	
10.17	Authorized Locations	
10.18	Wholesale Distribution Center Locations	
10.19	Market Basket Evaluation Form (Sample)	

9-16-09 10 - 2



State of California • Arnold Schwarzenegger, Governor State and Consumer Services Agency

### DEPARTMENT OF GENERAL SERVICES

707 Third Street • West Sacramento, CA 95605 • www.dgs.ca.gov

September 21, 2009

To All Bidders

# Addendum #1 to RFP 57224 Auto Parts

The addended pages of this RFP are available online under EProcurement (Bid Sync) located at <a href="https://www.bidsync.com/DPX?ac=view&auc=333831">https://www.bidsync.com/DPX?ac=view&auc=333831</a>. If you are unable to download a copy of the addenda from this website, an electronic Word version is available upon request. Changes to the RFP are attached.

The only change to the information and requirements is:

Exhibit 10.2, Scope of Work

Replace original Scope of Work with the attached Scope of Work.

Sincerely,

Eileen Tardiff Master Agreements Procurement Division (916) 375-4390

email: eileen.tardiff@dgs.ca.gov

DGS RFP 57224 EXHIBIT 10.2

### STATEMENT OF WORK

#### 1. INTRODUCTION

This Statement of Work ("Agreement") reflects the goods and services to be provided by the Contractor hereinafter referred to as the "Contractor," for the State of California hereinafter referred to as the "State" for the Auto Parts Contract. Section 6, Technical Requirements shall be incorporated as a part of this Statement of Work along with all attachments and exhibits.

### 2. SCOPE

The Contractor will provide the entire portfolio of quality after-market light duty auto parts to State departments and local governmental agencies in a timely and efficient manner. The Contractor will be the primary point for warehousing, data collection, reporting, and distribution of office supplies to the State.

### 3. PERIOD OF PERFORMANCE

The term of this Agreement is effective for three (3) years from the date of approved contract document with the option to extend two (2) additional years in various increments upon approval by the Department of General Services, Procurement Division and no work shall begin until such time.

During this period of performance, the Contractor(s) will receive purchase orders from both State and local governmental agencies for purchases made via the Auto Parts Contract.

### 4. CONTRACTOR RESPONSIBILITIES

- The Contractor will provide a Contract Manager (CM) that will work with the State's Contract Administrator (CA) to fully implement and manage the contract in accordance with Section 6.16.2.
- The Contractor will work closely with subcontractors (if applicable) to fully implement and manage the Auto Parts Contract.
- The Contractor shall make available to the State, technically competent personnel to accomplish the tasks and deliverables for the implementation and management of the contract. In addition, the Contractor will have a customer service unit, that is dedicated to this contract, to respond to user inquiries.

### 5. STATE RESPONSIBILITIES

- The State will provide a CA that will work with the Contractor's CM to fully implement and manage the contract.
- The State CA will be responsible for reviewing and approving each deliverable.

### 6. USER RESPONSIBILITIES

- The ordering agencies will provide a Program Manager or Point of Contact (POC) for each purchase to insure that the goods and/or services are delivered in accordance with the terms and conditions of the contract. This person will be the POC for arranging the delivery, installation, and maintenance.
- The ordering agencies will ensure the Contractor has proper access to the appropriate areas for deliveries for the Office Supplies Statewide Contract.

DGS RFP 57224 EXHIBIT 10.2

### 7. TASKS

The Contractor agrees to perform the required tasks in accordance with the Office Supplies Statewide Contract and the ordering agencies purchase order/contract.

### 8. DELIVERABLES

Contractor shall provide all deliverables identified in Section 6, Technical Requirements. Deliverables must be prepared in accordance with the RFP requirements and submitted to the CA by the dates as specified.

### 9. BILLING INFORMATION

The Contractor shall submit invoices in accordance with Invoicing.



State of California • Arnold Schwarzenegger, Governor State and Consumer Services Agency

### DEPARTMENT OF GENERAL SERVICES

707 Third Street • West Sacramento, CA 95605 • www.dgs.ca.gov

October 8, 2009

To All Bidders

# Addendum #2 to RFP 57224 Auto Parts

The addended pages of this RFP are available online under EProcurement (Bid Sync) located at <a href="https://www.bidsync.com/DPX?ac=view&auc=333831">https://www.bidsync.com/DPX?ac=view&auc=333831</a>. If you are unable to download a copy of the addenda from this website, an electronic Word version is available upon request. Changes to the RFP are attached.

The changes to the RFP are as follows:

- Section 5, page 5-7, "State of California Small Business Preference"
- Section 5, page 5-8, "DVBE Incentive"
- Section 6, page 6-6, "Minimum Order"
- Section 6, page 6-11, "Delivery Schedules"
- Section 6, page 6-11, "Delivery from Commercial Outlets"
- Section 6, page 6-12, "Free on Board (F.O.B. Destination (Shipping Terms)

Replace original pages with attached revised pages.

Sincerely,

Eileen Tardiff Master Agreements Procurement Division (916) 375-4390

email: eileen.tardiff@dgs.ca.gov

## 5.4.14 State of California DVBE Declaration (M)

Bidders who have been certified by California as a DVBE (or bidders who have obtained the participation of subcontractors certified by California as a DVBE) must complete the DVBE Declaration Form (STD 843) and include it with the proposal. A copy of this form can be downloaded at

http://www.documents.dgs.ca.gov/pd/poliproc/STD-

<u>843FillPrintFields.pdf#search=STD%20843&view=FitH&pagemode=none</u>. All disabled veteran owners and disabled veteran managers of the DVBE(s) must sign this form.

#### 5.5 State of California Preferences and Incentives

Bidders are not required to request the following preferences and incentives and the State's denial(s) of these preferences and incentives is not a basis for rejection of a proposal.

Bidders who are requesting a preference or incentive must certify they will comply with the requirements of the applicable Acts as appropriate. The State will monitor the life of a contract awarded with any applied preference or incentive for compliance to statutory, regulatory and contractual requirements. After award, the contractor is required to respond to the State's request for evidence of performance. The State will take appropriate corrective actions and may apply sanctions as necessary to enforce preference and incentive programs.

### 5.5.1 State of California Small Business Preference (NM)

(PLEASE NOTE: Small Business will not be an evaluation factor for this RFP. If the company is a Certified Small Business in the State of California, each Department purchasing parts from the company may use those purchases to help attain their 25% goal of utilizing a California Certified Small Business.)

- The regulations concerning the Small Business application and calculation of the small business preference, small business certification, responsibilities of small business, department certification can be viewed at <a href="https://www.pd.dgs.ca.gov/smbus">www.pd.dgs.ca.gov/smbus</a>. Access the regulations by clicking on "Small Business Regulations" in the right sidebar.
- 2. Non-Small Business (Non-SB) Sub-contractor Preference

A 5% bid preference is available to a Non-SB claiming 25% California certified small business sub-contractor participation. If applicable, submit a corresponding letter claiming the preference.

- 3. Non-SB if claiming the 25% preference are required to adhere to and provide with their bid the following:
  - a. If claiming the Non-SB sub-contractor preference, the bid response must include a list of the small business(es) with which the bidder commits to subcontract in an amount of at least twenty-five percent (25%) of the net bid price with one or more California certified small businesses in accordance with Government Code Section 14838(b) (1)(2).

9-16-09 5 - 7

- b. If Applicable, non-SB should provide a list of California Certified Small Business subcontractor's information with its bid response that contains the following:
  - Sub-contractor(s) name;
  - Sub-contractor(s) address;
  - Sub-contractor(s) phone number;
  - Sub-contractor(s) description of the work to be performed and/or products supplied; and
  - the sub-contractor(s) percentage amount of the net bid price (as specified in the solicitation) per sub-contractor.
- Each listed certified small business must perform a "Commercially Useful Function" (CUF) in the performance of the contract as defined in Government Code Section 14837(d)(4).

#### 4. 4. Small Business Preferences

Bidders claiming the 5% preference must be certified by California as a small business or must commit to subcontract at least 25% of the net bid price with one or more California certified small businesses. Certification must be obtained no later than 5:00 p.m. on the bid due date.

Small Businesses are desired and encouraged to participate in this bid. Section 14835, et seq. of the California Government Code requires a 5% preference be given to bidders who qualify as a small business. The rules and regulations of this law, including the definition of a small business for the delivery of goods and services, are contained in California Code of Regulations, Title 2, Section 1896, et seq. The small business preference is for California-based small businesses.

a. To claim the small business preference, which may not exceed \$50,000 for any bid, the firm must have it's principal place of business located in California, have a complete application (including proof of annual receipts) on file with the State Office of Small Business and Disabled Veteran Business Certification (OSDC) by 5:00 p.m. on the bid due date (see front page) and be verified by such office.

Bidders desiring to claim the preference must submit completed Exhibit 10.16, Small Business Preference Information and the Bidder's Declaration Form, with the proposal.

Questions regarding the preference approval process should be directed to the OSDS at (916) 375-4940. A copy of the regulations, instructions and format for claiming the small business preference is available at <a href="http://www.pd.dgs.ca.gov/smbus/default.htm">http://www.pd.dgs.ca.gov/smbus/default.htm</a>.

### 5.5.2 DVBE Incentive (NM)

(PLEASE NOTE: DVBE will not be an evaluation factor for this RFP. If the company is a DVBE in the State of California, each Department purchasing parts from the company may use those purchases to help attain their 3% goal of utilizing a DVBE.)

For this RFP, DVBE has been waived.

9-16-09 5 - 8

Ordering	1			
	Ordering  The contractor(s) shall accept orders from any State Department or local governmental agency. Contractor(s) must not refuse to accept orders from any State Department or local governmental agency without written authorization from the State Contract Administrator.	М		
	Purchase Execution  California State Departments will submit orders on a Purchasing Authority Purchase Order (Std. 65) via one of the ordering methods as specified in Section 6, Ordering Methods.  Local governmental agencies may submit orders on their own purchase document directly to the contractor via one of the ordering methods as specified in Section 6, Ordering Methods. However, local California governmental agencies must obtain a State issued billing code prior to submitting the order.  Contractor(s) shall not accept purchase documents from State departments or local governmental agencies that are incomplete or contain non-contract items.	M		
	Minimum Order  There is no minimum order on this contract.  For orders outside of the 30 mile radius, a \$150.00 minimum order will apply for Free Shipping.	M		
	Ordering Methods  Contractor(s) shall accept orders through the following methods:  • Electronic (email) • Phone • Facsimile • Mail • Internet/Web	M		

9-16-09 6 - 6

Delivery				
	Delivery Locations  Deliveries are to be made to the location specified on the individual order, which may include, but not be limited to inside buildings, high-rise office buildings, and receiving docks.	М		
	Delivery Schedules  Delivery of ordered product shall be completed within three (3) working days after receipt of an order, unless otherwise agreed to by the ordering agency. Since receiving hours for each ordering agency will vary by facility, it will be the Contractor's responsibility to check with each facility for their specific receiving hours before delivery occurs.  Contractor(s) must notify the ordering agency within 12 hours of scheduled delivery time, if delivery cannot be made within the time frame specified in the contract.	М		
	Delivery from Commercial Outlets  Delivery of standard in-stock items to locations within five (5) miles from company's commercial store shall be accomplished within one (1) hour. Delivery outside the five (5) mile range will be accomplished by a scheduled delivery.	M		
	Security Requirements for Institutions  Deliveries may be made to locations inside secure institutional grounds (such as the California State Prisons) that require prior clearances to be made for delivery drivers.  The Contractor(s) will be responsible for contacting the secure location for security clearance procedures, hours of operation for deliveries and service, dress code, and other rules of delivery. These Security clearance procedures may vary from facility to facility.  Deliveries that are delayed due to drivers not being cleared to enter institutional grounds may cause the contractor to be declared in default of contract requirements.	M		

9-16-09 6 - 11

М			
	M	M	M

9-16-09 6 - 12



State of California • Arnold Schwarzenegger, Governor State and Consumer Services Agency

### DEPARTMENT OF GENERAL SERVICES

707 Third Street • West Sacramento, CA 95605 • www.dgs.ca.gov

October 14, 2009

To All Bidders

### Addendum #3 to RFP 57224 Auto Parts

The addended pages of this RFP are available online under EProcurement (Bid Sync) located at <a href="https://www.bidsync.com/DPX?ac=view&auc=333831">https://www.bidsync.com/DPX?ac=view&auc=333831</a>. If you are unable to download a copy of the addenda from this website, an electronic Word version is available upon request. Changes to the RFP are attached.

The changes to the RFP are as follows:

- Section 1, page 1-5, 1-6, "Key Action Dates"
- Section 2, page 2-12, "Sealed Cost/Cost Evaluation"
- Section 5, page 5-3, "Contractor Responsibilityr" Section 5, page 5-4, "State of California Bidder's Declaration"
- Section 5, page 5-4, "Subcontractors (State of California)"
- Section 5, page 5-5, "Notice of Subcontractors (State of California)"
- Section 5, page 5-5, "State of California Commercially Useful Function"
- Section 5, page 5-7, "State of California Preferences and Incentives"
- Section 6, page 6-3, "Product Listing"
- Section 6, page 6-3, "Web Access"
- Section 6, page 6-4, "Submission and Approval""
- Section 7, page 7-2, "Pricing Structure"
- Section 7, page 7-3, "Important Details"
- Section 9, page 9-2, "State Evaluation Team"

Please note, catalogs are no longer required. Details are listed within the corrected sections. Also note that some Key Action Dates have changed.

Replace original pages with attached revised pages.

Sincerely,

Eileen Tardiff Master Agreements Procurement Division (916) 375-4390

email: eileen.tardiff@dgs.ca.gov

Department of General Services RFP 57224

# SECTION 1 Introduction and Overview of Requirements

### 1.6 Key Action Dates

Listed below are the important dates and times related to this RFP. If the State finds it necessary to change any of these dates before the Final Proposal submission, it will be accomplished by an addendum. All dates after the Final Proposal submission deadlines are approximate and may be adjusted as conditions indicate without addendum to the RFP.

Unless otherwise indicated below, Bidder submissions shall be due by the close of business on the date indicated. Close of business day shall be 5:00 p.m. All times are Pacific Time.

No.	Event/Action	Action Date	Time
1.	RFP Release	9/16/09	
2.	Last day to submit questions for clarification of RFP for Bidders' Conference	9/23/09	5:00 pm (PDT)
3.	Last day to submit Letter of Intent to Bid	9/28/09	5:00 pm (PDT)
4.	Bidders' Conference  Location: Ziggurat Building Department of General Services 707 Third Street, 2nd Floor West Sacramento, California 95605  See Section 2.2.6, Bidders' Conference for more information.	9/30/09	1:00 pm (PDT)
5.	Last day to submit final questions for clarification of RFP prior to submittal of Draft Proposals	10/5/09	5:00 pm (PDT)
6.	Last day to request a change in the requirements of the RFP	10/5/09	5:00 pm (PDT)
7.	Response to Bidders' questions released	10/8/09	5:00 pm (PDT)
8.	Last day to protest the RFP requirements	10/13/09	5:00 pm (PDT)
9.	Submission of <u>Draft</u> Proposals*	10/20/09	5:00 pm (PDT)
10.	Confidential Discussions with Individual Bidders	10/26/09	9:00 am to 5:00 pm (PDT)
11.	Submission of Final Proposals	10/30/09	5:00 pm (PDT)
12.	Posting of companies passing Phase I	11/12/09	
13.	Cost Evaluations begin	11/16/09	
14.	Cost Evaluations Posted	11/23/09	

10-14-09 1 - 5

# SECTION 1 Introduction and Overview of Requirements

No.	Event/Action	Action Date	Time
15.	Intent To Award	11/30/09	
16.	Last day to submit Notice of Intent to Protest Award	12/7/09	5:00 pm (PDT)
17.	Contract Award	12/8/09	
18.	Start Date of Agreement	1/15/10	

Additional action dates may be inserted as needed.

### 1.7 Americans with Disabilities Act

Following is the Policy of the Procurement Division of the State Department of General Services to Comply with Title II of the Americans with Disabilities Act (ADA) Compliance Policy of Nondiscrimination on the Basis of Disability.

To meet and carry out compliance with the nondiscrimination requirements of Title II, Americans with Disabilities Act (ADA), it is the policy of the Procurement Division (within the State Department of General Services) to make every effort to ensure that its programs, activities, employment opportunities, and services are available to all persons, including persons with disabilities.

For persons with a disability needing reasonable accommodation to participate in the Procurement process, or for persons having questions regarding reasonable accommodation for the Procurement process, please contact the Procurement Division at 916-375-4400 (main office). Procurement Division's TTY/TDD (telephone device for the deaf) and California Relay Service numbers are listed below. The bidder may also directly contact the Procurement Official identified in 1.5 above.

IMPORTANT: To ensure that the bidder's accommodation can be met, it is best that requests be submitted at least ten (10) working days before the scheduled event (i.e., bidder conference) or deadline due-date for procurement documents.

The Procurement Division TTY telephone number is:

Sacramento Office: (916) 376-1891

The California Relay Service Telephone Numbers are:

Voice: 1-800-735-2922 or 1-888-877-5379

TTY: 1-800-735-2929 or 1-888-877-5378

Speech to Speech: 1-800-854-7784

10-14-09 1 - 6

<sup>\*</sup> Although recommended, this key action date is an optional step in the RFP process.

# SECTION 2 Rules Governing Competition

### 2.3.6 Demonstrations (If required)

Not applicable

### 2.3.7 Sealed Cost/Cost Evaluation

Final Proposals will not have their sealed cost envelopes opened until the State has evaluated the technical and administrative submission. Cost Proposals will only be opened for responsive proposals from responsible Bidders. On the date cost evaluation begins, costs will be opened by the evaluation team. After the evaluation is computed and complete, the State will post a summary of the evaluation of each Bidder. This summary will be provided to all the Bidders and posted in the lobby for additional viewing.

### 2.3.8 Rejection of Proposals

The State may reject any or all proposals and may waive any immaterial deviation or defect in a proposal. The State's waiver of any immaterial deviation or defect shall in no way modify the RFP documents or excuse the Bidder from full compliance with the RFP specifications if awarded the contract.

#### 2.3.8.1 General

Final Proposals will be evaluated according to the procedures contained in Section 9, Evaluation.

#### 2.3.8.2 Evaluation Questions

During the evaluation and selection process, the State may require a Bidder's representative to answer specific questions and provide clarification, as long as the answers and clarification are in writing.

### 2.3.8.3 Errors in the Final Proposal

Errors in the Final Proposal may cause the rejection of that proposal. However, the State, in its sole discretion, may retain the proposal and make certain corrections. In determining if a correction will be made, the State will consider the conformance of the proposal to the format and content required by the RFP and any unusual complexity of the format and content required by the RFP.

The State, at its sole discretion, may correct obvious clerical errors.

The State, in its sole discretion, may correct discrepancies and arithmetic errors. If intent is not clearly established by the complete proposal submittal, the Master Hard Copy shall have priority over additional copies. The Proposal Master Hard Copy narrative shall have priority over the cost sheets. If necessary, the extensions and summary will be re-computed from the lowest level of detail, even if the lowest level of detail is obviously misstated. If a Bidder does not follow the instructions for computing costs which are not related to the contract (e.g., State personnel costs), the State

9-16-09 2 - 12

### **Non-Mandatory Requirements (NM):**

Some listed standards are identified as "Non-Mandatory" or "NM". Bidders are not required to comply with these requirements in order to be compliant with the RFP requirements. However, if a Bidder offers any of these non-mandatory requirements, the Bidder must meet the minimum requirements at stated in this section. Non-Mandatory items will not be considered when identifying Bidders who are responsive and responsible.

As specified with each non-mandatory requirement listed below, Bidders must:

- Indicate whether their proposal meets the individual requirement by marking either a "Yes" or "No" on Exhibit 10.5, Administrative Requirements Response; and
- Provide a narrative of the Bidder's compliance and/or response for each item.

### 5.4 Administrative Requirements

### 5.4.1 Bidder Responsibility (M)

Prior to award of the contract, the State must be assured that the Bidder(s) selected has all of the resources to successfully perform under the contract. This includes, but is not limited to, personnel in the numbers and with the skills required, equipment of appropriate type and in sufficient quantity, financial resources sufficient to complete performance under the contract, and experience in similar endeavors. If, during the evaluation process, the State is unable to assure itself of the Bidder's ability to perform under the contract, if awarded, the State has the option of requesting from the Bidder any information that the State deems necessary to determine the Bidder's responsibility. If such information is required, the Bidder will be so notified and will be permitted approximately five (5) working days to submit the information requested.

The Bidder must provide the State with sufficient information to allow the State to confirm the Bidder's ability to perform successfully under the contract.

### 5.4.2 Confidentiality (M)

Not Applicable

### 5.4.3 Contractor Responsibility (M)

The State requires that the contract resulting from RFP DGS 57224 be issued to multiple contractors who shall be responsible for successful performance of this RFP. A prime contractor shall be responsible for successful performance of any subcontractors and will be required to assume responsibility for design, development, testing, and implementation of all systems in their proposal.

Furthermore, the State will consider the prime contractor(s) to be the sole point of contact with regard to contractual matters, payment of any and all charges resulting from the outsource or purchase of the equipment and maintenance of the equipment for the term of the contract.

The contractor(s) will be responsible for compliance with requirements under the contract, even if requirements are delegated to subcontractors. All State policies, guidelines, and requirements apply to subcontractors. The contractor(s) and subcontractors shall not in any way represent themselves in the name of the State of California without prior written approval.

### 5.4.4 Contractor Name Change (M)

An amendment is required to change the contractor's name as listed on this agreement. Upon receipt of legal documentation of the name change the State will process the amendment. Payment of invoices presented with a new name cannot be paid prior to approval of said amendment.

### 5.4.5 State of California Bidder Declaration (M)

All bidders must complete the Bidder's Declaration form. When completing the declaration, bidders must identify all sub-contractors proposed for participation in the contract. Bidders awarded a contract are contractually obligated to use the sub-contractors for the corresponding work identified, unless the State agrees to a substitution.

At the State's option prior to award, bidders may be required to submit additional written clarifying information. Failure to submit the requested written information may be grounds for bid rejection.

Refer to <a href="http://www.documents.dgs.ca.gov/pd/delegations/GSPD105.pdf">http://www.documents.dgs.ca.gov/pd/delegations/GSPD105.pdf</a> for a downloadable version of this document.

### 5.4.5.1 Subcontractors (State of California) (M)

If the Bidder's proposal involves the use of subcontractors for a portion of the contract pertaining to a California Certified Small Business or a California Certified Disabled Veteran, the Bidder must explain how they will manage and control the work of those subcontractors.

The Bidder must agree that all requirements of those subcontractors will be adhered to and that requirements will apply to subcontractors even if subcontractor concurrence is not specifically defined in the Administrative Requirements. All subcontractors must comply fully with the administrative and technical requirements that are applicable with the portion of the work being delegated to the subcontractor.

Bidders awarded a contract are contractually obligated to use those subcontractors for the corresponding work identified unless the State agrees to a substitution and it is incorporated by amendment to the contract. The contractor must have written agreement from the State prior to replacement or substitution of any subcontractor.

# 5.4.5.2 Notice of Subcontractors (State of California) (M)

Upon award to a contractor requesting Small Business or DVBE participation, notice shall be given by the State to the California Certified Small Business or a California Certified Disabled Veteran listed in Section 5.4.5, Bidder Declaration Form, of their participation in the contract. Notification to the subcontractor by the prime contractor is encouraged immediately after award of an RFP.

There will be no assignment of responsibility to these subcontractors without prior written approval from the State of the Small Business or DVBE.

### 5.4.6 State of California Commercially Useful Function (M)

Suppliers, whether the Bidder or a subcontractor, who have a California certification for one or more of the socio-economic programs (e.g., Small Business or Disabled Veteran Business Enterprise), must perform a commercially useful function (CUF) in the resulting contract. CUF is defined in the Military and Veterans Code Section 999(b)(5)(B) for Disabled Veteran Business Enterprises and in the Government Code Section 14837(d)(4)(A) for Small Business. Bidders must describe compliance with CUF requirement.

At the State's option, Bidders may be required to submit additional written clarification regarding CUF. Failure to submit the requested written information as specified may be grounds for bid rejection.

### 5.4.7 Non-Collusion Affidavit (M)

Not applicable to this RFP

### 5.4.8 Payee Data Record (M)

All Bidders must complete a Payee Data Record, and include it with the proposal. A copy of this form is available for download at <a href="http://www.documents.dgs.ca.gov/osp/pdf/std204.pdf">http://www.documents.dgs.ca.gov/osp/pdf/std204.pdf</a>.

#### 5.4.9 State of California Seller's Permit (M)

This RFP is subject to all requirements set forth in Sections 6452, 6487, 7101 and 18510 of the Revenue and Taxation Code, and Section 10295 of the Public Contract Code, requiring suppliers to provide a copy of their retailer's seller's permit or certification of registration, and, if applicable, the permit or certification of all participating affiliates issued by the State of California's Board of Equalization.

Bidders are required to submit a copy of their company's Retailer's Seller's Permit, and include it with the proposal.

## 5.4.14 State of California DVBE Declaration (M)

Bidders who have been certified by California as a DVBE (or bidders who have obtained the participation of subcontractors certified by California as a DVBE) must complete the DVBE Declaration Form (STD 843) and include it with the proposal. A copy of this form can be downloaded at

http://www.documents.dgs.ca.gov/pd/poliproc/STD-

<u>843FillPrintFields.pdf#search=STD%20843&view=FitH&pagemode=none</u>. All disabled veteran owners and disabled veteran managers of the DVBE(s) must sign this form.

#### 5.5 State of California Preferences and Incentives

Bidders are not required to request the following preferences and incentives and the State's denial(s) of these preferences and incentives is not a basis for rejection of a proposal.

Bidders who are requesting a preference or incentive must certify they will comply with the requirements of the applicable Acts as appropriate. The State will monitor the life of a contract awarded with any applied preference or incentive for compliance to statutory, regulatory and contractual requirements. After award, the contractor is required to respond to the State's request for evidence of performance. The State will take appropriate corrective actions and may apply sanctions as necessary to enforce preference and incentive programs.

Claimed preferences will apply to the State of California purchases after award. Due to this being a multi-state contract, preferences will not apply toward the evaluation of the contract.

### 5.5.1 State of California Small Business Preference (NM)

(PLEASE NOTE: Small Business will not be an evaluation factor for this RFP. If the company is a Certified Small Business in the State of California, each Department purchasing parts from the company may use those purchases to help attain their 25% goal of utilizing a California Certified Small Business.)

- 1. The regulations concerning the Small Business application and calculation of the small business preference, small business certification, responsibilities of small business, department certification can be viewed at <a href="https://www.pd.dgs.ca.gov/smbus">www.pd.dgs.ca.gov/smbus</a>. Access the regulations by clicking on "Small Business Regulations" in the right sidebar.
- 2. Non-Small Business (Non-SB) Sub-contractor Preference

A 5% bid preference is available to a Non-SB claiming 25% California certified small business sub-contractor participation. If applicable, submit a corresponding letter claiming the preference.

- 3. Non-SB if claiming the 25% preference are required to adhere to and provide with their bid the following:
  - a. If claiming the Non-SB sub-contractor preference, the bid response must include a list of the small business(es) with which the bidder commits to subcontract in an amount of at least twenty-five percent (25%) of the net bid price with one or more California certified small businesses in accordance with Government Code Section 14838(b) (1)(2).

## **Non-Mandatory Requirements (NM):**

Some listed standards are identified as "Non-Mandatory" or "NM". Bidders are not required to comply with these requirements in order to be compliant with the RFP requirements. However, if a Bidder offers any of these non-mandatory requirements, the Bidder must meet the minimum requirements as stated in this section. Non-Mandatory items will not be considered when identifying Bidders who are responsive and responsible.

As specified with each "Non-Mandatory" or "NM" requirement listed below, Bidders must indicate whether their proposal meets the individual requirement by marking either a "Yes" or "No" in the column next to the requirement, provide the location of a thorough narrative response as specified in Section 8, Proposal Format.

### **Technical Requirements**

Section	Requirement Description	Туре	<b>Me</b> Yes	 Proposal Page Number
Products	5			
	Product Listing (Cost Worksheet)	M		
	The products included as part of this RFP are listed on Exhibit 10.9, Cost Worksheet. Cost Worksheet Light Duty Exhibit 10.9 must be completed as specified in Section 7, Cost.			
	The State reserves the right to eliminate any specific categories from the final award, if indicated by an analysis of RFP Proposals.			
WEB AC	CESS			
		M		
	Each bidder is required to have a current, established web ordering system. The ordering system shall include any and all parts available from the bidder which are covered in the Cost Sheets (Exhibit 10.9).			
	The products on the webpage must have:			
	Product name			
	<ul> <li>Description of the product</li> </ul>			
	Product number			
	Product Price			
	The contractor shall not have any catalog viewing restrictions for State departments or local governmental agencies trying to access the electronic catalogs.			

10-14-09 6 - 3

Submission and Approval	М		
Within thirty (30) calendar days of proposed award, the contractor must submit a preliminary version of the web and suggested website location to the State Contract Administrator for review. The web must have the contract discounts embedded into the invoiced pricing (WSCA pricing) so when agencies order, their final price is shown on the webpage.			
The State will review and determine acceptability of the website and website location. Once the electronic version has been approved by the State, the State shall provide written authorization to post versions on contractor's website.			
EXTRA INFORMATION	M		
Provide details regarding the contractor's approach to maintaining the up-to-date information in both the contractor's on-line and hard copy catalogs. Describe the frequency of updates and how these updates are communicated to eligible users.			

10-14-09 6 - 4

### 7. Cost

#### 7.1 Introduction

Cost is the primary evaluation criterion for the award for this RFP. Cost evaluation will be based on the highest discount (%) to the State, as calculated according to the methodology outlined in Section 9, Evaluation. All proposed discounts (%) for all Categories must include the cost of any services required in this RFP.

The State's intent is to structure the pricing format in order to facilitate a straightforward comparison among all bidders and foster competition to obtain the best market pricing. Consequently, the State requires that each bidder's quote be in the format outlined in this section. Therefore, Bidders are advised that failure to comply with the instructions listed in this section, such as submission of incomplete offers or use of alternative pricing structures or different formats than the one requested may result in the rejection of their proposals for non-responsiveness.

Note: It is imperative that cost information not be included in the body of the proposal. Cost information is to be submitted in accordance with Section 8, Proposal Format.

### 7.2 Pricing Structure

The pricing information is categorized under eighteen (18) main Categories.

The pricing proposal for this Category list will only be considered compliant if the bidder provides pricing for all Categories listed. A blank Category will be interpreted as a noncompliant pricing proposal and the bidder will be deemed non responsive.

The intent of this RFP is to award on a multiple-award basis. All mandatory Categories must be bid or the proposal will be considered non-responsive and will be rejected.

#### 7.3 Quantities

No quantities will be shown in Exhibit 10.9 Cost Worksheet, as the State has no estimate of the anticipated contract purchase pattern. Actual purchases may vary and the State will not guarantee any quantities will be purchased. Consequently, there is no guaranteed dollar value for this contract.

#### 7.4 Worksheet Instructions

Exhibit 10.9 Cost Worksheet Light Duty contains the worksheet that the Bidders shall use to enter cost information for Light Duty Cars and Trucks. The Light Duty Cost Worksheet **must** be filled out completely or the proposal will be rejected.

Bidders must follow the instructions below to complete the Cost Worksheet:

10-14-09 7 - 2

### 7.4.1 Completion of Worksheet

For Categories 1-18 contained in this worksheet, please input the following for each Category (unless otherwise instructed):

• The Discount (%) being offered to the State for that Category.

The minimum discount offered in the proposal is fixed for the life of the contract. However, the contractor can offer greater discounts at any time during the course of the contract.

### 7.4.2 Important details

- Internet Web Prices are to be those prices used for evaluation. The web prices will be pulled from each compliant bidder on the same day, same time for each individual part being evaluated.
- 2) The website provided shall consist of a publicly accessible on-line catalog and must contain a full range of auto parts in all categories.

#### 7.5 Cost Submittal

Final pricing will be submitted in a separate sealed envelope as specified in Section 8, Proposal Format.

Final pricing will be submitted in a hard copy version and CD-ROM format in a sealed envelope as part of the final proposal, refer to Section 3.2.3 and 3.2.5, Format and Contact. Please ensure that a whole number percentage discount (%) with no decimal places is entered as indicated in the instructions and examples for the pricing worksheets. The number that is submitted on the pricing CD as part of the final proposal, will be used for evaluation purposes.

### 7.6 Additional Savings

For the purpose of cost, if at any time a commercial sale price anywhere in the nation is offered to the public and is cheaper than a contract price, the sale price must be offered to the State in lieu of the contracted discount price.

10-14-09 7 - 3

#### 9. Evaluation

#### 9.1 Introduction

This Section describes how the State will evaluate the proposals and identify those proposals that meet the RFP's objectives and provide the best value for the State. It is the State's intent to conduct a comprehensive, impartial evaluation of all proposals received. The State will use a pass/fail and final cost, two-envelope method of selection.

#### 9.2 State Evaluation Team

The State has established a multi-state sourcing team that will be responsible for the review and evaluation of Bidder proposals in accordance with the process described in the RFP. The State may engage additional qualified individuals during the process to assist the State Evaluation Team in understanding technical, financial, legal, contractual, or program matters.

Evaluation Criteria: All Mandatory Administrative and Technical Requirements listed in Section 5 and Section 6 of RFP DGS-57224 will be scored on a pass/fail basis. All costs submitted per Section 7 Cost for RFP DGS- 57224 will be evaluated by the following criteria:

The State Evaluation Team will select a representative sample of like or comparable items from each category for each bidder. These samples will be referred to as the Market Basket for each bidder (see Exhibit 10.19). Each State which has a member on the evaluation team, has submitted multiple vehicles from their individual State fleets to evaluate. A part from each category will be selected for evaluation for each vehicle. To ensure fairness to all bidders, each part for each vehicle will be selected at the same time from each of the bidder's internet ordering system. The State will not be responsible for any price changes made to the internet websites during the evaluation process. The Evaluation Team's Subject Matter Experts will ensure that the parts evaluated from each bidder are comparable (to ensure fairness to each bidder). The State Evaluation Team will arrive at a final net cost to the State for each item listed in each Bidder's Market Basket, by applying the appropriate discount listed for each category. All items listed in each Bidder's Market Basket will then be totaled to arrive at a Grand Total. All bidder's Grand Totals that fall within the established 20% competitive range and are in full compliance with all the requirements of RFP DGS- 57224 will then be considered for award.

### 9.3 Receipt

Proposals must be delivered by the date and time stated in RFP Section 1.6, Key Action Dates. Each Proposal will be dated and time marked as it is received, and verified that it is properly sealed. Proposals will remain sealed until the designated time for opening. Final Proposals received after the date and time specified in RFP Section 1.6, Key Action Dates, for receipt of Final Proposals will be deemed non-responsive and will be rejected.

### 9.4 Negotiations

Not Applicable to this RFP

10-14-09 9 - 2



State of California • Arnold Schwarzenegger, Governor State and Consumer Services Agency

### DEPARTMENT OF GENERAL SERVICES

707 Third Street • West Sacramento, CA 95605 • www.dgs.ca.gov

October 27, 2009

To All Bidders

# Addendum #4 to RFP 57224 Auto Parts

The addended pages of this RFP are available online under EProcurement (Bid Sync) located at <a href="https://www.bidsync.com/DPX?ac=view&auc=333831">https://www.bidsync.com/DPX?ac=view&auc=333831</a>. If you are unable to download a copy of the addenda from this website, an electronic Word version is available upon request. Changes to the RFP are attached.

The changes to the RFP are as follows:

- Due date has been modified. The original due date for Final Proposals was October 30, 2009. The new due date is November 4, 2009. Replace page 1-5 with the attached page 1-5.
- Section 6, pages 6-24 6-27 (EPP Requirements); all requirements removed; delete pages from RFP.

Replace original pages with attached revised pages.

Sincerely,

Eileen Tardiff Master Agreements Procurement Division (916) 375-4390

email: eileen.tardiff@dgs.ca.gov

# SECTION 1 Introduction and Overview of Requirements

### 1.6 Key Action Dates

Listed below are the important dates and times related to this RFP. If the State finds it necessary to change any of these dates before the Final Proposal submission, it will be accomplished by an addendum. All dates after the Final Proposal submission deadlines are approximate and may be adjusted as conditions indicate without addendum to the RFP.

Unless otherwise indicated below, Bidder submissions shall be due by the close of business on the date indicated. Close of business day shall be 5:00 p.m. All times are Pacific Time.

No.	Event/Action	Action Date	Time
1.	RFP Release	9/16/09	
2.	Last day to submit questions for clarification of RFP for Bidders' Conference	9/23/09	5:00 pm (PDT)
3.	Last day to submit Letter of Intent to Bid	9/28/09	5:00 pm (PDT)
4.	Bidders' Conference  Location: Ziggurat Building Department of General Services 707 Third Street, 2nd Floor West Sacramento, California 95605  See Section 2.2.6, Bidders' Conference for more information.	9/30/09	1:00 pm (PDT)
5.	Last day to submit final questions for clarification of RFP prior to submittal of Draft Proposals	10/5/09	5:00 pm (PDT)
6.	Last day to request a change in the requirements of the RFP	10/5/09	5:00 pm (PDT)
7.	Response to Bidders' questions released	10/8/09	5:00 pm (PDT)
8.	Last day to protest the RFP requirements	10/13/09	5:00 pm (PDT)
9.	Submission of <u>Draft</u> Proposals*	10/20/09	5:00 pm (PDT)
10.	Confidential Discussions with Individual Bidders	10/26/09	9:00 am to 5:00 pm (PDT)
11.	Submission of Final Proposals	11/4/09	5:00 pm (PDT)
12.	Posting of companies passing Phase I	11/12/09	
13.	Cost Evaluations begin	11/16/09	
14.	Cost Evaluations Posted	11/23/09	

10-27-09 1 - 5



State of California • Arnold Schwarzenegger, Governor State and Consumer Services Agency

### DEPARTMENT OF GENERAL SERVICES

707 Third Street • West Sacramento, CA 95605 • www.dgs.ca.gov

November 3, 2009

To All Bidders

# Addendum #5 to RFP 57224 Auto Parts

The addended pages of this RFP are available online under EProcurement (Bid Sync) located at <a href="https://www.bidsync.com/DPX?ac=view&auc=333831">https://www.bidsync.com/DPX?ac=view&auc=333831</a>. If you are unable to download a copy of the addenda from this website, an electronic Word version is available upon request. Changes to the RFP are attached.

The changes to the RFP are as follows:

 Nondiscrimination language in the WSCA Terms and Conditions has been revised with the attached page.

Replace original pages with attached revised pages.

Sincerely,

Eileen Tardiff Master Agreements Procurement Division (916) 375-4390

email: eileen.tardiff@dgs.ca.gov

DGS RFP 57224 Exhibit 10.15

by the contractor. Responsibility and liability for loss or damage shall remain the Contractor until final inspection and acceptance when responsibility shall pass to the Buyer except as to latent defects, fraud and Contractor's warranty obligations. The minimum shipment amount will be found in the special terms and conditions. Any order for less than the specified amount is to be shipped with the freight prepaid and added as a separate item on the invoice. Any portion of an order to be shipped without transportation charges that is back ordered shall be shipped without charge.

**WARRANTY:** As used herein "Buyer" refers to any WSCA state agency or political subdivision. The contractor acknowledges that the Uniform Commercial Code applies to this contract. In general, the contractor warrants that: (a) the product will do what the salesperson said it would do, (b) the product will live up to all specific claims that the manufacturer makes in their advertisements, (c) the product will be suitable for the ordinary purposes for which such product is used, (d) the product will be suitable for any *special purposes* that the Buyer has relied on the contractor's skill or judgment to consider.

**AMENDMENTS:** The terms of this contract shall not be waived, altered, modified, supplemented or amended in any manner whatsoever without prior written approval of the WSCA Contract Administrator.

**ASSIGNMENT/SUBCONTRACT:** Contractor shall not assign, sell, transfer, subcontract or sublet rights, or delegate responsibilities under this contract, in whole or in part, without the prior written approval of the WSCA Contract Administrator.

**NONDISCRIMINATION:** The offeror agrees to abide by the provisions of Title VI and Title VII of the Civil Rights Act of 1964 (42 USC 2000e), which prohibit discrimination against any employee or applicant for employment, or any applicant or recipient of services, on the basis of race, religion, color, or national origin; and further agrees to abide by Executive Order No. 11246, as amended, which prohibits discrimination on basis of sex; 45 CFR 90 which prohibits discrimination on the basis of age, and Section 504 of the Rehabilitation Act of 1973, or the Americans with Disabilities Act of 1990 which prohibits discrimination on the basis of disabilities. The offeror further agrees to furnish information and reports to requesting State(s), upon request, for the purpose of determining compliance with these statutes. Offeror agrees to comply with each individual state's certification requirements, if any, as stated in the special terms and conditions. This contract may be canceled if the offeror fails to comply with the provisions of these laws and regulations. The offeror must include this provision in very subcontract relating to purchases by the States to insure that subcontractors and vendors are bound by this provision.

**SEVERABILITY:** If any provision of this contract is declared by a court to be illegal or in conflict with any law, the validity of the remaining terms and provisions shall not be affected; and the rights and obligations of the parties shall be construed and enforced as if the contract did not contain the particular provision held to be invalid.

**INSPECTIONS:** Goods furnished under this contract shall be subject to inspection and test by the Buyer at times and places determined by the Buyer. If the Buyer finds goods furnished to be incomplete or in compliance with proposal specifications, the Buyer may reject the goods and require Contractor to either correct them without charge or deliver them at a reduced price, which is equitable under the circumstances. If Contractor is unable or refuses to correct such goods within a time deemed reasonable by the Buyer, the Buyer may cancel the order in whole or in part. Nothing in this paragraph shall adversely affect the Buyer's rights including the rights and remedies associated with revocation of acceptance under the Uniform Commercial Code.

**PAYMENT:** Payment for completion of an contract is normally made within 30 days following the date the entire order id delivered or the date a correct invoice is received, whichever is later. After 45 days the Contractor may assess overdue account charges up to a maximum rate of one percent per month on the outstanding balance. Payments will be remitted by mail. Payments may be made via a State or political subdivision "Purchasing Card".

DGS RFP 57224 EXHIBIT 10.1
Letter of Intent to Bid

#### **LETTER OF INTENT TO BID**

Eileen Tardiff
Procurement Division
Department of General Services
707 Third Street, 2<sup>nd</sup> Floor
West Sacramento, CA 95605

Reference: RFP DGS 57224, Auto Parts

This is to notify you that it is our present intent to (submit/not submit\*) a proposal in response to the above referenced RFP number. The individual to whom information regarding this RFP should be transmitted is:

	Name:		
	Company:		
	Address:		
	City, State & Zip:		
	Phone Number:	( )	
	Facsimile Number:	( )	
	E-mail Address:		
□ I / We con Sincerely,	cur with the proposed o	ontract language as presented ir	the RFP.
Omocrety,			
Name (Signa	ture)		Date
Typed Name	& Title of Representati	/e	
Typed Name	of Company		
*If declining to	o bid, please state rea	son(s) why:	

#### STATEMENT OF WORK

#### 1. INTRODUCTION

This Statement of Work ("Agreement") reflects the goods and services to be provided by the Contractor hereinafter referred to as the "Contractor," for the State of California hereinafter referred to as the "State" for the Office Supplies Statewide Contract. Section 6, Technical Requirements shall be incorporated as a part of this Statement of Work along with all attachments and exhibits.

#### 2. SCOPE

The Contractor will provide the entire portfolio of quality office supply products to State departments and local governmental agencies in a timely and efficient manner. The Contractor will be the primary point for warehousing, data collection, reporting, and distribution of office supplies to the State.

#### 3. PERIOD OF PERFORMANCE

The term of this Agreement is effective for three (3) years from the date of approved contract document with the option to extend two (2) additional years in various increments upon approval by the Department of General Services, Procurement Division and no work shall begin until such time.

During this period of performance, the Contractor(s) will receive purchase orders from both State and local governmental agencies for purchases made via the Office Supplies Statewide Contract.

#### 4. CONTRACTOR RESPONSIBILITIES

- The Contractor will provide a Contract Manager (CM) that will work with the State's Contract Administrator (CA) to fully implement and manage the contract in accordance with Section 6.16.2.
- The Contractor will work closely with subcontractors (if applicable) to fully implement and manage the Office Supplies Statewide Contract.
- The Contractor shall make available to the State, technically competent personnel to accomplish the tasks and deliverables for the implementation and management of the contract. In addition, the Contractor will have a customer service unit, that is dedicated to this contract, to respond to user inquiries.

#### 5. STATE RESPONSIBILITIES

- The State will provide a CA that will work with the Contractor's CM to fully implement and manage the contract.
- The State CA will be responsible for reviewing and approving each deliverable.

#### 6. USER RESPONSIBILITIES

• The ordering agencies will provide a Program Manager or Point of Contact (POC) for each purchase to insure that the goods and/or services are delivered in accordance with the terms and conditions of the contract. This person will be the POC for arranging the delivery, installation, and maintenance.

• The ordering agencies will ensure the Contractor has proper access to the appropriate areas for deliveries for the Office Supplies Statewide Contract.

#### 6. TASKS

The Contractor agrees to perform the required tasks in accordance with the Office Supplies Statewide Contract and the ordering agencies purchase order/contract.

#### 7. DELIVERABLES

Contractor shall provide all deliverables identified in Section 6, Technical Requirements. Deliverables must be prepared in accordance with the RFP requirements and submitted to the CA by the dates as specified in Section 6.18.4.

#### 8. BILLING INFORMATION

The Contractor shall submit invoices in accordance with RFP Section 6.13.1, Invoicing.

STATE OF CALIFORNIA

STANDARD AGREEMENT

STD. 213 (NEW 06/03)

#### STANDARD AGREEMENT

			REGISTRATION NUMBER
		PURCHASING AUTHORITY NUMBER	AGREEMENT NUMBER
1.	This Agreement is entered into between the State Ag	gency and the Contractor named	below
	STATE AGENCY'S NAME	(horos	after called State)
		(Herea	arter caried State)
	CONTRACTOR'S NAME		(hereafter called
2.	The term of this		
	Agreement is:		
3.	The maximum amount of this Agreement is: \$ 0.00 (No guarantee of con	ntract expenditure)	
	, , , , , , , , , , , , , , , , , , ,	. ,	
4.	The parties agree to comply with the terms and concreference made a part of the Agreement:	litions of the following attachmen	its which are by this
	reference made a part of the Agreement.		

- Administrative Requirements, Section 5 of the RFP DGS-57224 (DGS will attach to the bidder's contract)
- Business and Technical Requirements, Section 6 of the RFP DGS-57224 (DGS will attach to the bidder's contractd)
- Final Pricing Worksheet from RFP DGS-57224 (DGS will attach to the bidder's contract)
- General Provisions (GSPD-401 non-IT Commodities, Revised and effective 04/12/2007). http://www.documents.dgs.ca.gov/pd/modellang/GPnonIT0407.pdf

Bidder's entire proposal and the entire RFP DGS-57224 are hereby incorporated by reference and made a part of this contract. All purchase orders issued under this contract incorporates the terms and conditions set forth in this contract.

IN WITNESS WHEREOF, this Agreement has been executed by the parties hereto.

CONTRACTOR	Department of General Services Use Only	
CONTRACTOR'S NAME (If other than an individual, state whether a c	orporation, partnership, etc.)	•
BY (Authorized Signature)	DATE SIGNED	
<u>K</u>		
PRINTED NAME AND TITLE OF PERSON SIGNING		
ADDRESS		
STATE OF CALIFORNIA		
AGENCY NAME		
Department of General Services, Procurement Division		
BY (Authorized Signature)	DATE SIGNED	
<u>k</u>		
PRINTED NAME AND TITLE OF PERSON SIGNING		
ADDDEGG		
ADDRESS	□ Evennt	
	Exempt	

#### COMPETITIVE BIDDING AND PROPOSAL RESPONSIVENESS

The purpose of competitive bidding is to secure public objectives in the most effective manner and avoid the possibilities of graft, fraud, collusion, etc. Competitive bidding is designed to benefit the public body (the State, in the present context), and is not for the benefit of the bidders. It is administered to accomplish its purposes with sole reference to the public interest. It is based upon full and free bidding to satisfy State specifications, and acceptance by the State of the most effective solution to the State's requirements, as determined by the evaluation criteria contained in the RFP.

Competitive bidding is not defined in any single statute but is more in the nature of a compendium of numerous court decisions. From such court decisions, the following rules have evolved, among others:

- 1) Request for Proposals must provide a basis for full and fair competitive bidding among bidders on a common standard, free of restrictions tending to stifle competition.
- 2) The State may modify the IFB/RFP, prior to the date fixed for Contract Award, by issuance of an addendum to all parties who are bidders.
- To have a valid bid, the bid must respond and conform to the invitation, including all the documents which are incorporated therein. A bid which does not literally comply may be rejected.
- 4) For a variance between the request for bids and the bid to be such as to preclude acceptance (the bid must be rejected), the variance or deviation must be a material one.
- 5) State agencies usually have the express or implied right to reject any and all bids in the best interests of the State. Bids cannot, however, be selectively rejected without cause.
- Bids cannot be changed after the time designated for receipt and opening thereof. Other than allowed by law, no negotiation as to the scope of the work, amount to be paid, or contractual terms is permitted. However, this does not preclude the State from clarifying the bidder's intent by asking questions and considering answers.
- 7) A competitive bid, once opened and declared, is in the nature of an irrevocable option and a contract right of which the public agency cannot be deprived without its consent, unless the requirements for rescission are present. All bids become public documents.
- 8) Bids cannot be accepted "in part," unless the invitation specifically permits such an award.
- 9) Contracts entered into through the competitive bidding process cannot later be amended, unless the Invitation for Bids includes a provision, to be incorporated in the contract awarded, providing for such amendment.

Since competitive procurement became the required method for securing certain IT goods or services, the State has received a number of bids which were deemed to be nonresponsive to the Invitation for Bids or which could not be considered as valid bids within the competitive bidding procedures. Nonresponsive bids or bids which contain qualifications must be rejected. Many of the causes for rejection arise from either an incomplete understanding of the competitive bidding process or administrative oversight on the part of the bidders. The following examples are illustrative of more common causes for rejection of bids. These examples are listed to assist potential bidders in submission of responsive bids.

- 1) A bid stated, "The prices stated within are for your information only and are subject to change."
- 2) A bid stated, "This proposal shall expire thirty (30) days from this date unless extended in writing by the \_\_\_\_ Company." (In this instance award was scheduled to be approximately 45 days after bid submittal date.)
- 3) A bid for lease of IT equipment contained lease plans of a duration shorter than that which had been requested in the IFB/RFP.
- 4) A personal services contract stated, ", in its judgment, believes that the schedules set by the State are extremely optimistic and probably unobtainable. Nevertheless, will exercise its best efforts..."
- 5) A bid stated, "This proposal is not intended to be of a contractual nature."
- 6) A bid contained the notation "prices are subject to change without notice."
- 7) A bid was received for the purchase of IT equipment with unacceptable modifications to the Purchase Contract.
- 8) A bid for lease of IT equipment contained lease plans of a duration longer than that which had been requested in the IFB/RFP with no provision for earlier termination of the contract.
- 9) A bid for lease of IT equipment stated, "...this proposal is preliminary only and the order, when issued, shall constitute the only legally binding commitment of the parties."
- 10) A bid was delivered to the wrong office.
- 11) A bid was delivered after the date and time specified in the IFB/RFP.
- 12) An IFB/RFP required the delivery of a performance bond covering 25 percent of the proposed contract amount. The bid offered a performance bond to cover "x" dollars which was less than the required 25 percent of the proposed contract amount.
- 13) A bid did not meet contract goal for DVBE participation and did not follow the steps required by the bid to achieve a "good faith effort."
- A bid appeared to meet contract goal for DVBE participation with the dollars submitted, but the supplier had miscalculated the bid costs. When these corrections were made by the State, the supplier's price had increased and the dollars committed for DVBE participation no longer met goal. The supplier had not followed the steps to achieve a "good faith effort."

#### ADMINISTRATIVE REQUIREMENTS RESPONSE

The Administrative Requirements Response must be completed by the Bidder and submitted according to Section 8, Proposal Format. This exhibit is designed for the Bidder's use to assist in meeting the State's administrative requirements of the RFP and for the State's use to readily identify if bidders are responsible and responsive to Section 5, Administrative Requirements.

Bidders must complete Sections A and B below as follows:

- Read the detail regarding each proposal item and its requirements. Do not depend on the brief summary title given to the Administrative Requirement description on this exhibit.
- Indicate whether their proposal meets the individual requirement by marking in either the "Yes" or "No" column.
- Initial next to the corresponding requirement. The designated individual must initial to the right of each item indicating the Bidder's intention to meet the requirement.
- Identify the specific page number in the Bidder's proposal where the requirement is addressed. Provide any additional exhibits as appropriate to illustrate responsiveness.

#### **SECTION A - Mandatory Requirements**

The following requirements are <u>mandatory</u> and must be responded to in accordance with Section 5, Administrative Requirements.

No.	RFP	Description	Meets	Req.	Confirm With	Location in Proposal
	Reference Description		Yes	No	Initials	(Page #)
1.	5.4.1	Bidder Responsibility				
2.	5.4.2	Confidentiality				
3.	5.4.3	Contractor Responsibility				
4.	5.4.4	Contractor Name Change				
5.	5.4.5	Bidder Declaration				
6.	5.4.5.1	Subcontractors				
7.	5.4.5.2	Notice of Subcontractors				
8.	5.4.6	California Disabled Veteran Business Enterprise (DVBE) Program Requirement				
9.	5.4.7	DVBE Declaration				
10.	5.4.8	Commercially Useful Function (CUF)				
11.	5.4.9	Payee Data Record				
12.	5.4.10	Seller Permit				
13.	5.4.11	Contract Terms				
14.	5.4.12	Non-Discrimination Toward WTO GPA Signatories				
15.	5.4.13	Darfur Contracting Act				

#### **SECTION B – Non-Mandatory Preferences/Incentives**

The following preferences/incentives are <u>non-mandatory</u> and therefore bidders are not required to apply for these preferences/incentives in order to be considered responsive to this RFP. Bidders should refer to Section 5.5, Preferences and Incentives, for specific requirements.

No.	RFP Reference	Description	Apply Prefer	Confirm With	Location in Proposal	
	Keierence		Yes	No	Initials	(Page #)
1.	5.5.1	Small Business Preference				
2.	5.5.2	DVBE Incentive (Waived)				
3.	5.5.3	Target Area Contract Preference Act (TACPA) Preference				
4.	5.5.4	Enterprise Zone Act (EZA) Preference				
5.	5.5.5	Local Agency Military Base Recovery Area (LAMBRA) Act Preference				

#### **RETAILER'S SELLER PERMIT INFORMATION**

In accordance with Senate Bill 1009, Bidders must provide their company's California retailer's seller's permit or certification of registration and, if applicable, the permit or certification of all participating affiliates, issued by California's State Board of Equalization (BOE), pursuant to all requirements as set forth in sections 6487, 7101 and section 6452.1, 6487.4, 18510 of the Revenue and Taxation Code, and Section 10295.1 of the Public Contract Code. In order to expedite the process of verifying the validity of the permit, provide the BOE permit number in the space provided below (or attach a copy of the permit with proposal).

Retailer's Seller's Permit Number:	
Signature of Representative	Date
Typed Name & Title of Representative	
Typed Name of Company	

#### **CONTRACTOR CONTRACT MANAGER INFORMATION**

Bidder must provide the contact information for the Contractor Contract Manager below:

Company Name:	
Contact Name:	
Address:	
City, State & Zip Code:	
Telephone Number:	
Facsimile Number:	
Email:	

#### **CONTRACTOR ORDERING INFORMATION**

#### **ORDER INFORMATION (Standard):**

Provide business email address, facsimile and address location to which agency orders will be sent (per Section 6.10.5) below:

Company Name:		
Address:		
City, State & Zip Code:		
Telephone Number:		
Facsimile Number:		
Email:		
Contact Name:		
ORDER INFORMATION (Emerg	ency Orders Only):	
Provide emergency order contact 6.10.6) below:	information for orders placed during a declared emergency (per	Section
Contact Name:		
Telephone Number:		
Facsimile Number:		
Email:		

### MINIMUM PRICING DISCOUNT LIGHT DUTY CARS/TRUCKS

CATALOG DISCOUNT IS A FIXED PERCENTAGE OFF CATALOG PRICING FOR EACH CATEGORY SPECIFIED. FOR EVALUATION PURPOSES, ALL PARTS SHALL BE NEW AND UNUSED. BIDDER MUST SUPPLY A HARD COPY OF ALL CATALOGS REFERENCED. BIDDER MUST IDENTIFY CATALOG NAME NUMBER AND DATE FOR EACH CATEGORY ON A SEPARATE SHEET.

CATEGORY	DESCRIPTION	PRICING DISCOUNT
1	AIR CONDITIONING	%
2	ALTERNATORS AND STARTERS	%
3	BATTERIES	%
4	BEARINGS, BALL AND ROLLER	%
5	BELTS AND HOSES	%
6	BRAKES	%
7	ELECTRICAL AND IGNITION	%
8	EMISSIONS AND EXHAUST	%
9	ENGINE AND DRIVE TRAIN	%
10	FILTERS, OIL, GAS, AIR, AND TRANS	%
11	GASKETS AND SEALS	%
12	HEATING AND COOLING (ENGINE)	%
13	LAMPS AND LIGHTING AND MIRRORS	%
14	OILS, AND LUBRICANTS REGULAR AND SYNTHETIC	%
15	PUMPS, FUEL AND WATER	%
16	SUSPENSION, SHOCKS, STRUTS, AND STEERING	%
17	WIPERS/WASHERS	%
18	WINTER ACCESSORIES	%

RFP DGS 57224 EXHIBIT 10.10

#### PC RC Certification Workbook

#### Post Consumer Recycled Content (PC RC) Percentages Worksheet

Solicitation No.:	RFP DGS 57224	
Bidder Name:		
Revision Date:		

#### Instructions:

- 1. For all line items offered to the State, Bidder/Contractor shall list the minimum percentage, if not exact percentage, of postconsumer material within the State Agency Buy Recycled Campaign (SABRC) Reportable Product Categories.
- 2. If the line item does not contain any content type matching the SABRC Product/material category, bidders shall enter "N/A" in the the percentage columns.
- 3. If the line item does not contains content type that matches the SABRC Product/material category but does not contain any post consumer recycled content of this type, bidders shall enter "0" in the the percentage columns.
- 4. Refer to SABRC Reportable Product Categories Table for more information regarding categories listed.

			s In	Pos	stconsur	ner Rec	ycled Co	ontent P	ercenta	ge (%) p	er SABR	C Produ	ict Cate	gory	
Contract Line Item Number (CLIN)	Item Description	SKU#	Product meets SABRC Minimum PCRC Content	Antifreeze	Compost, Co- Compost & Mulch	Glass Products	Lubricating Oils	Metal Products	Paint	Paper Products	Plastic Products	Printing and Writing Papers	Tire-Derived Product	Tires	
1	White copy paper	ABC12345	Yes	n/a	n/a	n/a	n/a	n/a	n/a	n/a	n/a	100	n/a	n/a	< Sample
															<u> </u>
															1
															1
															1
															1
															1
															]

RFP DGS 57224 EXHIBIT 10.10

#### PC RC Certification Workbook

#### **Letter of Certification**

#### Instructions:

All Bidders shall print and sign this "Letter of Certification", certifying the minimum percentage, if not exact percentage, listed in the spreadsheet entitled "PC RC Percentages Worksheet" is the post consumer recycled content (PC RC) material in each product listed and offered in this contract. The certification shall be furnished under penalty of perjury.

Pursuant to Public Contract Code 12205(a)(1), I certify under penalty of perjury under the laws of the State of California that the information provided in the PC RC Table worksheet in the Post Consumer Recycled Content (PC RC) Percentages Workbook is true and correct.					
Solicitation No.:					
Company Name:					
Address:					
Name (printed):	Title:				
Signature:	Date:				

#### REPORTABLE PRODUCT CATEGORIES TABLE

SABRC Product Categories	CIWMB* Code No.	Product Examples	Minimum Postconsumer Content Requirement
Antifreeze	8	Recycled antifreeze, and antifreeze containing a bittering agent or made from polypropylene or other similar non-toxic substance.	70 percent postconsumer material.
Compost, Co- compost and Mulch Products	3	Soil amendments, erosion controls, soil toppings, ground covers, weed suppressants, and organic materials used for water conservation; yard trimmings and wood byproducts that are separated from the municipal solid waste stream or other source of organic materials such as biosolids or other comparable substitutes such as livestock, horse, or other animal manure, food residues or fish processing byproducts; mechanical breakdown of materials.	80 percent recovered material that would otherwise be normally disposed of in a landfill.
Glass Products	4	Windows, test tubes, beakers, laboratory or hospital supplies, fiberglass (insulation), reflective beads, tiles, construction blocks, desktop accessories, flat glass sheets, loose-grain abrasives, deburring media, liquid filter media, and containers.	10 percent postconsumer, by weight.
Lubricating Oils	5	Intended for use in a crankcase, transmission, engine, power steering, gearbox, differential chainsaw, transformer dielectric, fluid, cutting, hydraulic, industrial, or automobile, bus, truck, vessel, plane, train, heavy equipment, or machinery powered by an internal combustion engine.	70 percent re-refined base oil.
Metal	11	Staplers, paper clips, steel furniture, desks, pedestals, scissors, jacks, rebar, pipe, plumbing fixtures, chairs, ladders, file cabinets, shelving, containers, lockers, sheet metal, girders, building and construction products, bridges, braces, nails, and screws.	10 percent postconsumer material, by weight.
Paint	7	Water-based paint, graffiti abatement, interior and exterior, and maintenance.	50 percent postconsumer paint (exceptions when 50 percent postconsumer content is not available or is restricted by a local air quality management district, then 10 percent postconsumer content may be substituted).
Paper Products	1	Paper janitorial supplies, cartons, wrapping, packaging, file folders, and hanging files, building insulation and panels, corrugated boxes, tissue, and toweling.	30 percent by fiber weight postconsumer fiber.
Plastic Products	6a	Printer or duplication cartridges, diskette, carpet, office products, plastic lumber, buckets, wastebaskets, containers, benches, tables, fencing, clothing, mats, packaging, signs, posts, binders, sheet, buckets, building products, garden hose, and trays.	10 percent postconsumer, by weight.
Printer or Duplication Cartridges	6b		A. Have 10 percent postconsumer material, or
			B. Are purchased as remanufactured, or C. Are backed by a vendor-offered program that will take back the printer cartridge after their useful life and ensure that the cartridge is recycled and comply with the definition of recycled as set forth in section Public Contract Code 12156.
Printing and Writing Papers	2	Copy, xerographic, watermark, cotton fiber, offset, forms, computer printout paper, white wove envelopes, manila envelopes, book paper, note pads, writing tablets, newsprint, and other uncoated writing papers, posters, index cards, calendars, brochures, reports, magazines, and publications.	30 percent by fiber weight postconsumer fiber.
Tires	9	Truck and bus tires, and those used on fleet vehicles and passenger cars.	Retreaded: Must use an existing casing that has undergone retreading or recapping process in accordance with Public Resource Code (commencing with section 42400).
Tire- Derived Products	10	Flooring, mats, wheelchair ramps, playground cover, parking bumpers, bullet traps, hoses, bumpers, truck bedliners, pads, walkways, tree ties, road surfacing, wheel chocks, rollers, traffic control products, mudflaps, and posts.	50 percent recycled used tires.

<sup>\*</sup> CIWMB - Calfornia Integrated Waste Management Board

# CALIFORNIA DEPARTMENT OF CORRECTIONS AND REHABILITATION

#### SPECIAL PROVISIONS

#### **Digest of Laws**

By signing this contract, the Contractor agrees that, if the provisions of the contract require the contractor to enter an institution, the Contractor and his/her employees shall be aware of and abide by the following rules:

#### DIGEST OF LAWS RELATED TO ASSOCIATION WITH PRISON INMATES

Individuals who are not employees of the California Department of Corrections and Rehabilitations (CDCR), but who are working in and around inmates who are incarcerated within California's institutions/facilities or camps, are to be apprised of the laws, rules and regulations governing conduct in associating with prison inmates. The following is a summation of pertinent information when non-departmental employees come in contact with prison inmates.

a. Persons who are not employed, by CDCR, but are engaged in work at any institution/facility or camp must observe and abide by all laws, rules and regulations governing the conduct of their behavior in associating with prison inmates. Failure to comply with these guidelines may lead to expulsion from CDCR institutions/facilities or camps.

SOURCE: California Penal Code (PC) Sections 5054 and 5058; California Code of Regulations (CCR), Title 15, Sections 3285 and 3415

b. CDCR does not recognize hostages for bargaining purposes. CDCR has a "NO HOSTAGE" policy and all prison inmates, visitors, and employees shall be made aware of this.

SOURCE: PC Sections 5054 and 5058; CCR, Title 15, Section 3304

c. All persons entering onto institution/facility or campgrounds consent to a search of their person, property or vehicle at any time. Refusal by individuals to submit to a search of their person, property or vehicle may be cause for denial of access to the premises.

SOURCE: PC Sections 2601, 5054 and 5058; CCR, Title 15, Sections 3173, 3177, and 3288

d. Persons normally permitted to enter an institution/facility or camp may be barred, for cause, by the CDCR Director, Warden and /or Regional Parole Administrator.

SOURCE: PC Sections 5054 and 5058; CCR, Title 15, Section 3176 (a)

e. It is illegal for an individual who has been previously convicted of a felony offense to enter into CDCR institutions/facilities or camps without the prior approval of the Warden. It is also illegal for an individual to enter onto these premises for unauthorized purposes or to refuse to leave said premises when requested to do so. Failure to comply with this provision could lead to prosecution.

SOURCE: PC Sections 602, 4570.5 and 4571; CCR, Title 15, Sections 3173 and 3289

f. Encouraging and/or assisting prison inmates to escape is a crime. It is illegal to bring firearms, deadly weapons, explosives, tear gas, drugs or drug paraphernalia on CDCR institutions/facilities or camp premises. It is illegal to give prison inmates firearms, explosives, alcoholic beverages, narcotics, or any drug or drug paraphernalia, including cocaine or marijuana.

SOURCE: PC Sections 2772, 2790, 4533, 4535, 4550, 4573, 4573.5, 4573.6 and 4574

g. It is illegal to give or take letters from prison inmates without the authorization of the Warden. It is also illegal to give or receive any type of gift and/or gratuities from prison inmates.

SOURCE: PC Sections 2540, 2541 and 4570; CCR, Title 15, Sections 3010, 3399, 3401, 3424 and 3425

h. In an emergency situation, the visiting program and other program activities may be suspended.

SOURCE: PC Section 2601; CCR, Title 15, Section 3383

i. For security reasons, visitors must not wear clothing that in any way resembles state issued prison inmate clothing (blue denim shirts, blue denim pants).

SOURCE: CCR, Title 15, Section 3171 (b) (3)

j. Interviews with SPECIFIC INMATES are not permitted. Conspiring with an inmate to circumvent policy and/or regulations constitutes a rule violation that may result in appropriate legal action.

SOURCE: CCR, Title 15, Section 3261.5, 3315 (3) (W), and 3177

#### **ASSEMBLY BILL 384 -- TOBACCO USE PROHIBITION**

Assembly Bill (AB) 384 was signed into law by Governor Arnold Schwarzenegger and takes effect on July 1, 2005. AB 384 bans the use of tobacco products (i.e., cigarettes, cigars, snuff, and chewing tobacco) by all staff, visitors, inmates, and all persons on the grounds of any institution. The only time tobacco use is permitted is during Department approved religious ceremonies and in residential staff housing where inmates are not present.

#### DARFUR CONTRACTING ACT CERTIFICATION

Pursuant to Public Contract Code section 10478, if a bidder or proposer currently or within the previous three years has had business activities or other operations outside of the United States, it must certify that it is not a "scrutinized" company as defined in Public Contract Code section 10476.

Therefore, to be eligible to submit a bid or proposal, please insert your company name and Federal ID Number and complete <u>only **one** of the following</u> three paragraphs (via initials for Paragraph # 1 or Paragraph # 2, or via initials and certification for Paragraph # 3):

Comp	any/Vendor Nam	e (Printed)		Federal ID Number		
Printe	ed Name and Title	e of Person Initialing (for	Options 1 or 2)			
1.	Initials		y have, or we have not had bess activities or other operation		d	
2.	Initials	We are a scrutinized company as defined in Public Contract Code section 10476, but we have received written permission from the Department of General Services (DGS) to submit a bid or proposal pursuant to Public Contract Code section 10477(b). A copy of the written permission from DC is included with our bid or proposal.				
3. +	Initials certification below	business activities but we certify below	OR or we have had within the portions outside withat we are not a scrutinized Contract Code section 104	of the United States, ed company		
I, the	ective proposer/b	low, CERTIFY UNDER F	PENALTY OF PERJURY that I and above in # 3. This certification is	n duly authorized to legally bir s made under the laws of the	nd the State of	
By (Au	uthorized Signatu	re)				
Printe	d Name and Title	of Person Signing				
Date I	Executed		Executed in the County and Stat	te of		

Note Regarding Change of Status - If your company has a change of status, with regard to this certification, during the term of your contract (s) then it is incumbent on your company to submit an updated Certification. For questions regarding a change of status, please contact the contract's designated State Contract Administrator.

Ex			

#### **CONTRACT USAGE REPORT - AUTO PARTS**

	Contract	or Name:																
	Contract	Number:																
	Reportin	g Period:																
Column 1	Column 2	Column 3	Column 4	Column 5	Column 6	Column 7	Column 8	Column 9	Column 10	Column 11	Column 12	Column 13	Column 14	Column 15	Column 16	Column 17	Column 18	Column 19
Supplier Contract Usage ID	Ordering Agency Name	Purchasing Authority Number (for State departments)	Agency Billing Code	Purchase Order Number	Purchase Order Date	Manufacturer Part Number	Manufacturer	SKU#/ Item#	Item Description		Quantity in Unit of Measure (If Applicable)	Index Date / Catalog Version	Quantity	List Price/M SRP	Contract Discoun t		Extended Contract Price Paid	Category #
											-							
			1								<b></b>							

#### **CUSTOMER REFERENCE FORM**

Contractor's (Bidder) Name:	
Agency Name (Bidder's Customer):	
Contract Number:	
Contract Duration:	
Dollar Amount of Contract:	
Products/Services Provided:	
Customer (Bidder's Customer) Contact Name and Title:	
Customer Phone Number:	
Customer Fax Number:	
Customer E Mail Address:	

**Ratings:** Please summarize contractor performance and circle in the column on the right the number which best corresponds to the performance rating for each question. If the score is either 1 or 5, please kindly provide an explanation.

Please follow the rating guidelines below for description of rating scale:

Rating Guidelines and Description of Rating Scale:						
Exceptional	(5)	Best-in-class performance. Performance met all contract requirements and exceeded several to the customer's benefit. No issues were encountered.				
Very Good	(4)	Performance met all contract requirements and exceeded some to the customer's benefit. There were a few minor issues, which were negligible.				
Satisfactory	(3)	Performance met contract requirements. There were some minor issues, and corrective actions taken by the contractor were acceptable.				
Marginal	(2)	Performance did not meet the contractual requirements. There were issues, some of a serious nature, for which corrective action was only somewhat effective.				
Unsatisfactory	(1)	Performance did not meet contractual requirements. There were serious issues and the contractor's corrective actions were ineffective.				

#### **CUSTOMER REFERENCE FORM**

Factors Rated	Questions	Comments (continue on additional sheets if desired) Mandatory if score is either 1 or 5	Rating
	How would you rate the contractor's geographic coverage and ability to deliver on time throughout all your locations?		① ② ③ ④ ⑤
Timeliness	How would you rate the contractor's product availability and fill rate?		02345
	How would you rate the contractor's turnaround time when contacted to provide on-site assistance?		① ② ③ ④ ⑤
Contract	How would you rate the experience of the contractor in managing large accounts?		①②③④⑤
Management	How would you rate the service provided by the contractor's assigned Contract Administrator and/or Project Manager?		① ② ③ ④ ⑤
Quality	6. How would you rate the quality of the contractor's value-added services (i.e. installation, assembly, design and CAD input services)?		① ② ③ ④ ⑤
Quality	7. How would you rate the performance of contractor's products compared to that of its competitors?		① ② ③ ④ ⑤
Ordering	How would you rate the contractor's ordering system?		0 2 3 4 5
Reporting	9. How would you rate the contractor's ability to provide reports in an accurate and timely manner?		① ② ③ ④ ⑤
Customer Satisfaction	How would you rate your level of overall satisfaction with the contractor?		① ② ③ ④ ⑤

Rater's Signature:	Date:

# Standard Contract Terms and Conditions Western States Contracting Alliance Request for Proposal

**PARTICIPANTS:** The Western States Contracting Alliance (herein WSCA) is a cooperative group contracting consortium for state government departments, institutions, agencies and political subdivisions (i.e., colleges, school districts, counties, cities, etc.,) for the states of Alaska, Arizona, California, Colorado, Hawaii, Idaho, Minnesota, Montana, Nevada, New Mexico, Oregon, South Dakota, Utah, Washington and Wyoming. Obligations under this contract are limited to those Participating States who have signed (and not revoked) an Intent to Contract at the time of award, or who have executed a Participating Addendum where contemplated by the solicitation. Financial obligations of Participating States are limited to the orders placed by the departments or other state agencies and institutions having available funds. Participating States incur no financial obligations on behalf of political subdivisions. Unless otherwise specified in the solicitation, the resulting award(s) will be permissive.

**QUALITY ESTIMATES:** WSCA does not guarantee to purchase any amount under the contract to be awarded. Estimated quantities are for the purposes of submitting proposals only and are not to be construed as a guarantee to purchase any amount.

**SPECIFICATIONS:** Any deviation from specifications must be clearly indicated by offeror, otherwise, it will be considered that the proposal is in strict compliance. When BRAND NAMES or manufacturers' numbers are stated in the specifications they are intended to establish a standard only and are not restrictive unless the RFP states "No substitute". Proposals will be considered on other makes, models or brands having comparable quality, style, workmanship and performance characteristics. Alternate proposals offering lower quality or inferior performance will not be considered.

**ACCEPTANCE OR REJECTION OF PROPOSALS:** WSCA reserves the right to accept or reject any or all proposals or parts of proposals, and to waive informalities therein.

**SAMPLES:** Generally, when required, samples will be specifically requested in the Request for Proposals. Samples, when required, are to be furnished free of charge. Except for those samples destroyed or mutilated in testing, samples will be returned at an offeror's request, transportation collect.

**CASH DISCOUNT TERMS:** Offeror may quote a cash discount based upon early payment; however, discounts offered for less than 30 days will not be considered in making the award. The date from which discount time is calculated shall be the date a correct invoice is received or receipt of shipment, whichever is later; except that if testing is performed, the date shall be the date of acceptance of the merchandise.

**TAXES:** Proposal prices shall be exclusive of state sales and federal excise taxes. Where the state government entities are not exempt from sales taxes on sales within their state, the contractor shall add the sales taxes on the billing invoice as a separate entry.

**MODIFICATION OR WITHDRAWAL OF PROPOSALS:** Proposals may be modified or withdrawn prior to the time set for the opening of proposals. After the time set for the opening of proposals no proposal may be modified or withdrawn, unless done in response to a request for a "Best and Final Offer" from WSCA.

**PATENTS, COPYRIGHTS, ETC.:** The Contractor shall release, indemnify and hold the Buyer, its officers, agents and employees harmless from liability of any kind or nature, including the Contractor's use of any copyrighted or un-copyrighted composition, secret process, patented or unpatented invention, article or appliance furnished or used in the performance of this contract.

**AWARD:** WSCA may award multiple contracts as the result of this solicitation. Awards shall be made to the responsible offeror(s) whose proposal is determined to be the most advantageous to WSCA, taking into consideration price and the other evaluation factors set forth in the RFP.

**NON-COLLUSION:** By signing the proposal the offeror certifies that the proposal submitted, has been arrived at independently and has been submitted without collusion with, and without any agreement, understanding or planned common course of action with, any other vendor of materials, supplies, equipment or services described in the Request for Proposal, designed to limit independent bidding or competition.

**CANCELLATION:** Unless otherwise stated in the special terms and conditions, any contract entered into as a result of this bid may be canceled by either party upon 60 days notice, in writing, prior to the effective date of the cancellation. Further, any Participating State may cancel its participation upon 30 days written notice, unless otherwise limited or stated in the special terms and conditions of this solicitation. Cancellation may be in whole or in part. Any cancellation under this provision shall not effect the rights and obligations attending orders outstanding at the time of cancellation, including any right of and Purchasing Entity to indemnification by the Contractor, rights of payment for goods/services delivered and accepted, and rights attending any warranty or default in performance in association with any order. Cancellation of the contract due to Contractor default may be immediate.

**DEFAULT AND REMEDIES:** Any of the following events shall constitute cause for WSCA to declare Contractor in default of the contract: 1. Nonperformance of contractual requirements; 2. A material breach of any term or condition of this contract WSCA shall issue a written notice of default providing a period in which Contractor shall have an opportunity to cure. Time allowed for cure shall not diminish or eliminate Contractor's liability for liquidated or other damages. If the default remains, after Contractor has been provided the opportunity to cure, WSCA may do one or more of the following: 1. Exercise any remedy provided by law; 2. Terminate this contract and any related contracts or portions thereof; 3. Impose liquidated damages; 4. Suspend contractor from receiving future proposal solicitations.

**LAWS AND REGULATIONS:** Any and all supplies, services and equipment offered and furnished shall comply fully with all applicable Federal and State laws and regulations.

**CONFLICT OF TERMS:** In the event of any conflict between these standard terms and conditions and any special terms and conditions which follow; the special terms and conditions shall govern.

**REPORTS:** The contractor shall submit quarterly reports to the WSCA Contract Administrator showing the quantities and dollar volume of purchases by each agency.

**HOLD HARMLESS:** The contractor shall release, protect, indemnify and hold WSCA and the respective states and their officers, agencies, employees, harmless from and against any damage, cost or liability, including reasonable attorney's fees for any or all injuries to persons, property or claims for money damages arising from acts or omissions of the contractor, his employees or subcontractors or volunteers.

**ORDER NUMBERS:** Contract order and purchase order numbers shall be clearly shown on all acknowledgments, shipping labels, packing slips, invoices, and on all correspondence.

**GOVERNING LAW:** This procurement and the resulting agreement shall be governed by and construed in accordance with the laws of the state sponsoring and administering the procurement. The construction and effect of any Participating Addendum or order against the contract(s) shall be governed by and construed in accordance with the laws of the Participating Entity's State. Venue for any claim, dispute or action concerning an order placed against the contract(s) or the effect of an Participating Addendum shall be in the Purchasing Entity's State.

**DELIVERY:** The prices offered shall be the delivered price to any WSCA state agency or political subdivision. All deliveries shall be F.O.B. destination with all transportation and handling charges paid

by the contractor. Responsibility and liability for loss or damage shall remain the Contractor until final inspection and acceptance when responsibility shall pass to the Buyer except as to latent defects, fraud and Contractor's warranty obligations. The minimum shipment amount will be found in the special terms and conditions. Any order for less than the specified amount is to be shipped with the freight prepaid and added as a separate item on the invoice. Any portion of an order to be shipped without transportation charges that is back ordered shall be shipped without charge.

**WARRANTY:** As used herein "Buyer" refers to any WSCA state agency or political subdivision. The contractor acknowledges that the Uniform Commercial Code applies to this contract. In general, the contractor warrants that: (a) the product will do what the salesperson said it would do, (b) the product will live up to all specific claims that the manufacturer makes in their advertisements, (c) the product will be suitable for the ordinary purposes for which such product is used, (d) the product will be suitable for any *special purposes* that the Buyer has relied on the contractor's skill or judgment to consider.

**AMENDMENTS:** The terms of this contract shall not be waived, altered, modified, supplemented or amended in any manner whatsoever without prior written approval of the WSCA Contract Administrator.

**ASSIGNMENT/SUBCONTRACT:** Contractor shall not assign, sell, transfer, subcontract or sublet rights, or delegate responsibilities under this contract, in whole or in part, without the prior written approval of the WSCA Contract Administrator.

**NONDISCRIMINATION:** The offeror agrees to abide by the provisions of Title VI and Title VII of the Civil Rights Act of 1964 (42 USC 2000e), which prohibit discrimination against any employee or applicant for employment, or any applicant or recipient of services, on the basis of race, religion, color, or national origin; and further agrees to abide by Executive Order No. 11246, as amended, which prohibits discrimination on basis of sex; 45 CFR 90 which prohibits discrimination on the basis of age, and Section 504 of the Rehabilitation Act of 1973, or the Americans with Disabilities Act of 1990 which prohibits discrimination on the basis of disabilities. The offeror further agrees to furnish information and reports to requesting State(s), upon request, for the purpose of determining compliance with these statutes. Offeror agrees to comply with each individual state's certification requirements, if any, as stated in the special terms and conditions. This contract may be canceled if the offeror fails to comply with the provisions of these laws and regulations. The offeror must include this provision in very subcontract relating to purchases by the States to insure that subcontractors and vendors are bound by this provision.

**SEVERABILITY:** If any provision of this contract is declared by a court to be illegal or in conflict with any law, the validity of the remaining terms and provisions shall not be affected; and the rights and obligations of the parties shall be construed and enforced as if the contract did not contain the particular provision held to be invalid.

**INSPECTIONS:** Goods furnished under this contract shall be subject to inspection and test by the Buyer at times and places determined by the Buyer. If the Buyer finds goods furnished to be incomplete or in compliance with proposal specifications, the Buyer may reject the goods and require Contractor to either correct them without charge or deliver them at a reduced price, which is equitable under the circumstances. If Contractor is unable or refuses to correct such goods within a time deemed reasonable by the Buyer, the Buyer may cancel the order in whole or in part. Nothing in this paragraph shall adversely affect the Buyer's rights including the rights and remedies associated with revocation of acceptance under the Uniform Commercial Code.

**PAYMENT:** Payment for completion of an contract is normally made within 30 days following the date the entire order id delivered or the date a correct invoice is received, whichever is later. After 45 days the Contractor may assess overdue account charges up to a maximum rate of one percent per month on the outstanding balance. Payments will be remitted by mail. Payments may be made via a State or political subdivision "Purchasing Card".

**FORCE MAJEURE:** Neither party to this contract shall be held responsible for delay or default caused by fire, riot, acts of God and/or war which is beyond that party's reasonable control. WSCA may terminate this contract after determining such delay or default will reasonably prevent successful performance of the contract.

**HAZARDOUS CHEMICAL INFORMATION:** The contractor will provide one set of the appropriate material safety data sheet(s) and container label(s) upon delivery of a hazardous material to the user agency. All safety data sheets and labels will be in accordance with each participating state's requirements.

**FIRM PRICE:** Unless otherwise stated in the special terms and conditions, for the purpose of award, offers made in accordance with this solicitation must be good and firm for a period of ninety (90) days from the date of proposal opening. Prices must remain firm for the full term of the contract.

**EXTENSION OF PRICES**: In the case of error in the extension of prices in the proposal, the unit prices will govern.

**PROPOSAL PREPARATION COSTS**: WSCA is not liable for any costs incurred by the offeror in proposal preparation.

**CONFLICT OF INTEREST:** The contractor certifies that it has not offered or given any gift or compensation prohibited by the state laws of any WSCA participants to any officer or employee of WSCA or participating sates to secure favorable treatment with respect to being awarded this contract.

**INDEPENDENT CONTRACTOR:** The contractor shall be an independent contractor, and as such shall have no authorization, express or implied to bind WSCA or the respective states to any agreements, settlements, liability or understanding whatsoever, and agrees not to perform any acts as agent for WSCA or the states, except as expressly set forth herein.

**POLITICAL SUBDIVISION PARTICIPATION:** Participation under this contract by political subdivisions (i.e., colleges, school districts, counties, cites, etc.,) of the WSCA participating states shall be voluntarily determined by the political subdivision. The contractor agrees to supply the political subdivisions based upon the same terms, conditions and prices.

**DEBARMENT:** The contractor certifies that neither it nor its principals are presently debarred, suspended, proposed for debarment, declared ineligible, or voluntarily excluded from participation in this transaction (contract) by any governmental department or agency. If the contractor cannot certify this statement, attach a written explanation for review by WSCA.

**RECORDS ADMINISTRATION:** The contractor will maintain, or supervise the maintenance of all records necessary to properly account for the payments made to the contractor for costs authorized by this contract. These records will be retained by the contractor for at least four years after the contract terminates, or until all audits initiated within the four years have been completed, whichever is later.

**AUDIT OF RECORDS:** The contractor agrees to allow WSCA, State and Federal auditors, and state agency staff access to all the records to this contract, for audit and inspection, and monitoring of services. Such access will be during normal business hours, or by appointment.

**Entity Participation:** Use of specific WSCA cooperative contracts by state agencies, political subdivisions and other entities (including cooperatives) authorized by individual state's statutes to use state contracts are subject to the approval of the respective State Chief Procurement Official. Issues of interpretation and eligibility for participation are solely within the authority of the respective State Chief Procurement Official.

Revision date: May 2009

#### **SMALL BUSINESS PREFERENCE INFORMATION**

Bidders desiring to claim the small business preference must complete this exhibit as specified in Section 5.5.1, Small Business Preference, and return it with the proposal as specified in Section 8, Proposal Format in order to be eligible for this preference. Bidders are also required to provide certification information on Exhibit 11.7, Bidder Declaration.

Note: Businesses must be certified by California. Questions regarding certifications should be directed to the Office of Small Business and DVBE Certification at (916) 375-4940.

A)	Are you a California certified small business claiming the small business preference?
	<ul><li>No (Continue to question B)</li><li>Yes (Provide bidder information below)</li></ul>
	State of California, Office of Small Business and DVBE Certification reference number:
	Bidder's company is a:   Non-Manufacturer   Manufacturer
B)	Are you a non-small business claiming at least twenty-five percent (25%) small business subcontractor preference?
	<ul> <li>No</li> <li>Yes (Provide subcontractor certification information on Exhibit 11.7, Bidder Declaration Form)</li> </ul>

#### **AUTHORIZED LOCATIONS**

Bidders are to list below locations (which may be company-owned distributors, franchised dealers or retail outlets) that are authorized to participate and have agreed to participate in this contract by supplying vehicle parts on a will-call basis only or delivery basis (Reference Section II, Paragraph 10). Indicated each Delivery Method available at each Location.

(Duplicate this page as necessary to list all locations participating in this contract.)

		Authorized Locations	<b>S</b>			
Name	Address	City/State/Zip	County	Order Placement Contact	Phone	Fax
		7				

Bidder's Company Name:
------------------------

#### WHOLESALE/DISTRIBUTION CENTER LOCATIONS

Automotive Parts, Supplies, And Accessories

	RETAIL STORES IN THIS STATE? (YES OR NO)	IF YES, HOW MANY?	CITY (YES)?	DISTRIBUTION WAREHOUSE(S) IN THIS STATE? (YES OR NO)	IF YES, HOW MANY?	CITY (YES)?
ALASKA	, ,			- /		
ARIZONA						
CALIFORNIA						
COLORADO						
HAWAII						
IDAHO						
LOUISIANA						
MAINE						
MINNESOTA						
MONTANA						
NEBRASKA						
NEVADA						
NEW HAMPSHIRE						
NORTH DAKOTA						
OREGON						
SOUTH DAKOTA						
TENNESSEE						
UTAH						
VERMONT						

#### MARKET BASKET EVALUATION FORM

BIDDER: Various
VEHICLE: State Contract Vehicle

## **Sample**

	T			PART		STATE	STATE
CATEGORY	DESCRIPTION	PART	BRAND	NUMBER	LIST PRICE	DISCOUNT	NET COST
	AIR CONDITIONING;	11.01					
1	Catalog Name, Number & Date	A/C Compressor					
	ALTERNATORS AND STARTERS;						
2	Catalog Name, Number & Date	Starter					
	BATTERIES:						
3	Catalog Name, Number & Date	Battery					
	BEARINGS, BALL AND ROLLER;						
4	Catalog Name, Number & Date	Hub Assy, Front					
	BELTS AND HOSES;	7,					
5	Catalog Name, Number & Date	Belt, Serpentine					
	BRAKES;						
6	Catalog Name, Number & Date	Brake Pads, Front					
	ELECTRICAL AND IGNITION;						
7	Catalog Name, Number & Date	Spark Plug					
	EMISSION AND EXHAUST;						
8	Catalog Name, Number & Date	Oxygen sensor					
	ENGINE AND DRIVE TRAIN;						
9	Catalog Name, Number & Date	Timing Belt					
10	FILTERS, OIL, GAS, AIR AND TRANS;						
	Catalog Name, Number & Date	Oil Filter					
	GASKETS AND SEALS;						
11	Catalog Name, Number & Date	Head Gasket					
12	HEATING AND COOLING (ENGINE);						
	Catalog Name, Number & Date	Radiator					
	LAMPS AND LIGHTING (MIRRORS);						
13	Catalog Name, Number & Date	Headlamp					
1	OILS AND LUBRICANTS REGULAR AND						
14	SYNTHETIC; Catalog Name, Number & Date	Quart of Oil					
1	PUMPS, FUEL AND WATER;	J., 5					
15	Catalog Name, Number & Date	Water Pump					
	SUSPENSION, SHOCKS, STRUTS AND						
4.0	STEERING;						
16	Catalog Name, Number & Date	Shock Absorber					
17	WIPERS/ WASHERS	Win or Diodo					
17	Catalog Name, Number & Date	Wiper Blade					
40	WINTER DRIVING EQUIPMENT; Catalog Name,	Coble Chains					
18	Number and Date	Cable Chains					